Table of Contents		

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

	QUARTERLY REPORT PURSUANT EXCHANGE ACT OF 1934	TO SECTION 13 OR 15(d) OF THE SECURITIES
	For the quarterly period ended September 30, 2	2005.
		OR
	TRANSITION REPORT PURSUANT EXCHANGE ACT OF 1934	TO SECTION 13 OR 15(d) OF THE SECURITIES
	For the transition period fromto _	·
	Commission	file number 0-27918
	Century Alur	ninum Company
	(Exact name of Regist	trant as specified in its Charter)
	Delaware (State of Incorporation)	13-3070826 (IRS Employer Identification No.)
	2511 Garden Road Building A, Suite 200 Monterey, California (Address of principal executive offices)	93940 (Zip Code)
	Registrant's telephone numbe	er, including area code: (831) 642-9300
ct of 1934		ports required to be filed by Section 13 or 15(d) of the Securities Exchange riod that the registrant was required to file such reports), and (2) has been No
Indicate	by check mark whether the registrant is an accelerated f	iler (as defined in Rule 12b-2 of the Exchange Act). ☑ Yes □ No
Indicate	by check mark whether the registrant is a shell company	y (as defined in Rule 12b-2 of the Exchange Act). ☐ Yes ☑ No
The reg	istrant had 32,174,654 shares of common stock outstandi	ing at October 28, 2005.

TABLE OF CONTENTS

PART I — FINANCIAL INFORMATION

Item 1. — Financial Statements

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Item 4. Controls and Procedures

Part II. OTHER INFORMATION

Item 4. Submission of Matters to a Vote of Stockholders

Item 6. Exhibit Index

SIGNATURES

Exhibit Index

EX-10.1: LOAN AND SECURITY AGREEMENT

EX-31.1: CERTIFICATION

EX-31.2: CERTIFICATION

EX-32.1: CERTIFICATION

PART I – FINANCIAL INFORMATION

Item 1. – Financial Statements.

CENTURY ALUMINUM COMPANY CONSOLIDATED BALANCE SHEETS

(Dollars in Thousands) (Unaudited)

	September 30, 2005	December 31, 2004 (Restated)
ASSETS		(riestates)
Current Assets:	Φ 55.047	¢ 44.160
Cash and cash equivalents	\$ 55,847	\$ 44,168
Restricted cash	2,028	1,678
Accounts receivable – net	80,510	79,576
Due from affiliates	17,617 106,208	14,371
Inventories Proposid and other asserts	22,348	111,284 10,055
Prepaid and other current assets Deferred taxes – current portion	14,294	
•		24,642
Total current assets	298,852	285,774
Property, plant and equipment – net	995,236	806,250
Intangible asset – net	78,316	86,809
Goodwill	94,844	95,610
Other assets	79,734	58,110
Total	\$ 1,546,982	\$ 1,332,553
LIABILITIES AND SHAREHOLDERS' EQUITY		
Current Liabilities:		
Accounts payable – trade	\$ 60,182	\$ 47,479
Due to affiliates	78,391	84,815
Accrued and other current liabilities	36,870	53,309
Accrued employee benefits costs — current portion	8,458	8,458
Long-term debt – current portion	558	10,582
Convertible senior notes	175,000	175,000
Industrial revenue bonds	7,815	7,815
Total current liabilities	367,274	387,458
Senior unsecured notes payable	250,000	250,000
Nordural debt	196,601	80,711
Accrued pension benefits costs – less current portion	13,421	10,685
Accrued postretirement benefits costs — less current portion	94,066	85,549
Other liabilities	33,290	34,961
Due to affiliates – less current portion	78,735	30,416
Deferred taxes	74,485	68,273
Total noncurrent liabilities	740,598	560,595
Contingencies and Commitments (See Note 8) Shareholders' equity:		
Common stock (one cent par value, 100,000,000 and 50,000,000 shares authorized at September 30, 2005 and December 31, 2004, respectively; 32,174,654 and 32,038,297 shares outstanding at		
September 30, 2005 and December 31, 2004, respectively)	322	320
Additional paid-in capital	418,876	415,453
Accumulated other comprehensive loss	(33,388)	(52,186)
Retained earnings	53,300	20,913
Total shareholders' equity	439,110	384,500
Total	\$ 1,546,982	\$ 1,332,553

CONSOLIDATED STATEMENTS OF OPERATIONS
(In Thousands, Except Per Share Amounts)
(Unaudited)

	Three months ended September 30,		Nine mon Septem	
	2005	2004	2005	2004
NET SALES:		(Restated)		(Restated)
	\$222,811	\$231,502	\$713,565	\$649,278
Third-party customers Related parties	48.025	42,815	125,923	120,866
Related parties	270,836	274,317	839,488	770,144
Cost of goods sold	240,778	230,835	712,515	641,630
Gross profit	30.058	43,482	126,973	128,514
Gloss plott	30,030	73,702	120,773	120,514
Selling, general and administrative expenses	8,104	7,567	24,946	16,966
Operating income	21,954	35,915	102,027	111,548
	,	,	,	,
Interest expense – third party	(6,213)	(10,552)	(19,413)	(32,308)
Interest expense – related party	_	_	_	(380)
Interest income	596	517	1,088	848
Net loss on forward contracts	(53,481)	(3,149)	(52,480)	(17,146)
Loss on early extinguishment of debt	_	(47,448)	(835)	(47,448)
Other income (expense)	(67)	(110)	703	(798)
Income (loss) before income taxes and equity in earnings of joint ventures	(37,211)	(24,827)	31,090	14,316
Income tax benefit (expense)	14,064	8,854	(12,010)	(5,477)
Income (loss) before equity in earnings of joint ventures	(23,147)	(15,973)	19,080	8,839
Equity in earnings of joint ventures	3,076	` <u> </u>	13,323	<u> </u>
Net income (loss)	(20,071)	(15,973)	32,403	8,839
Preferred dividends	` <u> </u>	` <u> </u>	<u> </u>	(769)
Net income (loss) applicable to common shareholders	\$(20,071)	\$(15,973)	\$ 32,403	\$ 8,070
EARNINGS (LOSS) PER COMMON SHARE:				
Basic	\$ (0.62)	\$ (0.50)	\$ 1.01	\$ 0.29
Diluted	\$ (0.62)	\$ (0.50)	\$ 1.01	\$ 0.29
WEIGHTED AVERAGE COMMON SHARES OUTSTANDING:				
Basic	32,162	31,754	32,120	27,542
Diluted	32,162	31,754	32,163	27,659

See notes to consolidated financial statements

CENTURY ALUMINUM COMPANY CONSOLIDATED STATEMENTS OF CASH FLOWS

(Dollars in Thousands) (Unaudited)

	Nine mon Septem	
	2005	2004
CACH ELOWIC EDOM ODED ATINIC A CTIVITATE		(Restated)
CASH FLOWS FROM OPERATING ACTIVITIES: Net income	\$ 32,403	\$ 8,839
Adjustments to reconcile net income to net cash provided by operating activities:	\$ 32,403	\$ 8,839
Unrealized net loss on forward contracts	49,934	4,712
Depreciation and amortization	42,306	36,889
Deferred income taxes	12,010	(2,860)
Pension and other post retirement benefits	11,253	7,253
(Gain) loss on disposal of assets	(20)	7,233
Non-cash loss on early extinguishment of debt	253	9,659
Changes in operating assets and liabilities:	233	9,039
Accounts receivable – net	(934)	(10,342)
Due from affiliates	(3,246)	(10,342)
Inventories	5,076	(4,212)
Prepaids and other current assets	(2,437)	(1,276)
Accounts payable – trade	6,668	7,730
Due to affiliates	2,480	4,606
Accrued and other current liabilities	(17,613)	7,850
Other – net	(10,909)	3,643
V 1	127,224	71,864
Net cash provided by operating activities	127,224	/1,804
CASH FLOWS FROM INVESTING ACTIVITIES:		
Nordural expansion	(200,641)	(17,482)
Purchase of other property, plant and equipment	(9,629)	(8,832)
Business acquisitions, net of cash acquired	(7,029) $(7,000)$	(184,869)
Restricted cash deposits	(350)	(104,009)
Proceeds from sale of property, plant and equipment	101	_
		(211 102)
Net cash used in investing activities	(217,519)	(211,183)
CACH ELOWG EDOM EINANGING A CENTETE		
CASH FLOWS FROM FINANCING ACTIVITIES:	100 027	105 500
Borrowings	188,937	425,569
Repayment of third party debt	(83,138)	(422,846)
Repayment of related party debt	(5.122)	(14,000)
Financing fees	(5,132)	(12,805)
Dividends	(16)	(3,311)
Issuance of common stock	1,323	214,982
Net cash provided by financing activities	101,974	187,589
NET INCREASE IN CASH AND CASH EQUIVALENTS	11,679	48,270
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	44,168	28,204
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 55,847	\$ 76,474

See notes to consolidated financial statements

CENTURY ALUMINUM COMPANY NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Dollars in thousands, except per share amounts)

1. General

The accompanying unaudited interim consolidated financial statements of Century Aluminum Company (the "Company" or "Century") should be read in conjunction with the audited consolidated financial statements for the year ended December 31, 2004. In management's opinion, the unaudited interim consolidated financial statements reflect all adjustments, which are of a normal and recurring nature, that are necessary for a fair presentation of financial results for the interim periods presented. Operating results for the first nine months of 2005 are not necessarily indicative of the results that may be expected for the year ending December 31, 2005. Certain reclassifications of 2004 information were made to conform to the 2005 presentation.

2. Acquisitions

Nordural Acquisition

The Company acquired Nordural in April 2004 and accounted for the acquisition as a purchase using the accounting standards established in Statement of Financial Accounting Standards ("SFAS") No. 141, "Business Combinations." In the first quarter of 2005, goodwill decreased \$766 from previously reported amounts at year-end as the result of asset allocation adjustments. The Company recognized \$94,844 of goodwill in the transaction after the first quarter adjustment. None of the goodwill is expected to be deductible for Icelandic tax purposes. During the second quarter of 2005, the Company determined that certain Nordural earnings would remain invested outside the United States indefinitely.

The purchase price for Nordural was \$195,346, allocated as follows:

Allocation of Purchase Price:	
Current assets	\$ 41,322
Property, plant and equipment	276,597
Goodwill	94,844
Current liabilities	(25,848)
Long-term debt	(177,898)
Other non-current liabilities	(13,671)
Total purchase price	\$ 195,346

The following table represents the unaudited pro forma results of operations for the period ended September 30, 2004 assuming the acquisition occurred on January 1, 2004. The unaudited pro forma amounts may not be indicative of the results that actually would have occurred if the transaction described above had been completed and in effect for the periods indicated. The pro forma results of operations reflect the retroactive restatement of earnings for a change in accounting principle, see Note 3.

	nber 30, 2004
Net sales	\$ 808,519
Net income	15,747
Net income available to common shareholders	14,978
Earnings per share:	
Basic	\$ 0.48
Diluted	\$ 0.48

3. Change in Accounting Principle

During the second quarter of fiscal 2005, the Company changed its method of inventory costing from last-in-first-out (LIFO) to first-in-first-out (FIFO). The Company believes that using the FIFO method provides better matching of expenses and revenues and provides more consistent inventory costing on a company-wide basis. Prior to the change, approximately 69% of the Company's inventory was valued based upon the LIFO method. The change has been applied retroactively and the financial statements have been restated for all prior periods presented. The effect of the change on net income for the three and nine months ended September 30, 2004 was an increase of \$76 and \$1,800, respectively. The effect of the change on retained earnings for the year ended December 31, 2004 was an increase of \$1,683. The effect of the accounting change on income and earnings per share during the three and nine month periods ended September 30, 2004, is as follows:

Three months ended September 30, 2004 (Restated)	Nine months ended September 30, 2004 (Restated)
(16,049)	6,270
76	1,800
(15,973)	8,070
(0.51)	0.23
0.01	0.06
(0.50)	0.29
	
(0.51)	0.23
0.01	0.06
(0.50)	0.29
	ended September 30, 2004 (Restated) (16,049) 76 (15,973) (0.51) 0.01 (0.50) (0.51) 0.01

4. Stock-Based Compensation

The Company has elected not to adopt the recognition provisions for employee stock-based compensation as permitted in SFAS No. 123, "Accounting for Stock-Based Compensation." As such, the Company accounts for stock based compensation in accordance with Accounting Principles Board Opinion No. 25 "Accounting for Stock Issued to Employees." No compensation cost has been recognized for the stock option portions of the plan because the exercise prices of the stock options granted were equal to the market value of the Company's stock on the date of grant. Had compensation cost for the Stock Incentive Plan been determined using the fair value method provided under SFAS No. 123, the Company's net income and earnings per share would have changed to the pro forma amounts indicated as follows:

		Three months ended September 30,		Nine months en September 30				
		2005	05 2004		2005		2004	
			(Restated)			(F	Restated)
Net income (loss) applicable to common shareholders	As Reported	\$ (20,071)	\$	(15,973)	\$	32,403	\$	8,070
Add: Stock-based employee compensation expense included in reported net income, net of related tax effects		514		360		2,197		1,406
Deduct: Total stock-based employee compensation expense determined under fair value based method for all awards, net of related tax effects		(678)		(462)		(2,631)		(1,641)
Pro forma net income (loss)		\$ (20,235)	\$	(16,075)	\$	31,969	\$	7,835
Basic earnings (loss) per share	As reported	\$ (0.62)	\$	(0.50)	\$	1.01	\$	0.29
	Pro forma	\$ (0.63)	\$	(0.51)	\$	1.00	\$	0.28
Diluted earnings (loss) per share	As reported	\$ (0.62)	\$	(0.50)	\$	1.01	\$	0.29
-	Pro forma	\$ (0.63)	\$	(0.51)	\$	0.99	\$	0.28

5. Inventories

Inventories consist of the following:

	September 30, 2005	December 31, 2004 (Restated)
Raw materials	\$ 47,358	\$ 54,186
Work-in-process	12,573	10,215
Finished goods	4,986	8,954
Operating and other supplies	41,291	37,929
	\$ 106,208	\$ 111,284

Inventories are stated at the lower of cost, using the first-in, first-out method, or market.

6. Goodwill and Intangible Asset

The Company recognized \$94,844 of goodwill in the Nordural acquisition, see Note 2. The Company will annually test its goodwill for impairment in the second quarter of the fiscal year and other times whenever events or circumstances indicate that the carrying amount of goodwill may exceed its fair value. If the carrying value of goodwill exceeds its fair value, an impairment loss will be recognized. The fair value is estimated using market comparable information.

The intangible asset consists of the power contract acquired in connection with the Company's acquisition of the Hawesville facility. The contract value is being amortized over its term (10 years) using a method that results in annual amortization equal to the percentage of a given year's expected gross annual benefit to the total as applied to the total recorded value of the power contract. As of September 30, 2005, the gross carrying amount of the intangible asset was \$155,986 with accumulated amortization of \$77,670. In April 2005, the Company made a \$7,000 post-closing payment to the Southwire Company ("Southwire"), a privately held wire and cable manufacturing company, related to the acquisition of the Hawesville facility. This payment satisfied in full the Company's obligation to pay contingent consideration to Southwire under the acquisition agreement. This post-closing payment obligation was allocated to the acquired fixed assets and intangible asset based on the allocation

percentages used in the original acquisition. The gross carrying amount of the intangible asset increased \$2,394 as a result of this liability.

For the nine month periods ended September 30, 2005 and September 30, 2004, amortization expense for the intangible asset totaled \$10,887 and \$9,245, respectively. For the three month periods ended September 30, 2005 and September 30, 2004, amortization expense for the intangible asset totaled \$3,673 and \$3,081, respectively.

For the year ending December 31, 2005, the estimated aggregate amortization expense for the intangible asset will be approximately \$14,561. The estimated aggregate amortization expense for the intangible asset for the following five years is as follows:

For the year ending December 31,

2006
2007
2008
2009
2010
Estimated Amortization Expense
\$13,048
\$13,048
\$13,991
\$15,076
\$16,149
\$16,379

The intangible asset is reviewed for impairment in accordance with SFAS 142, "Goodwill and Other Intangible Assets," whenever events or circumstances indicate that its net carrying amount may not be recoverable.

7. Debt

Secured First Mortgage Notes

In April 2005, the Company exercised its right to call the remaining \$9,945 of 11.75% senior secured first mortgage notes due 2008 that remained outstanding at 105.875% of the principal balance, plus accrued and unpaid interest. The early extinguishment of the notes resulted in a \$253 loss reported as other income (expense).

Revolving Line of Credit

Effective September 19, 2005, the Company replaced its revolving credit facility that was due to expire in March 2006 with a new \$100,000 senior secured revolving credit facility ("Credit Facility") with a syndicate of banks. The Credit Facility will mature September 19, 2010. The Company's obligations under the Credit Facility are unconditionally guaranteed by its domestic subsidiaries (other than Century Aluminum Holdings, Inc., Century Louisiana, Inc., and Nordural US LLC) and secured by a first priority security interest in all accounts receivable and inventory belonging to the Company and its subsidiary borrowers. The availability of funds under the Credit Facility is subject to a \$15,000 reserve and limited by a specified borrowing base consisting of certain eligible accounts receivable and inventory. Borrowings under the Credit Facility are, at the Company's option, at the LIBOR rate or bank base rate, plus or minus in each case an applicable margin. The Credit Facility is subject to customary covenants, including limitations on capital expenditures, additional indebtedness, affiliate transactions, liens, guarantees, mergers and acquisitions, dividends, distributions, capital redemptions and investments. The Company had no outstanding borrowings under the Credit Facility as of September 30, 2005. As of September 30, 2005, the Company had a borrowing availability of \$95,931 under the Credit Facility.

8. Contingencies and Commitments

Environmental Contingencies

The Company believes its current environmental liabilities do not have, and are not likely to have, a material adverse effect on the Company's financial condition, results of operations or liquidity. However, there can be no assurance that future requirements or conditions at currently or formerly owned or operated properties will not result in liabilities which may have a material adverse effect.

Century Aluminum of West Virginia, Inc. ("Century of West Virginia") continues to perform remedial measures at its Ravenswood facility pursuant to an order issued by the Environmental Protection Agency ("EPA") in 1994 (the "3008(h) Order"). Century of West Virginia also conducted a RCRA facility investigation ("RFI")

Notes to Consolidated Financial Statements — (continued)

under the 3008(h) Order evaluating other areas at the Ravenswood facility that may have contamination requiring remediation. The RFI has been approved by appropriate agencies. Century of West Virginia has completed interim remediation measures at two sites identified in the RFI, and the Company believes no further remediation will be required. A Corrective Measures Study, which will formally document the conclusion of these activities, is being completed with the EPA. The Company believes a significant portion of the contamination on the two sites identified in the RFI is attributable to the operations of third parties and is their financial responsibility.

Prior to the Company's purchase of the Hawesville facility, the EPA issued a final Record of Decision ("ROD") under the Comprehensive Environmental Response, Compensation and Liability Act. By agreement, Southwire is to perform all obligations under the ROD. Century Aluminum of Kentucky LLC ("Century Kentucky") has agreed to operate and maintain the ground water treatment system required under the ROD on behalf of Southwire, and Southwire will reimburse Century Kentucky for any expense that exceeds \$400 annually.

Century is a party to an EPA Administrative Order on Consent (the "Order") pursuant to which other past and present owners of an alumina refining facility at St. Croix, Virgin Islands have agreed to carry out a Hydrocarbon Recovery Plan to remove and manage hydrocarbons floating on groundwater underlying the facility. Pursuant to the Hydrocarbon Recovery Plan, recovered hydrocarbons and groundwater are delivered to the adjacent petroleum refinery where they are received and managed. Lockheed Martin Corporation ("Lockheed"), which sold the facility to one of the Company's affiliates, Virgin Islands Alumina Corporation ("Vialco"), in 1989, has tendered indemnity and defense of this matter to Vialco pursuant to the terms of the Lockheed–Vialco Asset Purchase Agreement. Management does not believe Vialco's liability under the Order or its indemnity to Lockheed will require material payments. Through September 30, 2005, the Company has expended approximately \$440 on the Recovery Plan. Although there is no limit on the obligation to make indemnification payments, the Company expects the future potential payments under this indemnification to comply with the Order will be approximately \$200, which may be offset in part by sales of recoverable hydrocarbons.

On May 5, 2005, a complaint was filed by the Commissioner of the Department of Planning and Natural Resources, in his capacity as Trustee for Natural Resources of the United States Virgin Islands, against the Company, Vialco and other parties. The complaint alleges damages to natural resources caused by alleged releases from the alumina refinery facility at St. Croix and the adjacent petroleum refinery. Lockheed has tendered indemnity and defense of the case to Vialco pursuant to terms of the Lockheed-Vialco Asset Purchase Agreement. The complaint seeks unspecified monetary damages, costs and attorney fees.

It is the Company's policy to accrue for costs associated with environmental assessments and remedial efforts when it becomes probable that a liability has been incurred and the costs can be reasonably estimated. The aggregate environmental-related accrued liabilities were \$738 and \$596 at September 30, 2005 and December 31, 2004, respectively. All accrued amounts have been recorded without giving effect to any possible future recoveries. With respect to cost for ongoing environmental compliance, including maintenance and monitoring, such costs are expensed as incurred.

Because of the issues and uncertainties described above, and the Company's inability to predict the requirements of the future environmental laws, there can be no assurance that future capital expenditures and costs for environmental compliance will not have a material adverse effect on the Company's future financial condition, results of operations, or liquidity. Based upon all available information, management does not believe that the outcome of these environmental matters will have a material adverse effect on the Company's financial condition, results of operations, or liquidity.

Legal Contingencies

The Company has pending against it or may be subject to various lawsuits, claims and proceedings related primarily to employment, commercial, environmental and safety and health matters. Although it is not presently possible to determine the outcome of these matters, management believes their ultimate disposition will not have a material adverse effect on the Company's financial condition, results of operations, or liquidity.

Power Commitments

The Hawesville facility currently purchases all of its power from Kenergy Corporation ("Kenergy"), a local retail electric cooperative, under a power supply contract that expires at the end of 2010. Kenergy acquires the power it provides to the Hawesville facility mostly from a subsidiary of LG&E Energy Corporation ("LG&E"), with delivery guaranteed by LG&E. The Hawesville facility currently purchases all of its power from Kenergy at fixed prices. The Company recently priced 46 megawatts ("MW") of previously unpriced power for 2006. As a result, approximately 85 MW or 18% of the Hawesville facility's power requirements are unpriced in 2006. The Hawesville facility's unpriced power requirements increase to 130 MW or 27% of its power requirements in calendar years 2007 through 2010. The Company is reviewing its options for its unpriced energy requirements.

The Company purchases all of the electricity requirements for the Ravenswood facility from Ohio Power Company, a unit of American Electric Power Company, under a fixed price power supply agreement that runs through December 31, 2005. Under a new power contract approved by the Public Services Commission of West Virginia, Appalachian Power Company has agreed to supply power to the Ravenswood facility from January 1, 2006 through December 31, 2010; provided that after December 31, 2007, Century Aluminum of West Virginia, Inc. may terminate the agreement by providing 12 months notice of termination. Power delivered under the new power supply agreement will be as set forth in currently published tariffs. Appalachian Power Company filed a rate case on September 26, 2005, seeking increases in its tariff rates. It has advised the Company it expects those rates to become effective July 1, 2006. The Company intends to contest the rate increase.

The Mt. Holly facility purchases all of its power from the South Carolina Public Service Authority at rates established by published schedules. The Mt. Holly facility's current power contract expires December 31, 2015. Power delivered through 2010 will be priced as set forth in currently published schedules, subject to adjustments for fuel costs. Rates for the period 2011 through 2015 will be as provided under then-applicable schedules.

The Nordural facility purchases power from Landsvirkjun, a power company jointly owned by the Republic of Iceland and two Icelandic municipal governments, under a contract due to expire in 2019. The power delivered to the Nordural facility under its current contract is from hydroelectric and geothermal sources, both competitively-priced and renewable sources of power in Iceland, at a rate based on the London Metal Exchange ("LME") price for primary aluminum. Nordural has entered into a power contract with Hitaveita Sujurnesja hf. ("HS") and Orkuveita Reykjavíkur ("OR") for the supply of the additional power required for expanding the plant's production capacity from 90,000 metric tons per year up to 220,000 metric tons per year. In addition, OR has conditionally agreed to supply the power required to further expand the plant's production capacity to 260,000 metric tons per year by late 2008. Power under these agreements will be generated from geothermal resources and prices will be LME-based. By the terms of a Second Amendment to the Landsvirkjun/Nordural Power Contract, dated as of April 21, 2004, Landsvirkjun has agreed on a best commercial efforts basis to provide backup power to Nordural should HS or OR be unable to meet the obligations of their contract to provide power for the Nordural expansion.

Labor Commitments

Approximately 81% of the Company's U.S. based work force are represented by the United Steelworker's of America (the "USWA") and are working under agreements that expire as follows: March 31, 2006 (Hawesville) and May 31, 2006 (Ravenswood).

Approximately 80% of Nordural's work force are represented by six national labor unions under an agreement that expires on December 31, 2009.

Other Commitments and Contingencies

The Company's income tax returns are periodically examined by various tax authorities. The Company is currently under audit by the Internal Revenue Service ("IRS") for the tax years through 2002. In connection with such examinations, the IRS has raised issues and proposed tax deficiencies. The Company is reviewing the issues raised by the IRS and plans to contest the proposed tax deficiencies. Based on current information, the Company does not believe that the outcome of the tax audit will have a material impact on the Company's financial condition or results of operations.

At September 30, 2005 and December 31, 2004, the Company had outstanding capital commitments related to the Nordural expansion of \$149,376 and \$218,800, respectively. The Company's cost commitments for the Nordural expansion may materially change depending on the exchange rate between the U.S. dollar and certain foreign currencies, principally the euro and the Icelandic krona.

9. Forward Delivery Contracts and Financial Instruments

As a producer of primary aluminum products, the Company is exposed to fluctuating raw material and primary aluminum prices. The Company routinely enters into fixed and market priced contracts for the sale of primary aluminum and the purchase of raw materials in future periods.

Primary Aluminum Sales Contracts

Contract	Customer	Volume	Term	Pricing
Pechiney Metal Agreement (1)	Pechiney	125,192 to 146,964 metric tons per year ("mtpy")	Through July 31, 2007	Based on U.S. Midwest market
Glencore Metal Agreement I (2)	Glencore	50,000 mtpy	Through December 31, 2009	LME-based
Glencore Metal Agreement II (3)	Glencore	20,000 mtpy	Through December 31, 2013	Based on U.S. Midwest market
Southwire Metal Agreement (4)	Southwire	108,862 mtpy (high purity molten aluminum)	Through March 31, 2011	Based on U.S. Midwest market
		27,216 mtpy (standard-grade molten aluminum)	Through December 31, 2008	Based on U.S. Midwest market

- (1) Pechiney has the right, upon 12 months notice, to reduce its purchase obligations by 50% under this contract.
- (2) Referred to as the "New Sales Contract" in the Company's 2004 Annual Report on Form 10-K. The Company accounts for the Glencore Metal Agreement I as a derivative instrument under SFAS No. 133. The Company has not designated the Glencore Metal Agreement I as "normal" because it replaced and substituted for a significant portion of a sales contract which did not qualify for this designation. Because the Glencore Metal Agreement I is variably priced, the Company does not expect significant variability in its fair value, other than changes that might result from the absence of the U.S. Midwest premium.
- (3) Referred to as the "Glencore Metal Agreement" in the Company's 2004 Annual Report on Form 10-K. The Company accounts for the Glencore Metal Agreement II as a derivative instrument under SFAS No. 133. Under the Glencore Metal Agreement II, pricing is based on then-current market prices, adjusted by a negotiated U.S. Midwest premium with a cap and a floor as applied to the current U.S. Midwest premium.
- (4) The Southwire Metal Agreement will automatically renew for additional five-year terms, unless either party provides 12 months notice that it has elected not to renew.

Tolling Contracts

Contract	Customer	Volume	Term	Pricing
Billiton Tolling Agreement (1)(2)	BHP Billiton	130,000 mtpy	Through December 31, 2013	LME-based
Glencore Tolling	Glencore	90,000 mtpy	Through July 2016	LME-based

⁽¹⁾ Substantially all of Nordural's existing sales consist of tolling revenues earned under a long-term Alumina Supply, Toll Conversion and Aluminum Metal Supply Agreement (the "Billiton Tolling Agreement") between

- Nordural and a subsidiary of BHP Billiton Ltd (together with its subsidiaries, "BHP Billiton"). Under the Billiton Tolling Agreement, Nordural receives an LME-based fee for the conversion of alumina, supplied by BHP Billiton, into primary aluminum.
- (2) In September 2005, Nordural and BHP Billiton amended the Billiton Tolling Agreement to increase the tolling arrangement from 90,000 metric tons to 130,000 metric tons of the per annum production capacity at the Nordural facility effective upon the completion of the expansion.
- (3) Nordural entered into a 10-year LME-based alumina tolling agreement with a subsidiary of Glencore International AG (together with its subsidiaries, "Glencore") for 90,000 metric tons of the expansion capacity at the Nordural facility. The term of the agreement will begin upon completion of the expansion, which is expected to be in late-2006.

Apart from the Pechiney Metal Agreement, Glencore Metal Agreement I, Glencore Metal Agreement II and Southwire Metal Agreement, the Company had forward delivery contracts to sell 93,533 metric tons and 113,126 metric tons of primary aluminum at September 30, 2005 and December 31, 2004, respectively. Of these forward delivery contracts, the Company had fixed price commitments to sell 7,522 metric tons and 6,033 metric tons of primary aluminum at September 30, 2005 and December 31, 2004, respectively, of which none were with Glencore.

Alumina Supply Agreements

The Company is party to long-term agreements with Glencore that supply a fixed quantity of alumina to the Company's Ravenswood and Mt. Holly facilities at prices indexed to the price of primary aluminum quoted on the LME. In addition, as part of the Company's acquisition of a joint venture interest in the Gramercy, Louisiana alumina refinery, the Company entered into a long-term agreement on November 2, 2004 with Gramercy Alumina LLC ("Gramercy Alumina") that supplies a fixed quantity of alumina to the Company's Hawesville facility at prices based on the alumina production costs at the Gramercy refinery. Gramercy Alumina, a joint venture company owned 50/50 by Century and Noranda Finance Inc., owns and operates the Gramercy alumina refinery. A summary of these agreements is provided below. The Company is reviewing options for future supplies of alumina. The Company's Nordural facility toll converts alumina provided by BHP Billiton, and will toll convert alumina provided by Glencore beginning in 2006 upon completion of the current expansion of the Nordural facility.

Facility	Supplier	Term	Pricing
Ravenswood	Glencore	Through December 31, 2006	LME-based
Mt. Holly	Glencore	Through December 31, 2006 (54% of requirement)	LME-based
Mt. Holly	Glencore	Through January 31, 2008 (46% of requirement)	LME-based
Hawesville	Gramercy Alumina	Through December 31, 2010	Cost-based

Anode Purchase Agreement

Nordural has a contract for the supply of anodes for its existing capacity which expires in 2013. Pricing for the anode contract is variable and is indexed to the raw material market for petroleum coke products, certain labor rates, and maintenance cost indices.

On September 30, 2005, Nordural and a subsidiary of Alcan Inc. ("Alcan") entered into an agreement for the supply of anodes for its planned expansion capacity. The term of the agreement is through December 31, 2015,

however after September 30, 2010 either party has the right upon 18 months notice to terminate the agreement. Pricing for the anode contract is variable and is updated and adjusted quarterly. Pricing is determined by reference to prices for petroleum coke products and energy costs, and certain wage and price indices.

Financial Sales Agreements

To mitigate the volatility in its unpriced forward delivery contracts, the Company enters into fixed price financial sales contracts, which settle in cash in the period corresponding to the intended delivery dates of the forward delivery contracts. Certain of these fixed price financial sales contracts are accounted for as cash flow hedges depending on the Company's designation of each contract at its inception.

Primary Aluminum Fixed Price Financial Sales Contracts as of:

		(Metric Tons)						
		September 30, 2005			December 31, 2004			
	Cash Flow Hedges	Derivatives	Total	Cash Flow Hedges	Derivatives	Total		
2005	51,750	_	51,750	193,083	_	193,083		
2006	142,750	25,200	167,950	142,750	25,200	167,950		
2007	119,500	50,400	169,900	119,500	50,400	169,900		
2008	9,000	100,200	109,200	9,000	75,000	84,000		
2009	_	105,000	105,000	_	75,000	75,000		
2010-2015		480,000	480,000		75,000	75,000		
Total	323,000	760,800	1,083,800	464,333	300,600	764,933		

The contracts accounted for as derivatives contain clauses that trigger additional shipment volume when the market price for a contract month is above the contract ceiling price. If the market price exceeds the ceiling price for all contract months through 2015, the maximum additional shipment volume would be 760,800 metric tons. These contracts will be settled monthly. The Company had no fixed price financial contracts to purchase aluminum at September 30, 2005 or December 31, 2004.

Additionally, to mitigate the volatility of the natural gas markets, the Company enters into fixed price financial purchase contracts, accounted for as cash flow hedges, which settle in cash in the period corresponding to the intended usage of natural gas.

Natural Gas Fixed Price Financial Purchase Contracts as of:

	(Thousand September 30, 2005	s of DTH) December 31, 2004
2005	970	2,880
2006	1,680	480
2007	780	480
2008	480	480
Total	3,910	4,320

Based on the fair value of the Company's fixed price financial sales contracts for primary aluminum and financial purchase contracts for natural gas that qualify as cash flow hedges as of September 30, 2005, accumulated other comprehensive loss of \$14,553 is expected to be reclassified as a reduction to earnings over the next 12 month period.

The forward financial sales and purchase contracts are subject to the risk of non-performance by the counterparties. However, the Company only enters into forward financial contracts with counterparties it determines to be creditworthy. If any counterparty failed to perform according to the terms of the contract, the accounting impact would be limited to the difference between the contract price and the market price applied to the contract volume on the date of settlement.

10. Supplemental Cash Flow Information

		onths ended ember 30,
	2005	2004
Cash paid for:		
Interest	\$ 27,098	\$ 36,152
Income tax	12,627	198
Cash received for:		
Interest	893	843
Income tax refunds	_	135
Non-cash investing activities:		
Accrued Nordural expansion costs	6,311	_

11. Asset Retirement Obligations

The reconciliation of the changes in the asset retirement obligations is as follows:

	For the	Nine months	
	ended S		For the Year ended
		2005	December 31, 2004
Beginning balance, ARO liability	\$	17,232	\$ 16,495
Additional ARO liability incurred		1,354	1,383
ARO liabilities settled		(2,515)	(3,379)
Accretion expense		1,045	2,733
Ending balance, ARO liability	\$	17,116	\$ 17,232

12. New Accounting Standards

In May 2005, the FASB issued SFAS No. 154, "Accounting Changes and Error Corrections." This Statement replaces the guidance in APB Opinion No. 20, "Accounting Changes" and FASB Statement No. 3, "Reporting Accounting Changes in Interim Financial Statements." The Statement provides guidance on the accounting for and reporting of accounting changes and error corrections. It requires retrospective application as the required method for reporting a change in accounting principle, unless impracticable. The Statement differentiates retrospective application for changes in accounting principle and changes in reporting entity from restatement for corrections of errors. In addition, the reporting of a correction of an error by restating previously issued financial statements is also addressed by this Statement. The Statement is effective for fiscal year 2006 and thereafter.

In December 2004, the Financial Accounting Standards Board ("FASB") issued SFAS No. 123(R), "Share Based Payment." This Statement is a revision of FASB Statement No. 123, "Accounting for Stock-Based Compensation" and supersedes Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees." This statement focuses primarily on accounting for transactions in which a company obtains services in share-based payment transactions. This Statement will require the Company to recognize the grant date fair value of an award of equity-based instruments to employees and the cost will be recognized over the period in which the employees are required to provide service. The Statement is effective for fiscal year 2006 and thereafter. The Company is currently assessing the Statement and does not expect the impact of adopting SFAS No. 123(R) will have a material effect on the Company's financial position and results of operations.

In November 2004, the FASB issued SFAS No. 151, "Inventory Costs." This Statement amends the guidance in Accounting Research Bulletin No. 43, Chapter 4, "Inventory Pricing" to clarify the accounting treatment for certain inventory costs. In addition, the Statement requires that the allocation of production overheads be based on the facilities' normal production capacity. The Statement is effective for fiscal year 2006 and thereafter. The Company is currently assessing the Statement and has not yet determined the impact of adopting SFAS No. 151 on the Company's financial position and results of operations.

13. Comprehensive Income and Accumulated Other Comprehensive Income (Loss)

Comprehensive Income:

	Nine mon Septem	
	2005	2004
		(Restated)
Net income	\$ 32,403	\$ 8,839
Other comprehensive income (loss):		
Net unrealized loss on financial instruments, net of tax of \$93 and \$13,806, respectively	(410)	(24,229)
Net amount reclassified to income, net of tax of (\$11,062) and (\$1,306), respectively	19,208	2,349
Comprehensive income (loss)	\$51,201	\$(13,041)

Composition of Accumulated Other Comprehensive Loss:

	Sep	otember 30, 2005	De	cember 31, 2004
Net unrealized loss on financial instruments, net of tax of \$17,042 and \$28,011	\$	(30,315)	\$	(49,113)
Minimum pension liability adjustment, net of tax of \$1,728 and \$1,728		(3,073)		(3,073)
Total accumulated other comprehensive loss	\$	(33,388)	\$	(52,186)
14 Fornings Por Shore			_	

14. Earnings Per Share

The following table provides a reconciliation of the computation of the basic and diluted earnings per share:

	Three months ended September 30,					
	Income	2005 Shares	Per-Share	Income	2004 Shares	Per-Share
						<u></u>
				(Restated)		
Net loss	\$(20,071)			\$(15,973)		
Less: Preferred stock dividends	_			_		
Basic EPS:						
Loss applicable to common shareholders	(20,071)	32,162	\$ (0.62)	(15,973)	31,754	\$ (0.50)
Diluted EPS:						
Loss applicable to common shareholders with assumed conversions	<u>\$(20,071</u>)	32,162	<u>\$ (0.62)</u>	<u>\$(15,973</u>)	31,754	<u>\$ (0.50)</u>
			Nine months end	ed September 30,		
	Income	2005 Shares	Per-Share	Income	2004 Shares	Per-Share
	Income	Shares	rei-Silare	Income	Shares	rei-snate
				(Restated)		
Net income	\$ 32,403			\$ 8,839		
Less: Preferred stock dividends	_			(769)		
Basic EPS:						
Income applicable to common shareholders	32,403	32,120	\$ 1.01	8,070	27,542	\$ 0.29
Effect of Dilutive Securities:						
Plus: Incremental shares	_	43		_	117	
Diluted EPS:						
Income applicable to common shareholders with assumed conversions	<u>\$ 32,403</u>	<u>32,163</u> 15	<u>\$ 1.01</u>	<u>\$ 8,070</u>	27,659	\$ 0.29

Options to purchase 299,413 and 313,179 shares of common stock were outstanding during the periods ended September 30, 2005 and 2004, respectively. For the three month periods ending September 30, 2004 and 2005 all options were excluded from the calculation of diluted EPS because the Company has a net loss for those periods. For the nine month periods ending September 30, 2005 and 2004, 31,000 and 10,000 options, respectively, were not included in the calculation of diluted EPS because the option's exercise price exceeded the average market price of the common stock.

15. Components of Net Periodic Benefit Cost

	Pension Benefits			
	Three months ended		Nine mon Septem	
	September 30, 2005 2004		2005	2004
Service cost	\$ 980	\$ 846	\$ 2,941	\$ 2,524
Interest cost	1,171	1,066	3,512	3,195
Expected return on plan assets	(1,475)	(1,187)	(4,425)	(3,563)
Amortization of prior service cost	741	210	2,222	631
Amortization of net gain	157	81	<u>471</u>	244
Net periodic benefit cost	\$ 1,574	\$ 1,016	\$ 4,721	\$ 3,031
		<u> </u>		<u> </u>
		Other Postemplo	yment Benefits	
	Three mon		Nine mon	
	Septem		Septem	
	2005	2004	2005	2004
Service cost	\$ 1,258	\$ 890	\$ 3,774	\$ 3,192
Interest cost	2,219	1,672	6,658	5,663
Expected return on plan assets	_	_	<u> </u>	_
Amortization of prior service cost	(219)	(84)	(658)	(253)
Amortization of net gain	929	299	2,786	1,532
Net periodic benefit cost	\$ 4,187	\$ 2,777	\$ 12,560	\$ 10,134

16. Condensed Consolidating Financial Information

The Company's 7.5% Senior Notes due 2014, and 1.75% Convertible Senior Notes due 2024 are guaranteed by each of the Company's material existing and future domestic subsidiaries. These notes are not guaranteed by the Company's foreign subsidiaries (the "Non-Guarantor Subsidiaries"). During the second quarter of 2005, Century Aluminum of Kentucky, LLC (the "LLC") became a guarantor subsidiary. In the periods presented prior to the current reporting period, the LLC was classified with the Non-Guarantor Subsidiaries. The Company's policy for financial reporting purposes is to allocate corporate expenses or income to subsidiaries. For the three months ended September 30, 2005 and 2004, the Company allocated total corporate expenses of \$53,017 and \$48,274 to its subsidiaries, respectively. For the nine months ended September 30, 2005 and 2004, the Company allocated total corporate expenses of \$51,030 and \$48,330 to its subsidiaries, respectively. Additionally, the Company charges interest on certain intercompany balances.

CENTURY ALUMINUM COMPANY Notes to Consolidated Financial Statements — (continued)

The following summarized condensed consolidating balance sheets as of September 30, 2005 and December 31, 2004, condensed consolidating statements of operations for the three and nine months ended September 30, 2005 and September 30, 2004 and the condensed consolidating statements of cash flows for the nine months ended September 30, 2005 and September 30, 2004 present separate results for Century Aluminum Company, the Guarantor Subsidiaries and the Non-Guarantor Subsidiaries.

This summarized condensed consolidating financial information may not necessarily be indicative of the results of operations or financial position had the Company, the Guarantor Subsidiaries or the Non-Guarantor Subsidiaries operated as independent entities.

CONDENSED CONSOLIDATING BALANCE SHEET As of September 30, 2005

	Combined Guarantor Subsidiaries	Non-	ombined -Guarantor bsidiaries	The Company	Reclassifications and Eliminations	Consolidated
Assets:						
Cash and cash equivalents	\$ 17,514	\$	5,591	\$ 32,742	\$ —	\$ 55,847
Restricted cash	2,028		_	_	_	2,028
Accounts receivables – net	69,142		11,368	_	_	80,510
Due from affiliates	212,516		_	674,773	(869,672)	17,617
Inventories	87,608		15,795	_	2,805	106,208
Prepaid and other current assets	13,546		4,066	4,736	_	22,348
Deferred taxes — current portion	11,296			2,998		14,294
Total current assets	413,650		36,820	715,249	(866,867)	298,852
Investment in subsidiaries	13,823		_	339,104	(352,927)	_
Property, plant and equipment – net	459,260		535,641	335	_	995,236
Intangible asset – net	78,316		_	_	_	78,316
Goodwill	_		94,844	_	_	94,844
Deferred taxes — less current portion	_		_	33,555	(33,555)	_
Other assets	49,520		11,412	18,802		79,734
Total assets	\$1,014,569	\$	678,717	\$1,107,045	\$ (1,253,349)	\$1,546,982
		-				
Liabilities and shareholders' equity:						
Accounts payable – trade	\$ 32,858	\$	27,324	\$ —	\$ —	\$ 60,182
Due to affiliates	96,747		46,093	156,004	(220,453)	78,391
Industrial revenue bonds	7,815		_	_		7,815
Accrued and other current liabilities	8,894		3,159	24,817	_	36,870
Long-term debt — current portion	_		558	_	_	558
Accrued employee benefits costs – current portion	8,458		_	_	_	8,458
Convertible senior notes				175,000		175,000
Total current liabilities	154,772		77,134	355,821	(220,453)	367,274
Senior unsecured notes payable	_			250,000	_	250,000
Nordural debt	_		196,601	_	_	196,601
Accrued pension benefits costs – less current portion	_		_	13,421	_	13,421
Accrued post retirement benefits costs – less current						
portion	93,172		_	894	_	94,066
Other liabilities/intercompany loan	395,010		286,188	_	(647,908)	33,290
Due to affiliates – less current portion	30,936		_	47,799	_	78,735
Deferred taxes – less current portion	89,272		17,274		(32,061)	74,485
Total non-current liabilities	608,390		500,063	312,114	(679,969)	740,598
Shareholders' equity:						
Common stock	60		12	322	(72)	322
Additional paid-in capital	252,734		75,190	418,876	(327,924)	418,876
Accumulated other comprehensive income (loss)	(33,388)			(33,388)	33,388	(33,388)
Retained earnings (accumulated deficit)	32,001		26,318	53,300	(58,319)	53,300
Total shareholders' equity	251,407		101,520	439,110	(352,927)	439,110
Total liabilities and equity	\$1,014,569	\$	678,717	\$1,107,045	\$ (1,253,349)	\$1,546,982
	+1,011,000	4	5.5,717	+1,101,010	+ (1,200,01)	+1,0.0,000

CONDENSED CONSOLIDATING BALANCE SHEET

As of December 31, 2004 (Restated)

	Combined Guarantor Subsidiaries	Combined Non-Guarantor Subsidiaries	The Company	Reclassifications And Eliminations	Consolidated
Assets:	<u></u>				
Cash and cash equivalents	\$ 185	\$ 1,759	\$ 42,224	\$ —	\$ 44,168
Restricted cash	1,174	504	_	_	1,678
Accounts receivable — net	71,051	8,449	76	_	79,576
Due from affiliates	168,328	8,474	684,458	(846,889)	14,371
Inventories	73,515	38,688		(919)	111,284
Prepaid and other assets	1,514	4,299	4,242	_	10,055
Deferred taxes — current portion	24,018	293		331	24,642
Total current assets	339,785	62,466	731,000	(847,477)	285,774
Investment in subsidiaries	66,393	_	270,178	(336,571)	_
Property, plant and equipment — net	464,418	341,692	140	_	806,250
Intangible asset — net	_	86,809	_	_	86,809
Goodwill	_	95,610	_	_	95,610
Other assets	20,391	16,792	20,927		58,110
Total assets	\$ 890,987	\$ 603,369	\$1,022,245	\$ (1,184,048)	\$1,332,553
Liabilities and shareholders' equity:					
Accounts payable – trade	\$ 12,000	\$ 35,479	\$ —	\$ —	\$ 47,479
Due to affiliates	84,151	2,499	162,150	(163,985)	84,815
Industrial revenue bonds	7,815			_	7,815
Accrued and other current liabilities	15,545	10,023	27,741		53,309
Long term debt — current portion		704	9,878	_	10,582
Accrued employee benefits costs - current portion	6,507	1,951		_	8,458
Convertible senior notes			175,000		175,000
Total current liabilities	126,018	50,656	374,769	(163,985)	387,458
Senior unsecured notes payable	_	_	250,000	_	250,000
Nordural debt	_	80,711	_	_	80,711
Accrued pension benefit costs — less current portion	_	_	10,685	_	10,685
Accrued postretirement benefit costs - less current					
portion	56,947	27,812	790		85,549
Other liabilities/intercompany loan	479,213	239,124	_	(683,376)	34,961
Due to affiliates — less current portion	30,416	10.270	1.501	<u> </u>	30,416
Deferred taxes	47,509	19,379	1,501	(116)	68,273
Total noncurrent liabilities	614,085	367,026	262,976	(683,492)	560,595
Shareholders' equity:					
Common stock	59	13	320	(72)	320
Additional paid-in capital	188,424	242,818	415,453	(431,242)	415,453
Accumulated other comprehensive income (loss)	(51,665)	(521)	(52,186)	52,186	(52,186)
Retained earnings (accumulated deficit)	14,066	(56,623)	20,913	42,557	20,913
Total shareholders' equity	150,884	185,687	384,500	(336,571)	384,500
Total liabilities and shareholders' equity	\$ 890,987	\$ 603,369	\$1,022,245	\$ (1,184,048)	\$1,332,553

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS For the Three months ended September 30, 2005

	Combined Guarantor Subsidiaries	Combined Non-Guarantor Subsidiaries	The Company	Reclassifications and Eliminations	Consolidated
Net sales:					
Third-party customers	\$ 189,456	\$ 33,355	\$ —	\$ —	\$ 222,811
Related parties	48,025				48,025
	237,481	33,355	_	_	270,836
Cost of goods sold	217,924	23,345	_	(491)	240,778
Gross profit	19,557	10,010		491	30,058
Selling, general and administrative expenses	7,904	200			8,104
Operating income	11,653	9,810	_	491	21,954
Interest expense – third party	(6,158)	(55)	_	_	(6,213)
Interest income (expense) – affiliates	6,283	(6,283)		_	
Interest income	446	150	_	_	596
Net loss on forward contracts	(53,481)	_	_	_	(53,481)
Other income (expense), net	86	(153)			(67)
Income (loss) before income taxes and equity in					
earnings (loss) of subsidiaries and joint ventures	(41,171)	3,469		491	(37,211
Income tax (expense) benefit	14,366	(125)		(177)	14,064
Income (loss) before equity in earnings (loss) of					
subsidiaries and joint ventures	(26,805)	3,344	_	314	(23,147)
Equity in earnings (loss) of subsidiaries and joint					
ventures	1,316	1,760	(20,071)	20,071	3,076
Net income (loss)	\$ (25,489)	\$ 5,104	<u>\$(20,071)</u>	\$ 20,385	\$ (20,071)

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS For the Three Months Ended September 30, 2004 (Restated)

	Combined Guarantor Subsidiaries	Combined Non-Guarantor Subsidiaries	The Company	Reclassifications and Eliminations	Consolidated
Net sales:					
Third-party customers.	\$ 200,407	\$ 31,095	\$ —	\$ —	\$ 231,502
Related parties	42,815	<u></u>		<u> </u>	42,815
	243,222	31,095	_	_	274,317
Cost of goods sold	206,271	111,624	_	(87,060)	230,835
Reimbursement from owner		(87,101)		87,101	
Gross profit (loss)	36,951	6,572	_	(41)	43,482
Selling, general and administrative expenses	7,567	<u> </u>			7,567
Operating income (loss)	29,384	6,572	_	(41)	35,915
Interest expense – third party	(6,142)	(4,410)	_	_	(10,552)
Interest income	370	118	_	29	517
Net loss on forward contracts	(3,149)		_	_	(3,149)
Loss on early extinguishment of debt	(47,448)	_	_	_	(47,448)
Other income (expense), net	2	(125)		13	(110)
Income (loss) before income taxes and equity in					
earnings (loss) of subsidiaries	(26,983)	2,155	_	1	(24,827)
Income tax (expense) benefit	9,488	(1,806)		1,172	8,854
Income (loss) before equity in earnings (loss) of					
subsidiaries	(17,495)	349	_	1,173	(15,973)
Equity in earnings (loss) of subsidiaries	(1,912)		(15,973)	17,885	
Net income (loss)	<u>\$ (19,407)</u>	\$ 349	<u>\$(15,973)</u>	\$ 19,058	\$ (15,973)

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS For the Nine months ended September 30, 2005

	Combined Guarantor	Combined Non-Guarantor Subsidiaries	The	Reclassifications and Eliminations	Consolidated
Net sales:	<u>Subsidiaries</u>	Subsidiaries	Company	Eliminations	Consondated
Third-party customers	\$ 612,045	\$ 101,520	\$ —	\$ —	\$ 713,565
Related parties	125,923				125,923
	737,968	101,520			839,488
Cost of goods sold	646,270	71,444		(5,199)	712,515
Gross profit	91,698	30,076		5,199	126,973
Selling, general and administrative expenses	24,746	200			24,946
Operating income	66,952	29,876		5,199	102,027
Interest expense – third party	(18,811)	(602)	_	_	(19,413)
Interest income (expense) – affiliates	17,616	(17,616)		_	_
Interest income	864	224	_	_	1,088
Net loss on forward contracts	(52,480)	_	_	_	(52,480)
Loss on early extinguishment of debt	(835)	_	_	_	(835)
Other income (expense), net	34	669			703
Income before income taxes and equity in earnings					
(loss) of subsidiaries and joint ventures	13,340	12,551	_	5,199	31,090
Income tax expense	(7,422)	(2,716)		(1,872)	(12,010)
Income before equity in earnings (loss) of subsidiaries	5,918	9,835	_	3,327	19,080
Equity in earnings (loss) of subsidiaries and joint					
ventures	6,166	7,157	32,403	(32,403)	13,323
Net income (loss)	\$ 12,084	\$ 16,992	\$32,403	\$ (29,076)	\$ 32,403

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS For the Nine months ended September 30, 2004 (Restated)

	Combined Guarantor	Combined Non-Guarantor	The	Reclassifications and	
	Subsidiaries	Subsidiaries	<u>Company</u>	Eliminations	Consolidated
Net sales:					
Third-party customers	\$ 596,700	\$ 52,578	\$ —	\$ —	\$ 649,278
Related parties	120,866				120,866
	717,566	52,578	_	_	770,144
Cost of goods sold	596,377	295,180	_	(249,927)	641,630
Reimbursement from owners		(250,042)		250,042	
Gross profit (loss)	121,189	7,440	_	(115)	128,514
Selling, general and administrative expenses	16,966				16,966
Operating income (loss)	104,223	7,440	_	(115)	111,548
Interest expense — third party	(25,053)	(7,255)	_	_	(32,308)
Interest expense – related party	(380)	_	_	_	(380)
Interest income	627	140	_	81	848
Net loss on forward contracts	(17,146)	_	_	_	(17,146)
Loss on early extinguishment of debt	(47,448)	_		_	(47,448)
Other income (expense), net	(679)	(152)		33	(798)
Income (loss) before income taxes and equity in					
earnings (loss) of subsidiaries	14,144	173	_	(1)	14,316
Income tax (expense) benefit	(5,740)	(3,250)	_	3,513	(5,477)
Equity in earnings (loss) of subsidiaries	(5,733)		8,839	(3,106)	
Net income (loss)	\$ 2,671	\$ (3,077)	\$ 8,839	\$ 406	\$ 8,839

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS For the Nine Months Ended September 30, 2005

	Combined Guarantor Subsidiaries	Combined Non-guarantor Subsidiaries	The <u>Company</u>	Consolidated
Net cash provided by operating activities	\$ 45,562	\$ 81,662	<u>\$</u>	\$ 127,224
Investing activities:				
Nordural expansion	_	(200,641)	_	(200,641)
Purchase of property, plant and equipment, net	(7,689)	(1,604)	(336)	(9,629)
Business acquisitions, net of cash acquired	_	_	(7,000)	(7,000)
Restricted cash deposits	(350)	_	_	(350)
Proceeds from sale of property, plant and equipment	48	53		101
Net cash used in investing activities	(7,991)	(202,192)	(7,336)	(217,519)
Financing activities:				
Borrowings	_	188,937	_	188,937
Repayment of debt	_	(73,193)	(9,945)	(83,138)
Financing fees	_	(4,617)	(515)	(5,132)
Intercompany transactions	(20,242)	13,235	7,007	_
Dividends	_	_	(16)	(16)
Issuance of common stock			1,323	1,323
Net cash provided by (used in) financing activities	(20,242)	124,362	(2,146)	101,974
Net increase (decrease) in cash and cash equivalents	17,329	3,832	(9,482)	11,679
Cash and cash equivalents, beginning of period	185	1,759	42,224	44,168
Cash and cash equivalents, end of period	\$ 17,514	\$ 5,591	\$32,742	\$ 55,847

Notes to Consolidated Financial Statements — (continued)

CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS For the Nine months ended September 30, 2004 (Restated)

	Combined Guarantor Subsidiaries	Combined Non-Guarantor Subsidiaries	The Company	Reclassifications and Eliminations	Consolidated
Net cash provided by (used in) operating activities	\$ (16,952)	\$ 88,816	<u>\$</u>	<u> </u>	\$ 71,864
Investing activities					
Purchase of property, plant and equipment — net	(5,437)	(3,395)	_	_	(8,832)
Nordural expansion	_	(17,482)	_	_	(17,482)
Acquisitions, net of cash acquired	_	_	(184,869)	_	(184,869)
Net cash used in investing activities	(5,437)	(20,877)	(184,869)		(211,183)
Financing activities:					
Borrowings	_	569	425,000	_	425,569
Repayment of debt – third party	_	(107,791)	(315,055)	_	(422,846)
Repayment of debt – related party	_	_	(14,000)	_	(14,000)
Financing fees	_	_	(12,805)	_	(12,805)
Dividends	_	_	(3,311)	_	(3,311)
Intercompany transactions	22,285	69,197	(91,482)	_	_
Issuance of common stock		<u></u>	214,982		214,982
Net cash provided by (used in) financing activities	22,285	(38,025)	203,329	_	187,589
Net increase (decrease) in cash	(104)	29,914	18,460		48,270
Cash, beginning of period	104	· —	28,100	_	28,204
Cash, end of period	<u> </u>	\$ 29,914	\$ 46,560	<u> </u>	\$ 76,474

17. Subsequent Event

On November 9, 2005, the Company announced that Logan W. Kruger will succeed Craig Davis as President and Chief Executive Officer. Craig Davis will continue to serve as Chairman of the Board of Directors.

FORWARD-LOOKING STATEMENTS – CAUTIONARY STATEMENT UNDER THE PRIVATE SECURITIES REFORM ACT OF 1995.

This Quarterly Report on Form 10-Q contains forward-looking statements. The Company has based these forward-looking statements on current expectations and projections about future events. Many of these statements may be identified by the use of forward-looking words such as "expects," "anticipates," "plans," "believes," "projects," "estimates," "intends," "should," "could," "would," and "potential" and similar words. These forward-looking statements are subject to risks, uncertainties and assumptions including, among other things, those discussed under Part I, Item 2, "Management's Discussion and Analysis of Financial Condition and Results of Operations," and Part I, Item 1, "Financial Statements and Supplementary Data," and:

- The Company's high level of indebtedness reduces cash available for other purposes, such as the payment of dividends, and limits the Company's ability to incur additional debt and pursue its growth strategy;
- The cyclical nature of the aluminum industry causes variability in the Company's earnings and cash flows;
- The loss of a customer to whom the Company delivers molten aluminum would increase the Company's production costs;
- Glencore International AG owns a large percentage of the Company's common stock and has the ability to influence matters requiring shareholder approval;
- The Company could suffer losses due to a temporary or prolonged interruption of the supply of electrical power to its facilities, which can be caused by unusually high demand, blackouts, equipment failure, natural disasters or other catastrophic events;
- Due to volatile prices for alumina and electricity, the principal cost components of primary aluminum production, the Company's production costs could be materially impacted if the Company experiences changes to or disruptions in its current alumina or power supply arrangements, or if production costs at the Company's alumina refining operations increase significantly or if the Company is unable to obtain affordable power for those portions of its power requirements that are currently unpriced;
- By expanding the Company's geographic presence and diversifying its operations through the acquisition of bauxite mining, alumina refining and additional aluminum reduction assets, the Company is exposed to new risks and uncertainties that could adversely affect the overall profitability of its business;
- Changes in the relative cost of certain raw materials and energy compared to the price of primary aluminum could affect the Company's margins;
- Most of the Company's employees are unionized and any labor dispute or failure to successfully renegotiate an existing labor agreement could materially impair the Company's ability to conduct its production operations at its unionized facilities;
- The Company is subject to a variety of environmental laws that could result in unanticipated costs or liabilities;
- The Company may not realize the expected benefits of its growth strategy if it is unable to successfully integrate the businesses it acquires; and
- The Company cannot guarantee that the Company's subsidiary Nordural will be able to complete its expansion in the time forecast or without significant cost overruns or that the Company will be able to realize the expected benefits of the expansion.

Although the Company believes the expectations reflected in its forward-looking statements are reasonable, the Company cannot guarantee its future performance or results of operations. All forward-looking statements in this filing are based on information available to the Company on the date of this filing; however, the Company is not obligated to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. When reading any forward-looking statements in this filing, the reader should consider the risks described above and elsewhere in this report as well as those described in the Company's Annual Report on Form

10-K for the year ended December 31, 2004. Given these uncertainties and risks, the reader should not place undue reliance on these forward-looking statements.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Results of Operations

The following discussion reflects Century's historical results of operations, which do not include results for the Nordural facility until it was acquired in April 2004 and the Company's equity interest in the earnings of Gramercy Alumina LLC ("GAL") and St. Ann Bauxite Limited ("SABL") until the Company acquired a 50% joint venture interest in those companies in October 2004. All periods have been restated to reflect the Company's change in inventory valuation during the second quarter of 2005.

Century's financial highlights include:

	Three months ended September 30,		Nine months ended September 30,		
	2005	2004	2005	2004	
		(In thousands, except per share data)			
		(Restated)		(Restated)	
Net sales:					
Third-party customers	\$222,811	\$231,502	\$713,565	\$649,278	
Related party customers	48,025	42,815	125,923	120,866	
Total	\$270,836	\$274,317	\$839,488	\$770,144	
					
Net income (loss)	\$ (20,071)	\$ (15,973)	\$ 32,403	\$ 8,839	
Net income (loss) applicable to common shareholders	\$ (20,071)	\$ (15,973)	\$ 32,403	\$ 8,070	
Earnings per common share:					
Basic	\$ (0.62)	\$ (0.50)	\$ 1.01	\$ 0.29	
Diluted	\$ (0.62)	\$ (0.50)	\$ 1.01	\$ 0.29	

Net sales: Net sales for the three months ended September 30, 2005 decreased \$3.5 million to \$270.8 million. Reduced shipment volumes of 6.9 million pounds in the current period, which were primarily due to a reduced pot count at the Hawesville facility resulted in a loss of \$6.0 million in net sales compared to the previous year period. Improved price realizations for primary aluminum in the third quarter 2005, \$2.5 million, due to higher London Metal Exchange ("LME") prices which were partially offset by lower Midwest premiums for primary aluminum sales in the current period, partially offset the loss in net sales caused by lower shipment volumes.

Net sales for the nine months ended September 30, 2005 increased \$69.3 million or 9%, to \$839.5 million. Higher price realizations for primary aluminum in the current period, due to improved LME prices for primary aluminum, contributed an additional \$39.4 million in sales that were partially offset by \$19.0 million in reduced direct shipment revenues. Direct shipments were 23.4 million pounds less than the previous year period due to reduced pot count at the Hawesville facility and fewer days in the first nine months of 2005 versus 2004. The additional volume provided by Nordural for the nine months ended September 30, 2005 contributed \$48.9 million to the September 2005 year to date net sales increase.

Gross profit: Gross profit for the three months ended September 30, 2005 decreased \$13.4 million to \$30.1 million from \$43.5 million for the same period in 2004. Reduced shipment volumes during the third quarter 2005 compared to the third quarter 2004 negatively impacted gross profit by \$1.7 million which was partially offset by improved price realizations net of increased alumina costs by \$0.7 million. Higher net operating costs of \$12.4 million during the current quarter were comprised of higher raw material costs and increased replacement of pot

cells, \$4.3 million; higher power and natural gas costs, \$3.6 million; increased net amortization and depreciation charges, \$1.1 million; and other spending of \$3.4 million.

Gross profit for the nine months ended September 30, 2005 decreased \$1.5 million to \$127.0 million from \$128.5 million, for the same period in 2004. Improved price realizations net of increased alumina costs improved gross profit by \$39.0 million and the net increased shipment volume, a result of the Nordural facility acquisition, contributed \$11.5 million in additional gross profit. Offsetting these gains were \$52.0 million in net cost increases during the current period comprised of: higher raw material costs and increased replacement of pot cells, \$15.1 million; higher power and natural gas costs, \$11.2 million; increased cost of Gramercy alumina, \$10.8 million; increased net amortization and depreciation charges, \$5.3 million; increased pension and other post-employment benefit accruals, \$3.0 million and other increased spending, \$6.6 million.

Selling, general and administrative expenses: Selling, general and administrative expenses for the three months ended September 30, 2005 increased \$0.5 million to \$8.1 million relative to the same period in 2004. Selling, general and administrative expenses for the nine months ended September 30, 2005 increased \$8.0 million to \$24.9 million relative to the same period in 2004. Approximately 64%, or \$5.1 million of the increase, was a result of increased compensation and pension expense, with the remaining increase in expense associated with increased audit, other professional fees and other general expenses. In addition, allowance for bad debts was reduced \$0.6 million in the nine months ended September 30, 2004, reflecting the settlement of a claim.

Net gain/loss on forward contracts: Net loss on forward contracts for the three months ended September 30, 2005 was \$53.5 million as compared to a net loss of \$3.1 million for the same period in 2004. For the nine months ended September 30, 2005, net loss on forward contracts was \$52.5 million as compared to a net loss of \$17.1 million for the same period in 2004. The loss reported for the three and nine months ended September 30, 2005, was primarily a result of mark-to-market losses associated with the Company's long term financial sales contracts with Glencore which do not qualify for cash flow hedge accounting. The loss reported for the three and nine month periods ended September 30, 2004, primarily relates to the early termination of a fixed price forward sales contract with Glencore.

Tax provision: Income tax benefit for the three month period ended September 30, 2005 increased \$5.2 million from the same period in 2004. Income tax expense for the nine month period ended September 30, 2005 increased \$6.5 million from the same period in 2004 due to the changes in income (loss) before income taxes and changes in the equity in earnings of joint ventures which were partially offset by the discontinuance of accrual for United States taxes on Nordural's earnings resulting from a decision made in 2005 that such earnings would remain invested outside the United States indefinitely.

Equity in earnings of joint ventures: Equity in earnings from the Gramercy assets, which were acquired on October 1, 2004, was \$3.1 million and \$13.3 million for the three and nine months ended September 30, 2005, respectively. These earnings represent the Company's share of profits from third party bauxite, hydrate and chemical grade alumina sales.

Liquidity and Capital Resources

The Company's statements of cash flows for the nine months ended September 30, 2005 and 2004 are summarized below:

		nths ended nber 30,
	2005	2004
	(dollars in	thousands)
Net cash provided by operating activities	\$ 127,224	\$ 71,864
Net cash used in investing activities	(217,519)	(211,183)
Net cash provided by financing activities	_ 101,974	187,589
Net increase in cash and cash equivalents	\$ 11,679	\$ 48,270

Net cash from operating activities in the first nine months of 2005 increased \$55.3 million to \$127.2 million from the comparable 2004 period. The increase in net cash provided by operating activities during the first nine months of 2005 was the result of the April 2004 Nordural facility acquisition, and improved market conditions as discussed above.

The Company's net cash used in investing activities for the nine month period ended September 30, 2005 was \$217.5 million, primarily a result of the ongoing expansion of the Nordural facility. The Company's remaining net cash used for investing activities consisted of capital expenditures to maintain and improve plant operations and a payment of \$7.0 million to Southwire in connection with the 2001 acquisition of the Hawesville facility. The Company was required to make post-closing payments of up to \$7.0 million if the LME price exceeded specified levels during any of the seven years following closing. The payment was made in April 2005. During the nine month period ended September 30, 2004, the Company used cash to acquire the Nordural facility, commence the Nordural expansion project and for capital expenditures to maintain and improve plant operations.

Net cash provided by financing activities during the first nine months of 2005 was \$102.0 million as a result of borrowings under Nordural's \$365.0 million senior term loan facility. Amounts borrowed under the term loan facility during the period were used to finance a portion of the costs associated with the ongoing expansion of the Nordural facility. During the nine months ended September 30, 2005, the Company used cash of \$83.0 million to retire the Nordural's senior term loan facility, the senior secured first mortgage notes and debt related to the Landsvirkjun power contract.

Liquidity

The Company's principal sources of liquidity are cash flow from operations, available borrowings under the Company's revolving credit facility and Nordural's term loan facility. The Company believes these sources will provide sufficient liquidity to meet working capital needs, fund capital improvements, and provide for debt service requirements. At September 30, 2005, the Company had borrowing availability of \$95.9 million under its revolving credit facility, subject to customary covenants, with no outstanding borrowings. As of September 30, 2005, the Company had remaining borrowing availability of \$177.0 million under Nordural's \$365.0 million term loan facility.

The Company's principal uses of cash are operating costs, payments of principal and interest on the Company's outstanding debt, the funding of capital expenditures and investments in related businesses, working capital and other general corporate requirements. During 2004, the Company refinanced its public debt obligations and commenced work on the expansion of the Nordural facility, which the Company believes are transactions that may favorably impact the current and future financial condition and results of operations of the Company.

Capital Resources

The Company anticipates capital expenditures of approximately \$15.0 to \$20.0 million in 2005, exclusive of the Nordural expansion. The revolving credit facility limits, under certain circumstances, the Company's ability to make capital expenditures at its U.S. reduction facilities; however, the Company does not expect that the limitations will interfere with its ability to maintain its properties and business and comply with environmental requirements.

The Company has commenced work on an expansion of the Nordural facility that will increase its annual production capacity from 90,000 metric tons to 220,000 metric tons. The construction of the expansion capacity to 220,000 metric tons per year ("mtpy") is projected to be substantially completed by mid to late-2006. Start-up of the final 8,000 mtpy of capacity will depend on the timing of the availability of power. The Company estimates the expansion will cost approximately \$474.0 million. The Company plans to finance the current expansion project through cash flow and borrowings under Nordural's term loan facility, which is non-recourse to Century Aluminum Company.

The Nordural expansion will require approximately \$260.0 to \$280.0 million of capital expenditures in 2005. Through September 30, 2005, the Company had outstanding capital commitments related to the Nordural expansion of \$149.4 million. The Company's cost commitments for the Nordural expansion may materially change depending on the exchange rate between the U.S. dollar and certain foreign currencies, principally the euro and the Icelandic krona. Approximately 64% of the expected project costs for the Nordural expansion are denominated in currencies other than the U.S. dollar, primarily the euro and the krona. As of September 30, 2005, the Company had no hedges to mitigate the Company's foreign currency exposure.

In February 2005, Nordural closed and borrowed under a new \$365.0 million senior term loan facility. Amounts borrowed under the term loan facility were used to refinance debt under Nordural's existing term loan facility, and are being used to finance a portion of the costs associated with the ongoing expansion of the Nordural facility and for Nordural's general corporate purposes. Amounts borrowed under Nordural's term loan facility generally bear interest at a margin over the applicable Eurodollar rate.

The Company has agreements with Hitaveita Sujurnesja hf. ("HS") and Orkuveita Reykjavíkur ("OR") to purchase the power required for the expansion of the production capacity of the Nordural facility from 90,000 mtpy to 220,000 mtpy. OR has also agreed to deliver additional power annually, which will allow a further expansion to 260,000 metric tons by late 2008. The power agreement and the construction of additional production capacity are each subject to the satisfaction of certain conditions. The Company is considering various options for financing the additional capacity.

Other Contingencies

The Company's income tax returns are periodically examined by various tax authorities. The Company is currently under audit by the Internal Revenue Service ("IRS") for the tax years through 2002. In connection with such examinations, the IRS has raised issues and proposed tax deficiencies. The Company is reviewing the issues raised by the IRS and has filed an administrative appeal within the IRS, contesting the proposed tax deficiencies. The Company believes that its tax position is well-supported and, based on current information, does not believe that the outcome of the tax audit will have a material impact on the Company's financial condition or results of operations.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Commodity Prices

The Company is exposed to the price of primary aluminum. The Company manages its exposure to fluctuations in the price of primary aluminum by selling aluminum at fixed prices for future delivery and through financial instruments as well as by purchasing alumina under certain of its supply contracts at prices tied to the same indices as the Company's aluminum sales contracts (see Item 1, Notes to the Consolidated Financial Statements, Note 9 – Forward Delivery Contracts and Financial Instruments). The Company's risk management activities do not include trading or speculative transactions.

Apart from the Pechiney Metal Agreement, Glencore Metal Agreement I, Glencore Metal Agreement II and Southwire Metal Agreement, the Company had forward delivery contracts to sell 93,533 metric tons and 113,126 metric tons of primary aluminum at September 30, 2005 and December 31, 2004, respectively. Of these forward delivery contracts, the Company had fixed price commitments to sell 7,522 metric tons and 6,033 metric tons of primary aluminum at September 30, 2005 and December 31, 2004, respectively, of which none were with Glencore.

Primary Aluminum Fixed Price Financial Sales Contracts as of:

		(Metric Tons)						
		September 30, 2005			December 31, 2004			
	Cash Flow	Cash Flow			Cash Flow			
	Hedges	Derivatives	Total	Hedges	Derivatives	Total		
2005	51,750	_	51,750	193,083	_	193,083		
2006	142,750	25,200	167,950	142,750	25,200	167,950		
2007	119,500	50,400	169,900	119,500	50,400	169,900		
2008	9,000	100,200	109,200	9,000	75,000	84,000		
2009	_	105,000	105,000	_	75,000	75,000		
2010-2015		480,000	480,000		75,000	75,000		
Total	323,000	760,800	1,083,800	464,333	300,600	764,933		

The contracts accounted for as derivatives contain clauses that trigger additional shipment volume when the market price for a contract month is above the contract ceiling price. If the market price exceeds the ceiling price for all contract months through 2015, the maximum additional shipment volume would be 760,800 metric tons. These contracts will be settled monthly. The Company had no fixed price financial purchase contracts to purchase aluminum at September 30, 2005 or December 31, 2004.

Additionally, to mitigate the volatility of the natural gas markets, the Company enters into fixed price financial purchase contracts, accounted for as cash flow hedges, which settle in cash in the period corresponding to the intended usage of natural gas.

Natural Gas Fixed Price Financial Purchase Contracts as of:

	(Thousands of DTH)		
	September 30, 2005	December 31, 2004	
2005	970	2,880	
2006	1,680	480	
2007	780	480	
2008	480	480	
Total	3,910	4,320	

On a hypothetical basis, a \$20 per ton increase in the market price of primary aluminum is estimated to have an unfavorable impact of \$4.1 million after tax on accumulated other comprehensive income for the contracts designated as cash flow hedges, and \$9.7 million on net income for the contracts designated as derivatives, for the period ended September 30, 2005 as a result of the forward primary aluminum financial sales contracts outstanding at September 30, 2005.

On a hypothetical basis, a \$0.50 per DTH decrease in the market price of natural gas is estimated to have an unfavorable impact of \$1.3 million after tax on accumulated other comprehensive income for the period ended September 30, 2005 as a result of the forward natural gas financial purchase contracts outstanding at September 30, 2005. Based on the fair value of the Company's fixed price financial sales contracts for primary aluminum and financial purchase contracts for natural gas that qualify as cash flow hedges as of September 30, 2005, accumulated other comprehensive loss of \$14.6 million is expected to be reclassified as a reduction to earnings over the next 12 month period.

The Company's metals and natural gas risk management activities are subject to the control and direction of senior management. The metals related activities are regularly reported to the Board of Directors of Century.

This quantification of the Company's exposure to the commodity price of aluminum is necessarily limited, as it does not take into consideration the Company's inventory or forward delivery contracts, or the offsetting impact on the sales price of primary aluminum products. Because all of the Company's alumina contracts, except the alumina contract with GAL for the Hawesville facility, are indexed to the LME price for aluminum, they act as a natural hedge for approximately 12% of the Company's production. As of September 30, 2005, approximately 55% and 43% (excluding 25,200 metric tons of potential additional volume under the Company's derivative sales contracts) of the Company's production for the years 2005 and 2006, respectively, was either hedged by the alumina contracts, Nordural electrical power and tolling contracts, and/or by fixed price forward delivery and financial sales contracts.

Nordural. Presently, substantially all of Nordural's revenues are derived from a Toll Conversion Agreement with a subsidiary of BHP Billiton Ltd. whereby Nordural converts alumina provided to it by BHP Billiton into primary aluminum for a fee based on the LME price for primary aluminum. Because of this agreement, Nordural's revenues are subject to the risk of decreases in the market price of primary aluminum; however, Nordural is not exposed to increases in the price for alumina, the principal raw material used in the production of primary aluminum. In addition, under its power contract, Nordural purchases power at a rate which is a percentage of the LME price for primary aluminum, providing Nordural with a natural hedge against downswings in the market for primary aluminum.

Nordural is exposed to foreign currency risk due to fluctuations in the value of the U.S. dollar as compared to the euro and the Icelandic krona. Under its Toll Conversion and power contracts, Nordural's revenues and power costs are based on the LME price for primary aluminum, which is denominated in U.S. dollars. There is no currency risk associated with these contracts. Nordural's labor costs are denominated in Icelandic krona and a portion of its anode costs are denominated in euros. As a result, an increase or decrease in the value of those currencies relative to the U.S. dollar would affect Nordural's operating margins.

Nordural does not currently have financial instruments to hedge commodity or currency risk. Nordural may hedge such risks in the future, including the purchase of aluminum put options to hedge Nordural's commodity risk.

Interest Rates

Interest Rate Risk. The Company's primary debt obligations are the \$250.0 million of outstanding senior unsecured notes, \$175.0 million of outstanding convertible notes, the Nordural debt, including \$188.0 million of borrowings under its revolving credit facility, and the \$7.8 million in industrial revenue bonds ("IRBs") that the Company assumed in connection with the Hawesville acquisition. Because the senior unsecured notes and convertible notes bear a fixed rate of interest, changes in interest rates do not subject the Company to changes in future interest expense with respect to these borrowings. Borrowings under the Company's revolving credit facility, if any, are at variable rates at a margin over LIBOR or the bank base rate, as defined in the revolving credit facility. The IRBs bear interest at variable rates determined by reference to the interest rate of similar instruments in the industrial revenue bond market. At September 30, 2005, Nordural had approximately \$197.2 million of long-term debt consisting primarily of obligations under the Nordural loan facility. Borrowings under Nordural's loan facility bear interest at a margin over the applicable Eurodollar rate. At September 30, 2005, Nordural had \$189.9 million of liabilities which bear interest at a variable rate.

At September 30, 2005, the Company had \$197.7 million of variable rate borrowings. A hypothetical one percentage point increase in the interest rate would increase the Company's annual interest expense by \$2.0 million, assuming no debt reduction. The Company does not currently hedge its interest rate risk, but may do so in the future through interest rate swaps which would have the effect of fixing a portion of its floating rate debt.

The Company's primary financial instruments are cash and short-term investments, including cash in bank accounts and other highly rated liquid money market investments and government securities.

Item 4. Controls and Procedures

a. Evaluation of Disclosure Controls and Procedures

As of September 30, 2005, the Company carried out an evaluation, under the supervision and with the participation of the Company's management, including the Company's Chief Executive Officer and the Chief Financial Officer, of the effectiveness of the Company's disclosure controls and procedures. Based upon that evaluation, the Company's management, including the Chief Executive Officer and the Chief Financial Officer, concluded that the Company's disclosure controls and procedures were effective.

b. Changes in Internal Control over Financial Reporting

During the quarter ended September 30, 2005, the Company had changes in the following processes of internal control over financial reporting:

Nordural ehf converted information systems to SAP from Concord.

Apart from these items, there have not been any changes in the Company's internal controls over financial reporting that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

Part II. OTHER INFORMATION

Item 4. Submission of Matters to a Vote of Stockholders

The Annual Meeting of Company's stockholders was held August 10, 2005. The following are the results of stockholder voting on proposals that were presented and adopted:

1. The election of the following directors for a term of three (3) years expiring at the Annual Meeting of Stockholders to be held in 2008:

	For	Withheld
Craig A. Davis	24,073,690	7,163,061
Robert E. Fishman	29,168,197	2,068,554
Jack E. Thompson	29,166,697	2,070,054

2. To amend the Company's Restated Certificate of Incorporation, as amended to increase the number of authorized shares of the Company's common stock, par value \$.01 per share from 50,000,000 to 100,000,000.

For	27,972,119
Against	3,255,751
Withheld	8,881
Broker non-votes	_

3. To amend and restate the Company's 1996 Stock Incentive Plan.

For	23,407,053
Against	3,217,896
Withheld	500,888
Broker non-votes	4,110,914

4. To amend and restate the Company's Non-employee Directors' Stock Option Plan.

For	23,365,638
Against	3,260,730
Withheld	499,469
Broker non-votes	4,110,914

5. To ratify the appointment of Deloitte & Touche LLP as the Company's independent registered public accounting firm for the fiscal year ending December 31, 2005.

For	30,476,896
Against	755,317
Withheld	4,538
Broker non-votes	_

Item 6. Exhibit Index

		Inco	rporated by R	eference	
Exhibit Number	Description of Exhibit	Form	File No.	Filing Date	Filed <u>Herewith</u>
10.1	Loan and Security Agreement, dated as of September 19, 2005, among Century Aluminum Company, Berkeley Aluminum, Inc., Century Aluminum of West Virginia, Inc., Century Kentucky, Inc., and NSA, Ltd., as borrowers, the lenders and Bank of America, N.A. as agent for the lenders and Bank of America Securities LLC, as lead arranger.				X
31.1	Rule 13a-14(a)/15d-14(a) Certification of the Chief Executive Officer.				X
31.2	Rule 13a-14(a)/15d-14(a) Certification of the Chief Financial Officer.				X
32.1	Section 1350 Certifications.				X
	24				

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Exhibit Index

		Inco	rporated by F	Reference	
Exhibit Number	Description of Exhibit	Form	File No.	Filing Date	Filed Herewith
10.1	Loan and Security Agreement, dated as of September 19, 2005, among Century Aluminum				X
	Company, Berkeley Aluminum, Inc., Century Aluminum of West Virginia, Inc., Century				
	Kentucky, Inc., and NSA, Ltd., as borrowers, the lenders and Bank of America, N.A. as				
	agent for the lenders and Bank of America Securities LLC, as lead arranger*.				
31.1	Rule 13a-14(a)/15d-14(a) Certification of the Chief Executive Officer.				X
31.2	Rule 13a-14(a)/15d-14(a) Certification of the Chief Financial Officer.				X
32.1	Section 1350 Certifications.				X

^{*} Schedules and exhibits are omitted and will be furnished to the Securities and Exchange Commission upon request.

CENTURY ALUMINUM COMPANY

BERKELEY ALUMINUM, INC.

CENTURY ALUMINUM OF WEST VIRGINIA, INC.

CENTURY KENTUCKY, INC.

NSA, LTD.

LOAN AND SECURITY AGREEMENT

dated as of September 19, 2005

\$100,000,000

BANK OF AMERICA, N.A.,

as Agent and as Lead Arranger

CREDIT SUISSE, CAYMAN ISLANDS BRANCH, as Syndication Agent

JPMORGAN CHASE BANK, N.A., as Co-Documentation Agent

CITIBANK, N.A., as Co-Documentation Agent

TABLE OF CONTENTS

SECTI	ON 1. CREDIT FACILITY	1
1.1	Loans	1
1.2	Letters of Credit	3
1.3	Facility Increase	8
SECTI	ON 2. INTEREST, FEES AND CHARGES	9
	Interest	9
	Computation of Interest and Fees	10
	Fee Letter	10
	Letter of Credit Fees	10
2.5	Unused Line Fee	11
2.6	Audit Fees	11
2.7	Reimbursement of Expenses	12
	Payment of Charges	12
2.9	No Deductions	12
SECTI	ON 3. LOAN ADMINISTRATION	13
	Manner of Borrowing Revolving Credit Loans/LIBOR Option	13
	Payments	17
	Mandatory and Optional Prepayments	19
	Application of Payments and Collections; Business Day Convention	20
3.5	All Loans to Constitute One Obligation	21
3.6	Loan Account	21
3.7	Statements of Account	21
3.8	Increased Costs	21
3.9	Suspension of LIBOR Portions	22
3.10	Sharing of Payments, Etc.	23
3.11	Indemnity for Returned Payments	24
3.12	Nature and Extent of Each Borrower's Liability	24
3.13	Lender's Obligation to Mitigate; Replacement of Lenders	25
SECTI	ON 4. TERM AND TERMINATION	26
4.1	Term of Agreement	26
4.2	Termination	26
SECTI	ON 5. SECURITY INTERESTS	27
	Security Interest in Collateral	27
	Excluded Collateral	28
	Lien Perfection; Further Assurances	28
SECTI	ON 6. COLLATERAL ADMINISTRATION	28
	General	28
	Administration of Accounts	29
	Administration of Inventory	30
	•	

i

6.4	Payment of Charges	30
7.1 7.2	ON 7. REPRESENTATIONS AND WARRANTIES General Representations and Warranties Reaffirmation of Representations and Warranties Survival of Representations and Warranties	31 31 35 35
8.1	ON 8. COVENANTS AND CONTINUING AGREEMENTS Affirmative Covenants Negative Covenants	35 35 38
9.1	ON 9. CONDITIONS PRECEDENT Conditions Precedent to Closing and Initial Loans and Letters of Credit Conditions Precedent to Each Loan and Letter of Credit	43 43 45
10.1 10.2 10.3 10.4	ON 10. EVENTS OF DEFAULT; RIGHTS AND REMEDIES ON DEFAULT Events of Default Acceleration of the Obligations Other Remedies Setoff and Sharing of Payments Remedies Cumulative; No Waiver	46 46 48 48 50 50
11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9	Authorization and Action Agent's Reliance, Etc. Bank and its Affiliates Lender Credit Decision Indemnification Rights and Remedies to Be Exercised by Agent Only Agency Provisions Relating to Collateral Agent's Right to Purchase Commitments Resignation of Agent; Appointment of Successor Audit and Examination Reports; Disclaimer by Lenders USA Patriot Act	51 51 52 52 53 53 53 54 54 55 56
12.1 12.2 12.3 12.4 12.5 12.6 12.7 12.8 12.9 12.10	N 12. MISCELLANEOUS Right of Sale, Assignment, Participations Amendments, etc. Power of Attorney Indemnity Sale of Interest Severability Successors and Assigns Cumulative Effect; Conflict of Terms Execution in Counterparts Notice Consent	56 58 59 60 61 61 61 61 61 62
14.1	Consent	UΖ

6.
63
63
63
63
63
64
65
6.5
6.5

LOAN AND SECURITY AGREEMENT

THIS LOAN AND SECURITY AGREEMENT is made as of this 19th day of September, 2005, by and among BANK OF AMERICA, N.A. (together with its successors and assigns, "Bank"), a national banking association, individually as a Lender, as issuer of letters of credit hereunder (in such capacity, together with any Affiliate of Bank appointed by Bank as its successor or assignee hereunder, "Letter of Credit Issuer"), and as Agent (in such capacity, together with its successors and assigns, "Agent") for itself and any other financial institution which is or becomes a party hereto (each such financial institution, including Bank, is referred to hereinafter individually as a "Lender" and collectively as the "Lenders"), the LENDERS, and CENTURY ALUMINUM COMPANY, a Delaware corporation ("Century"), BERKELEY ALUMINUM, INC., a Delaware corporation ("Berkeley Aluminum"), CENTURY ALUMINUM OF WEST VIRGINIA, INC., a Delaware corporation ("Century West Virginia"), CENTURY KENTUCKY, INC., a Delaware corporation ("Century Kentucky"), and NSA, LTD., a Kentucky limited partnership ("NSA", and together with Century, Berkeley Aluminum, Century West Virginia, and Century Kentucky, and each other Person that becomes a party hereto from time to time as a borrower, "Borrowers"). Capitalized terms used in this Agreement and not otherwise specifically defined herein have the meanings assigned to them in Appendix A.

SECTION 1. CREDIT FACILITY

Subject to the terms and conditions of, and in reliance upon the representations and warranties made in, this Agreement and the other Loan Documents, Lenders agree, severally and not jointly, to make a credit facility (the "Facility") of up to the Revolving Credit Maximum Amount available upon Borrowers' request therefor, as follows:

1.1 Loans.

1.1.1 Revolving Credit Loans . Subject to the terms and conditions of, and in reliance upon the representations and warranties made in, this Agreement and the other Loan Documents, each Lender agrees, severally and not jointly, for so long as no Default or Event of Default exists, to make Revolving Credit Loans to Borrowers from time to time during the period from the Closing Date to but not including the last day of the Term, as requested by Borrower Representative in the manner set forth in subsection 3.1.1 hereof, up to a maximum principal amount at any time outstanding equal to the lesser of (i) such Lender's Revolving Loan Commitment minus the product of such Lender's Revolving Loan Percentage and the LC Exposure, and (ii) the product of such Lender's Revolving Loan Percentage and an amount equal to (A) the Borrowing Base at such time minus (B) the LC Exposure at such time minus (C) the Minimum Availability Reserve minus (D) Reserves, if any. Agent shall have the right to establish reserves in such amounts, and with respect to such matters, as Agent may deem necessary from time to time in its Reasonable Credit Judgment in consultation with Borrower Representative based on facts or circumstances not existing on the Closing Date or existing on the Closing Date but not known to Agent on the Closing Date (such reserves, "Reserves"), including (it being understood that the following list may justify reserves but does not

necessarily require them), with respect to (i) price adjustments, damages, unearned discounts, returned products or other matters for which credit memoranda are issued in the ordinary course of any Borrower's business; (ii) sums properly chargeable against Borrowers' Loan Account as Revolving Credit Loans under any section of this Agreement; (iii) amounts owing by any Borrower to any Person (other than a Lender pursuant to the Loan Documents) to the extent secured by a Lien on any of the Collateral; (iv) amounts owing by any Borrower in connection with Product Obligations; (v) dividends declared by a Borrower or Guarantor but not yet paid (but only to the extent the amount of such dividends exceeds the amount of Borrowers' immediately available funds held in Dominion Accounts); and (vi) Dilution. Notwithstanding anything herein to the contrary, reserves will not duplicate (i) eligibility criteria contained in the definitions of "Eligible Accounts" and "Eligible Inventory", and vice versa and (ii) reserves or criteria deducted in computing book value or the net amount of Eligible Accounts or Eligible Inventory. Any changes after the Closing Date by way of establishing new reserve categories or formula changes will require five Business Days' prior notice to Borrower Representative. The Revolving Credit Loans shall be repayable as set forth in Section 3.

1.1.2 Overadvances. Insofar as Borrower Representative may request (such request to be made in the manner set forth in subsection 3.1.1 hereof), and Agent (or other Lenders if required below) may be willing in their sole and absolute discretion to approve, Revolving Credit Loans to Borrowers at a time when the unpaid balance of Revolving Credit Loans plus the LC Exposure plus the Minimum Availability Reserve plus Reserves exceeds, or would exceed with the making of any such Revolving Credit Loan, the Borrowing Base (such Loan or Loans being herein referred to individually as an "Overadvance" and collectively, as "Overadvances"), Agent shall enter such Overadvances as debits in the Loan Account. Any Overadvance made pursuant to the terms hereof shall be made by all Lenders ratably in accordance with their respective Revolving Loan Percentages. Overadvances in the aggregate amount of \$1,000,000 or less may, unless a Default or Event of Default has occurred and is continuing, be made in the sole and absolute discretion of Agent. Overadvances in an aggregate amount of more than \$1,000,000 but less than \$3,000,000 may, unless a Default or an Event of Default has occurred and is continuing, be made in the sole and absolute discretion of the Majority Lenders. Overadvances in an aggregate amount of \$3,000,000 or more and Overadvances to be made after the occurrence and during the continuance of a Default or an Event of Default shall require the consent of all Lenders. All Overadvances shall be repaid on demand and shall bear interest as provided in this Agreement for Revolving Credit Loans generally. The foregoing notwithstanding, in no event, unless otherwise consented to by all Lenders, (w) shall any Overadvances be outstanding for more than 60 consecutive days, (x) after all outstanding Overadvances have been repaid, shall Agent or Lenders make any additional Overadvances unless 60 days or more have expired since the last date on which any Overadvances were outstanding, (y) shall Overadvances be outstanding on more than 90 days within any 180 day period, or (z) shall Agent make Revolving Credit Loans on behalf of Lenders under this subsection 1.1.2 to the extent such Revolving Credit Loans would cause a Lender's share of the Revolving Credit Loans to exceed such Lender's Revolving Loan Commitment minus such Lender's Revolving Loan Percentage of the LC Exposure.

- 1.1.3 Swingline Loans . In order to reduce the frequency of transfers of funds from Lenders to Agent for making Revolving Credit Loans and for so long as no Default or Event of Default has occurred and is continuing, Agent shall be permitted (but not required) to make Revolving Credit Loans to Borrowers upon request by Borrower Representative (such Revolving Credit Loans to be designated as "Swingline Loans"); provided that the aggregate amount of Swingline Loans outstanding at any time will not (i) exceed \$5,000,000; (ii) when added to the principal amount of Agent's other Revolving Credit Loans then outstanding plus Agent's Revolving Loan Percentage of the LC Exposure, exceed Agent's Revolving Credit Commitment; or (iii) when added to the principal amount of all other Revolving Credit Loans then outstanding plus the LC Exposure, exceed the Borrowing Base. Within the foregoing limits, Borrowers may borrow, repay and reborrow Swingline Loans. All Swingline Loans shall be treated as Revolving Credit Loans for purposes of this Agreement, except that (a) all Swingline Loans shall be Base Rate Portions and (b) notwithstanding anything herein to the contrary (other than as set forth in the next succeeding sentence), all principal and interest paid with respect to Swingline Loans shall be for the sole account of Agent in its capacity as the lender of Swingline Loans. Notwithstanding the foregoing, not more than two Business Days after (1) Lenders receive notice from Agent that a Swingline Loan has been advanced in respect of a drawing under a Letter of Credit or (2) in any other circumstance, demand is made by Agent after the occurrence and during the continuance of an Event of Default, each Lender shall irrevocably and unconditionally purchase and receive from Agent, without recourse or warranty from Agent, an undivided interest and participation in each Swingline Loan to the extent of such Lender's Revolving Loan Percentage thereof, by paying to Agent, in same day funds, an amount equal to such Lender's Revolving Loan Percentage of such Swingline Loan.
- 1.1.4 Agent Loans . After the occurrence and during the continuance of an Event of Default, Agent may, in its sole and absolute discretion, make Revolving Credit Loans on behalf of Lenders, in an aggregate amount not to exceed \$3,000,000, if Agent, in its Reasonable Credit Judgment, deems that such Revolving Credit Loans are necessary or desirable (i) to protect all or any portion of the Collateral or (ii) to enhance the likelihood, or maximize the amount of, repayment of the Loans and the other Obligations or (iii) to pay any other amount chargeable to Borrowers pursuant to this Agreement, including costs, fees and expenses as described in Section 2 (hereinafter, "Agent Loans"); provided, that in no event shall the outstanding principal amount of the Revolving Credit Loans exceed the aggregate Revolving Loan Commitments. Each Lender shall be obligated to advance its Revolving Loan Percentage of each Agent Loan. If Agent Loans are made pursuant to the preceding sentence, then (a) the Borrowing Base shall be deemed increased by the amount of such permitted Agent Loans, but only for so long as Agent allows such Agent Loans to be outstanding, and (b) all Lenders that have committed to make Revolving Credit Loans shall be bound to make, or permit to remain outstanding, such Agent Loans based upon their Revolving Loan Percentages in accordance with the terms of this Agreement.

1.2 Letters of Credit.

- 1.2.1 <u>Agreement to Issue or Cause To Issue</u>. Subject to the terms and conditions of this Agreement, Letter of Credit Issuer agrees to issue for the account of Borrowers one or more commercial/documentary or standby letters of credit ("<u>Letter of Credit</u>").
- 1.2.2 Amounts; Outside Expiration Date. Letter of Credit Issuer shall not have any obligation to issue any Letter of Credit hereunder at any time if: (i) the greater of the maximum face amount or the maximum available amount of the requested Letter of Credit is greater than the Unused Letter of Credit Subfacility at such time; (ii) the maximum undrawn amount of the requested Letter of Credit and all commissions, fees, and charges due from Borrowers in connection with the opening thereof would cause Availability to be exceeded at such time; or (iii) such Letter of Credit has an expiration date less than 14 days prior to the Stated Termination Date or more than 12 months from the date of issuance for standby letters of credit and 180 days for commercial/documentary letters of credit. Unless otherwise consented to by Agent, all Letters of Credits must call for sight drafts to be drawn and must be issued in US Dollars.
- 1.2.3 Other Conditions. In addition to conditions precedent contained in Section 9, the obligation of Letter of Credit Issuer to issue any Letter of Credit is subject to the following conditions precedent having been satisfied in a manner reasonably satisfactory to Letter of Credit Issuer:
 - (a) Borrower Representative shall have delivered to Letter of Credit Issuer at least three Business Days prior to the proposed date of issuance, an application in customary form and substance and reasonably satisfactory to Letter of Credit Issuer for the issuance of the Letter of Credit, and such other documents as may be required pursuant to the terms thereof; and the form and terms of the proposed Letter of Credit shall be reasonably satisfactory to Letter of Credit Issuer, it being understood that if any provision of any letter of credit application is inconsistent with any provision of this Agreement, then the provisions of this Agreement shall govern to the extent of any such inconsistency; and
 - (b) as of the date of issuance, no order of any court, arbitrator or Governmental Authority shall purport by its terms to enjoin or restrain Letter of Credit Issuer from issuing letters of credit of the type and in the amount of the proposed Letter of Credit; and no law, rule or regulation applicable to Letter of Credit Issuer and no request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over Letter of Credit Issuer shall prohibit, or request that Letter of Credit Issuer refrain from the issuance of letters of credit generally or the issuance of such Letters of Credit.
- 1.2.4 <u>Disbursement Procedures</u>. Letter of Credit Issuer shall, promptly after its receipt thereof, examine all documents purporting to represent a demand for payment under a Letter of Credit. Letter of Credit Issuer shall promptly notify Agent and Borrower Representative by telephone, facsimile, or e-mail of such demand for payment and whether Letter of Credit Issuer has made or will make a disbursement pursuant thereto; provided that any failure to give or delay in giving such

notice will not relieve Borrowers of their obligation to reimburse Letter of Credit Issuer and Lenders with respect to any such disbursement.

- 1.2.5 Payments Pursuant to Letters of Credit. Any draw under a Letter of Credit shall immediately constitute an Obligation hereunder payable on demand, and shall be deemed to constitute a request by Borrower Representative to Agent for a Borrowing of a Revolving Credit Loan that consists entirely of a Base Rate Portion in the amount of such drawing. The funding date of each such Borrowing shall be the date of the applicable drawing. Borrowers shall pay Letter of Credit Issuer the amount of all other charges and fees payable to Letter of Credit Issuer in connection with any Letter of Credit immediately when due, irrespective of any claim, setoff, defense or other right which Borrowers may have at any time against Letter of Credit Issuer or any other Person.
 - 1.2.6 Indemnification; Assumption of Risk by Borrowers; Certain Authorizations .
 - (a) <u>Indemnification</u>. In addition to amounts payable as elsewhere provided in this <u>Section 1.2</u>, Borrowers agree to protect, indemnify, pay and save Lenders, Agent and Letter of Credit Issuer harmless from and against any and all claims, demands, liabilities, damages, losses, costs, charges and expenses (including reasonable attorneys' fees) which such Person may incur or be subject to as a consequence, direct or indirect, of the issuance of any Letter of Credit. Borrowers' obligations under this <u>subsection 1.2.6</u> shall survive payment of all other Obligations.
 - (b) <u>Assumption of Risk by Borrowers</u>. As among Borrowers, Lenders, Agent and Letter of Credit Issuer, Borrowers assume all risks of the acts and omissions of, or misuse of any of the Letters of Credit by, the respective beneficiaries of such Letters of Credit. In furtherance and not in limitation of the foregoing, Lenders, Agent and Letter of Credit Issuer shall not be responsible for: (1) the form, validity, sufficiency, accuracy, genuineness or legal effect of any document submitted by any Person in connection with the application for and issuance of and presentation of drafts with respect to any of the Letters of Credit, even if it should prove to be in any or all respects invalid, insufficient, inaccurate, fraudulent or forged; (2) the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign any Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason; (3) the failure of the beneficiary of any Letter of Credit to comply duly with conditions required in order to draw upon such Letter of Credit; (4) errors, omissions, interruptions, or delays in transmission or delivery of any messages, by mail, cable, telegraph, telex or otherwise, whether or not they be in cipher; (5) errors in interpretation of technical terms; (6) any loss or delay in the transmission or otherwise of any document required in order to make a drawing under any Letter of Credit or of the proceeds thereof; (7) the misapplication by the beneficiary of any Letter of Credit of the proceeds of any drawing under such Letter of Credit; (8) any consequences

arising from causes beyond the reasonable control of such Person, including any act or omission, whether rightful or wrongful, of any present or future de jure or de facto Governmental Authority; or (9) Letter of Credit Issuer's honor of a draw for which the draw or any certificate fails to comply in any respect with the terms of the Letter of Credit. None of the foregoing shall affect, impair or prevent the vesting of any rights or powers of Agent or any Lender under this subsection 1.2.6. Without limiting the generality of the foregoing, it is expressly understood and agreed that the absolute and unconditional obligation of Borrowers hereunder to reimburse drawings under Letters of Credit will not be excused by any action, omission, or failure to act by any Person or any other circumstance and that, except as provided in the following proviso, no action or omission by Agent, any Lender or Letter of Credit Issuer in respect of Letters of Credit shall result in any liability of any such Person to Borrowers; provided, however, that, the foregoing shall not be construed to excuse Letter of Credit Issuer from liability to any Borrower to the extent of any direct damages (as opposed to special, indirect, consequential or punitive damages claims in respect of which are hereby waived by Borrowers to the extent permitted by applicable law) suffered by such Borrower that are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from Letter of Credit Issuer's gross negligence or willful misconduct in determining whether drafts and other documents presented under a Letter of Credit comply with the terms thereof.

(c) <u>Certain Authorizations</u>. Borrowers hereby authorize and direct Letter of Credit Issuer to name any one or more of the Borrowers as the "Account Party" with respect to any Letter of Credit. Borrowers also authorize Letter of Credit Issuer to deliver to Agent all instruments, documents and other writings and Property received by Letter of Credit Issuer pursuant to such Letter of Credit, and to accept and rely upon Agent's instructions and agreements with respect to all matters arising in connection with the Letter of Credit or the application therefor.

1.2.7 Participations in Letters of Credit .

- (a) <u>Purchase of Participations</u>. Immediately upon issuance of any Letter of Credit in accordance with <u>subsection 1.2.1</u>, each Lender shall be deemed to have irrevocably and unconditionally purchased and received without recourse or warranty, an undivided interest and participation equal to such Lender's Revolving Loan Percentage of the greater of the maximum face or the maximum available amount of such Letter of Credit.
- (b) <u>Sharing of Reimbursement Obligation Payments</u>. Whenever Agent receives a payment from Borrowers on account of reimbursement obligations in respect of a Letter of Credit as to which Agent has previously received payment from a Lender for the account of Letter of Credit Issuer, Agent shall promptly pay to such Lender such Lender's Revolving Loan Percentage of such payment from Borrowers.

- (c) <u>Documentation</u>. Upon the request of any Lender, Agent shall furnish to such Lender copies of any Letter of Credit, reimbursement agreements executed in connection therewith, applications for any Letter of Credit, and such other documentation as may reasonably be requested by such Lender.
- (d) <u>Obligations Irrevocable</u>. The obligations of each Lender to fund its ratable portion of Revolving Credit Loans to be made as a result of a drawing under a Letter of Credit shall be irrevocable and shall not be subject to any qualification or exception whatsoever, including any of the following circumstances:
 - (i) any lack of validity or enforceability of this Agreement or any of the other Loan Documents;
- (ii) the existence of any claim, setoff, defense or other right which any Borrower may have at any time against a beneficiary named in a Letter of Credit or any transferee of any Letter of Credit (or any Person for whom any such transferee may be acting), any Lender, Agent, Letter of Credit Issuer, or any other Person, whether in connection with this Agreement, any Letter of Credit, the transactions contemplated herein or any unrelated transactions (including any underlying transactions between any Borrower or any other Person and the beneficiary named in any Letter of Credit);
- (iii) any draft, certificate or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect;
 - (iv) the surrender or impairment of any security for the performance or observance of any of the terms of any of the Loan Documents;
 - (v) the occurrence of any Default or Event of Default; or
 - (vi) the failure of a Borrower to satisfy the applicable conditions precedent to the issuance thereof.
- 1.2.8 Recovery or Avoidance of Payments; Refund of Payments In Error. If any payment by or on behalf of any Borrower received by Agent with respect to any Letter of Credit and distributed by Agent to Lenders on account of their respective participations therein is thereafter set aside, avoided or recovered from Agent in connection with any receivership, liquidation or bankruptcy proceeding, Lenders shall, upon demand by Agent, pay to Agent their respective Revolving Loan Percentages of such amount set aside, avoided or recovered, together with interest at the rate required to be paid by Agent upon the amount required to be repaid by it. Unless Agent receives notice from Borrower Representative prior to the date on which any payment is due to Lenders that Borrowers will not make such payment in full as and when required, Agent may assume that Borrowers have made such payment in full to Agent on such date in immediately available funds and Agent may (but shall not be so required), in reliance upon such assumption, distribute to each Lender on such due date an amount equal to the

amount then due such Lender. If and to the extent Borrowers have not made such payment in full to Agent, each Lender shall repay to Agent on demand such amount distributed to such Lender, together with interest thereon at the Federal Funds Rate for each day from the date such amount is distributed to such Lender until the date repaid.

- 1.2.9 <u>Indemnification by Lenders</u>. To the extent not reimbursed by Borrowers and without limiting the obligations of Borrowers hereunder, Lenders agree to indemnify Letter of Credit Issuer ratably in accordance with their respective Revolving Loan Percentages, for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including attorneys' fees) or disbursements of any kind and nature whatsoever that may be imposed on, incurred by or asserted against Letter of Credit Issuer in any way relating to or arising out of any Letter of Credit or the transactions contemplated thereby or any action taken or omitted by Letter of Credit Issuer under any Letter of Credit or any Loan Document in connection therewith; <u>provided</u> that no Lender shall be liable for any of the foregoing to the extent it arises from the gross negligence or willful misconduct of the Person to be indemnified. Without limitation of the foregoing, each Lender agrees to reimburse Letter of Credit Issuer promptly upon demand for its Revolving Loan Percentage of any costs or expenses payable by Borrowers to Letter of Credit Issuer, to the extent that Letter of Credit Issuer is not promptly reimbursed for such costs and expenses by Borrowers. The agreement contained in this Section shall survive payment in full of all other Obligations.
- 1.2.10 Supporting Letter of Credit; Cash Collateral . If, notwithstanding the provisions of subsection 1.2.2 and Section 4, any Letter of Credit is outstanding upon the termination of this Agreement, then upon such termination, Borrowers shall deposit with Agent, for the ratable benefit of Agent and Lenders, with respect to each Letter of Credit then outstanding, either (i) a standby letter of credit in form and substance reasonably satisfactory to Agent, issued by an issuer reasonably satisfactory to Agent (a "Supporting Letter of Credit") or (ii) cash collateral, in either case in an amount equal to 102% of the greatest amount for which such Letter of Credit may be drawn plus any fees and expenses associated with such Letter of Credit, under which Supporting Letter of Credit or cash collateral Agent shall be entitled to draw amounts necessary to reimburse Agent and Lenders for payments to be made by Agent and Lenders under such Letter of Credit and any fees and expenses associated with such Letter of Credit. Such Supporting Letter of Credit or cash collateral shall be held by Agent, for the ratable benefit of Agent and Lenders, as security for, and to provide for the payment of, the aggregate undrawn amount of such Letters of Credit remaining outstanding.
- 1.3 Facility Increase. Borrower Representative may from time to time request an increase in the Revolving Credit Maximum Amount and the aggregate Revolving Loan Commitments by an aggregate amount of up to \$25,000,000 (each such increase, a "Facility Increase"). Each Facility Increase shall be made on notice given by Borrower Representative to Agent no later than 12:00 noon (New York City time) 30 days prior to the date of the proposed Facility Increase. Each such notice (a "Notice of Facility Increase") shall (i) specify the date of such proposed Facility Increase (the "Facility Increase Effective Date"), (ii) specify the aggregate amount of such proposed

Facility Increase, which shall be in an amount not less than \$10,000,000 (the "Facility Increase Amount"), and (iii) certify that, at such time, no Default or Event of Default shall have occurred and be continuing (provided that by accepting a requested Facility Increase, Borrower shall be deemed to have represented to Lenders that no Default or Event of Default shall have occurred and be continuing at the time the Facility Increase becomes effective). Agent shall give each Lender prompt notice of Agent's receipt of a Notice of Facility Increase. Agent may approach the existing Lenders to provide the Facility Increase, or, at Borrowers' request, Agent shall invite such other financial institutions selected by Borrowers and reasonably acceptable to Agent to provide the Facility Increase and become Lenders (such existing Lenders and other financial institutions, the "Offerees"). Each Offeree shall have until 3:00 p.m. (New York City time) on the fifth Business Day preceding the Facility Increase Effective Date to commit in writing to all or a portion of the Facility Increase. If the Offerees deliver commitments with respect to such Facility Increase in an amount in excess of the Facility Increase Amount, then Agent shall allocate the Facility Increase to the Offerees committing to the Facility Increase on any basis Agent determines appropriate in consultation with Borrower Representative. On the Facility Increase Effective Date, (A) each Offeree committing to a portion of such Facility Increase shall execute an assumption agreement satisfactory to Agent pursuant to which such Offeree agrees to be bound by the terms of this Agreement as a Lender, (B) the Revolving Credit Maximum Amount and the Revolving Loan Commitments will be increased by the Facility Increase Amount in accordance with the allocations determined by Agent, and (C) each Lender, after giving effect to such Facility Increase, shall purchase or sell the Loans held by it from or to the other Lenders, as directed by Agent, such that after giving effect to such purchases and sales each Lender holds its ratable portion of the outstanding Loans. If the commitments of the Offerees in respect of such Facility Increase are less than the Facility Increase Amount, none of the Lenders shall have any obligation to commit to the uncommitted portion of such Facility Increase, and Borrower Representative may elect either to reduce the Facility Increase Amount accordingly or to terminate the request for a Facility Increase. Notwithstanding the foregoing, no Facility Increase shall be effected unless the conditions set forth in <u>Section 9.2</u> are satisfied on the Facility Increase Effective Date.

SECTION 2. INTEREST, FEES AND CHARGES

2.1 Interest.

2.1.1 <u>Rates of Interest</u>. Interest shall accrue on the principal amount of the Base Rate Portions outstanding at the end of each day at a fluctuating rate per annum equal to the Applicable Margin then in effect <u>plus</u> the Base Rate. Said rate of interest shall increase or decrease by an amount equal to any increase or decrease in the Base Rate, effective as of the opening of business on the day that any such change in the Base Rate occurs. If Borrower Representative exercises the LIBOR Option as provided in <u>Section 3.1</u>, interest shall accrue on the principal amount of the LIBOR Portions outstanding at the end of each day at a rate per annum equal to the Applicable Margin then in effect <u>plus</u> the LIBOR applicable to each LIBOR Portion for the corresponding Interest Period.

- 2.1.2 <u>Default Rate of Interest</u>. At the option of the Majority Lenders, after the occurrence and during the continuance of an Event of Default, all of the Obligations shall bear interest at a rate per annum equal to 2.0% <u>plus</u> the interest rate otherwise applicable thereto (the "Default Rate").
- 2.1.3 <u>Maximum Interest</u>. In no event whatsoever shall the aggregate of all amounts deemed interest hereunder or under the Notes and charged or collected pursuant to the terms of this Agreement or pursuant to the Notes exceed the highest rate permissible under any law which a court of competent jurisdiction shall, in a final determination, deem applicable hereto (the "<u>Maximum Rate</u>"). If any provisions of this Agreement or the Notes are in contravention of any such law, such provisions shall be deemed amended to conform thereto. If at any time the amount of interest paid hereunder is limited by the Maximum Rate, and the rate at which interest accrues hereunder is subsequently below the Maximum Rate, the rate at which interest accrues hereunder shall remain at the Maximum Rate, until such time as the aggregate interest paid hereunder equals the amount of interest that would have been paid had the Maximum Rate not applied.
- 2.2 <u>Computation of Interest and Fees</u>. Interest on LIBOR Portions and Letter of Credit fees shall be calculated daily and shall be computed on the actual number of days elapsed over a year of 360 days, and interest on Base Rate Portions and all other Obligations shall be calculated daily and computed on the actual number of days elapsed over a year of 365 days (or 366 days, as the case may be). For the purpose of computing interest hereunder, all items of payment received by Agent shall be deemed applied by Agent on account of the Obligations (subject to final payment of such items) on the Business Day that such funds become immediately available to Agent in an account in Agent's name.
- 2.3 <u>Fee Letter</u>. Borrowers shall pay to Agent certain fees and other amounts in accordance with the terms of the fee letter between Borrowers and Agent dated as of the date hereof, as amended, restated or modified from time to time (the "<u>Fee Letter</u>").
 - 2.4 <u>Letter of Credit Fees</u>. Borrowers shall pay to Agent:
 - (a) for standby Letters of Credit, for the ratable benefit of Lenders a per annum fee equal to (a) in the case of standby letters of credit 100% supported by cash collateral in a form acceptable to Agent, 0.75%, and (b) in the case of all other standby letters of credit, the Applicable Margin then in effect for LIBOR Portions, in each case multiplied by the aggregate undrawn face amount of such Letters of Credit outstanding from time to time during the term of this Agreement, which fee shall be due and payable quarterly in arrears on January 1, April 1, July 1, and October 1 of each year;
 - (b) for documentary Letters of Credit, for the ratable benefit of Lenders, a per annum fee equal to the Applicable Margin then in effect for LIBOR Portions minus 0.50%, multiplied by the aggregate undrawn face amount of such Letters of Credit outstanding from time to time during the term of this

Agreement, which fee shall be due and payable quarterly in arrears on January 1, April 1, July 1, and October 1 of each year;

- (c) with respect to all Letters of Credit, for the account of Letter of Credit Issuer only, a per annum fronting fee equal to 0.125% of the aggregate face amount of such Letters of Credit outstanding from time to time during the term of this Agreement, which fronting fee shall be due and payable quarterly in arrears on January 1, April 1, July 1, and October 1 of each year; and
- (d) with respect to all Letters of Credit, for the account of Letter of Credit Issuer, (A) such normal and customary issuance, processing and administration charges associated therewith and (B) such presentation, amendment, renewal and other costs and charges as are separately agreed by Borrower Representative and Letter of Credit Issuer from time to time. All such fees, costs and charges shall be due and payable within two Business Days following the issuance by Agent of a reasonably detailed invoice and request for payment thereof to Borrower Representative. The issuance charges shall be deemed fully earned by Letter of Credit Issuer upon issuance of the applicable Letter of Credit.
- 2.5 <u>Unused Line Fee</u>. Borrowers shall pay to Agent, for the ratable benefit of Lenders, a fee (the "<u>Unused Line Fee</u>") equal to the percentage per annum set forth below (the "<u>Unused Line Fee Percentage</u>") multiplied by the average daily amount by which (a) the Revolving Credit Maximum Amount exceeds (b) the sum of (i) the outstanding principal balance of the Revolving Credit Loans and (ii) the LC Exposure (the excess of (a) over (b), the "<u>Unused Line</u>"); provided, that for purposes of allocating the Unused Line Fee among Lenders, outstanding Swingline Loans shall not be included as part of the outstanding balance of the Loans for purposes of calculating such fees owed to Lenders other than Agent. The Unused Line Fee shall be payable quarterly in arrears on January 1, April 1, July 1, and October 1 of each year.

Unused Line (as a % of Revolving	Unused Line Fee
Credit Maximum Amount)	Percentage
50% or less	0.20%
Greater than 50% but less than or equal to 75%	0.25%
Greater than 75%	0.375%

2.6 <u>Audit Fees</u>. Borrowers shall pay to Agent audit fees and expenses in connection with audits (including visits to the facilities of the Borrowers) of the books and records and the amount, value, location, and types of Collateral, which audit fees and expenses shall consist of an \$850 per day per field examiner charge for employees of Agent plus all reasonable and documented out-of-pocket expenses incurred by Agent in connection with such audits, whether such audits are conducted by employees of Agent or by third parties hired by Agent. Notwithstanding the foregoing, so long as no Default or Event of Default has occurred and is continuing, Borrowers' obligation to pay for

audits in any 12-month period shall be limited as follows (it being understood that Agent shall not be prohibited from conducting additional audits at its own expense): (a) so long as Availability remains greater than or equal to \$35,000,000 during such 12-month period, one audit, and (b) otherwise, two audits.

- 2.7 Reimbursement of Expenses. Borrowers agree to reimburse (i) Bank for all reasonable and documented out-of-pocket costs and expenses (including legal fees and expenses of Bank's external counsel) of Bank associated with this Agreement or any of the other Loan Documents and the transactions contemplated hereby and thereby, including (A) the negotiation and preparation of this Agreement or any of the other Loan Documents, any amendment of or modification of this Agreement or any of the other Loan Documents, or any syndication or attempted syndication of the Obligations and (B) the administration of this Agreement or any of the other Loan Documents and the transactions contemplated hereby and thereby; and (ii) Agent or any Lender for reasonable and documented legal or accounting expenses or any other reasonable and documented costs or out-of-pocket expenses in connection with (A) any litigation, contest, dispute, suit, proceeding or action (whether instituted by Agent, any Lender, any Borrower or any other Person) relating to the Collateral, this Agreement or any of the other Loan Documents, (B) any attempt to enforce any rights of Agent or any Lender against any Borrower or any other Person which may be obligated to Agent or any Lender by virtue of this Agreement or any of the other Loan Documents, including the Account Debtors, or (C) after the occurrence and during the continuance of an Event of Default, any attempt to inspect, verify, protect, preserve, restore, collect, sell, liquidate or otherwise dispose of or realize upon the Collateral; provided that Borrowers shall not be responsible to Agent or any Lender for such costs and out-of-pocket expenses to the extent determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Person or to the extent such costs and expenses result from a claim brought by Borrowers against Agent or any Lender for breach in bad faith of such Person's obligations hereunder if Borrowers have obtained a final and nonappealable judgment from a court of competent judgment in Borrowers' favor on such claim.
- 2.8 <u>Payment of Charges</u>. All amounts properly chargeable to Borrowers under any Loan Document shall be Obligations, shall be, unless specifically otherwise provided, payable on demand, and shall bear interest from the date demand was made or such amount is due, as applicable, until paid in full at the rate applicable to the Base Rate Portion from time to time. Notwithstanding the foregoing, amounts chargeable to Borrowers under <u>Sections 2.6</u> and <u>2.7</u> shall be due and payable within two Business Days following the date of issuance by Agent of a reasonably detailed invoice and request for payment thereof to Borrower Representative.

2.9 No Deductions.

(a) Any and all payments or reimbursements made hereunder shall be made free and clear of and without deduction for any and all taxes, levies, imposts, deductions, charges or withholdings, and all liabilities with respect thereto; excluding, however, the following: taxes imposed on the income of Agent or any Lender or

franchise taxes imposed by the jurisdiction under the laws of which Agent or any Lender is organized or doing business or any political subdivision thereof and taxes imposed on the income of Agent or any Lender by the jurisdiction of Agent's or such Lender's applicable lending office or any political subdivision thereof or franchise taxes (all such non-excluded taxes, levies, imposts, deductions, charges or withholdings and all liabilities with respect thereto, "Tax Liabilities"). If any Borrower shall be required by law to deduct any such Tax Liabilities from or in respect of any sum payable hereunder to Agent or any Lender, then the sum payable hereunder shall be increased as may be necessary so that, after all required deductions are made, Agent or such Lender receives an amount equal to the sum it would have received had no such deductions been made.

(b) A Lender or transferee of Lender shall not be entitled to any additional payments under <u>subsection 2.9(a)</u> before it has satisfied the requirements of <u>subsection 12.1.3</u>. If any Lender becomes subject to any Tax Liability and is not entitled to any additional payments under <u>subsection 2.9(a)</u> Borrowers shall take such steps (at such Foreign Lender's expense) as such Foreign Lender shall reasonably request to assist such Foreign Lender to recover such Tax Liability.

SECTION 3. LOAN ADMINISTRATION

- 3.1 <u>Manner of Borrowing Revolving Credit Loans/LIBOR Option</u>. Borrowings under the credit facility established pursuant to <u>Section 1</u> hereof shall be as follows:
- 3.1.1 Loan Requests. A request for a Revolving Credit Loan shall be made, or shall be deemed to be made, in the following manner: (a) Borrower Representative shall give Agent notice of its intention to borrow, which notice shall be irrevocable and shall specify (i) the amount of the proposed borrowing of a Revolving Credit Loan (which shall be no less than \$500,000 or an integral multiple of \$100,000 in excess thereof in the case of LIBOR Portions (it being understood that there shall be no minimum borrowing amount with respect to Base Rate Portions)) and (ii) the proposed borrowing date, which shall be a Business Day, no later than 1:00 pm (New York City time) on the proposed borrowing date (or in accordance with subsection 3.1.6, 3.1.7 or 3.1.8, as applicable, in the case of a request for a LIBOR Portion); provided, however, that no such request may be made after the occurrence and during the continuance of a Default or an Event of Default; and (b) the failure of Borrowers to pay on the date such payment is due (without giving effect to any grace periods prior to which such nonpayment would constitute an Event of Default) any amount required to be paid under this Agreement, or the Notes, whether as interest or for any other Obligation, shall be deemed irrevocably to be a request for a Revolving Credit Loan on the due date in the amount required to pay such interest or other Obligation.
- 3.1.2 <u>Disbursement</u>. Borrowers hereby irrevocably authorize Agent to disburse the proceeds of each Revolving Credit Loan requested or deemed to be requested pursuant to <u>subsection 3.1.1</u> as follows: (i) the proceeds of each Revolving Credit Loan requested pursuant to clause (a) of <u>subsection 3.1.1</u> shall be disbursed by Agent in lawful money of the United States of America in immediately available funds,

(A) in the case of any Borrowing on the Closing Date, in accordance with the terms of the written disbursement letter from Borrower Representative, and (B) in the case of each subsequent Borrowing, by wire transfer to such bank account as may be specified by Borrower Representative to Agent from time to time pursuant to a written direction, and (ii) the proceeds of each Revolving Credit Loan deemed requested pursuant to clause (b) of subsection 3.1.1 shall be disbursed by way of direct payment of the relevant Obligation. If at any time any Loan is funded by Agent or Lenders in excess of the amount requested or deemed requested by Borrower Representative, Borrowers agree to repay the excess to Agent (without interest) promptly after the earlier to occur of (a) any Borrower's discovery of the error and (b) notice thereof to Borrower Representative from Agent or any Lender, and the failure of Borrowers to so return any such excess shall be deemed to be an irrevocable request for a Revolving Credit Loan on the date such excess was received by Borrowers in the amount of such excess.

3.1.3 Payment by Lenders.

- (a) Agent shall give to each Lender prompt written notice by facsimile, telex or cable of the receipt by Agent from Borrower Representative of any request for a Revolving Credit Loan. Each such notice shall specify the requested date and amount of such Revolving Credit Loan, whether such Revolving Credit Loan shall be subject to the LIBOR Option, and the amount of each Lender's advance thereunder (in accordance with its applicable Revolving Loan Percentage). Each Lender shall, not later than 1:00 p.m. (Chicago, Illinois time) on such requested date, wire to a bank designated by Agent the amount of that Lender's Revolving Loan Percentage of the requested Revolving Credit Loan. The failure of any Lender to make the Revolving Credit Loans to be made by it shall not release any other Lender of its obligations hereunder to make its Revolving Credit Loan. Neither Agent nor any Lender shall be responsible for the failure of any other Lender to make the Revolving Credit Loan to be made by such other Lender. The foregoing notwithstanding, Agent, in its sole discretion, may from its own funds make a Revolving Credit Loan on behalf of any Lender. In such event, the Lender on behalf of whom Agent made the Revolving Credit Loan shall reimburse Agent for the amount of such Revolving Credit Loan made on its behalf, on a weekly (or more frequent, as determined by Agent in its sole discretion) basis. On each such settlement date, Agent will pay to each Lender the net amount owing to such Lender in connection with such settlement, including amounts relating to Loans, fees, interest and other amounts payable hereunder. The entire amount of interest attributable to such Revolving Credit Loan for the period from the date on which such Revolving Credit Loan was made by Agent on such Lender's behalf until Agent is reimbursed by such Lender, shall be paid to Agent for its own account.
- (b) If Agent makes Revolving Credit Loans available to Borrowers and any Lender fails to either make available to Agent its Revolving Loan Percentage of such Revolving Credit Loan or reimburse Agent as provided in paragraph (a) above (such Lender, a "<u>Defaulting Lender</u>"), Agent will notify Borrower Representative of such failure to fund and, upon demand by Agent,

Borrowers shall pay such amount to Agent for Agent's account, together with interest thereon for each day elapsed since the date of such borrowing, at a rate per annum equal to the interest rate applicable at the time to the Revolving Credit Loans comprising that particular borrowing. Agent shall not be obligated to transfer to a Defaulting Lender any payments made by Borrowers to Agent for the Defaulting Lender's benefit; nor shall a Defaulting Lender be entitled to the sharing of any payments hereunder. Amounts payable to a Defaulting Lender shall instead be paid to or retained by Agent. In its discretion, Agent may, in connection with disbursing the proceeds of Revolving Credit Loans made pursuant to a notice of borrowing, include in such disbursement the amount of all such payments received or retained by it for the account of such Defaulting Lender. Any amounts so loaned to Borrowers shall bear interest at the rate applicable to Base Rate Portions and for all other purposes of this Agreement shall be treated as if they were Revolving Credit Loans; provided, however, that for purposes of voting or consenting to matters with respect to the Loan Documents and determining Revolving Loan Percentages, such Defaulting Lender shall be deemed not to be a "Lender". Until a Defaulting Lender cures its failure to fund its Revolving Loan Percentage of any borrowing (A) such Defaulting Lender shall not be entitled to any portion of the Unused Line Fee, (B) the Unused Line Fee shall accrue in favor of Lenders which have funded their respective Revolving Loan Percentages of such requested borrowing and shall be allocated among such performing Lenders ratably based upon their relative Revolving Loan Commitments calculated without regard to the Revolving Loan Commitments of the Defaulting Lender, and (C) the Unused Line Fee shall be calculated as if the Defaulting Lender's entire Revolving Loan Commitment had been funded. This subsection 3.1.3(b) shall remain effective with respect to a Defaulting Lender until such time as such Lender shall no longer be in default of any of its obligations under this Agreement. The terms of this subsection 3.1.3(b) shall not be construed to increase or otherwise affect the Revolving Loan Commitment of any Lender, or relieve or excuse the performance by any Borrower of its duties and obligations hereunder. Any payments by Borrower pursuant to this subsection 3.1.3(b) on account of a Defaulting Lender shall be without prejudice to any claims Borrowers may have against such Defaulting Lender.

3.1.4 Issuance of Letters of Credit.

(a) <u>Request for Issuance</u>. Borrower Representative shall notify Letter of Credit Issuer of a requested Letter of Credit at least three Business Days prior to the proposed issuance date. Such notice shall be irrevocable and shall specify the original face amount of the Letter of Credit requested, the Business Day of issuance of such requested Letter of Credit, whether such Letter of Credit may be drawn in a single or in partial draws, the Business Day on which the requested Letter of Credit is to expire and the beneficiary of the requested Letter of Credit. Borrower Representative shall attach to such notice the proposed form of the Letter of Credit.

- (b) Responsibilities of Agent; Issuance . As of the Business Day immediately preceding the requested issuance date of the Letter of Credit, Agent shall determine the amount of the applicable Unused Letter of Credit Subfacility and Availability. If (i) the face amount of the requested Letter of Credit is less than the Unused Letter of Credit Subfacility and (ii) the amount of such requested Letter of Credit and the fronting fees due from Borrowers pursuant to subsection 2.4(c) in connection with the opening thereof would not exceed Availability, Agent shall so notify Letter of Credit Issuer and Letter of Credit Issuer shall issue the requested Letter of Credit on the requested issuance date so long as the other conditions hereof are met.
- (c) <u>No Extensions or Amendment</u>. Letter of Credit Issuer shall not be obligated to extend or amend any Letter of Credit issued pursuant hereto unless the applicable conditions of <u>Section 1.2</u> are met as though a new Letter of Credit were being requested and issued.
- 3.1.5 Method of Making Requests. Unless a Default or an Event of Default has occurred and is continuing, (i) telephonic or electronic requests for Revolving Credit Loans to Agent shall be permitted, (ii) Letter of Credit Issuer may, in its discretion, permit electronic transmittal of requests for Letters of Credit to it, and (iii) Agent may, in Agent's discretion, permit electronic transmittal of instructions, authorizations, agreements or reports to Agent. Unless Borrower Representative specifically directs Agent or Letter of Credit Issuer in writing not to accept or act upon telephonic or electronic communications from any Borrower, neither Agent nor Letter of Credit Issuer shall have any liability to Borrowers for any loss or damage suffered by any Borrower as a result of Agent's or Letter of Credit Issuer's honoring of any requests, execution of any instructions, authorizations or agreements or reliance on any reports communicated to it telephonically or electronically and purporting to have been sent to Agent or Letter of Credit Issuer by any Borrower, and, so long as Agent and Letter of Credit Issuer act in good faith, neither Agent nor Letter of Credit Issuer shall have any duty to verify the origin of any such communication. Each telephonic request for a Revolving Credit Loan or Letter of Credit accepted by Agent (or Letter of Credit Issuer, as applicable) hereunder shall be promptly followed by a written or electronic confirmation of such request from Borrower Representative to Agent or Letter of Credit Issuer, as applicable.
- 3.1.6 <u>LIBOR Portions</u>. Provided that as of both the date of the LIBOR Request and the first day of the Interest Period, no Default or Event of Default has occurred and is continuing, if Borrowers desire to obtain a LIBOR Portion, Borrower Representative shall give Agent a LIBOR Request no later than 1:00 p.m. (New York City time) on the third Business Day prior to the requested borrowing date. Each LIBOR Request shall be irrevocable and binding on Borrowers. In no event shall Borrowers be permitted to have outstanding at any one time LIBOR Portions with more than five different Interest Periods.
- 3.1.7 <u>Conversion of Base Rate Portions</u>. Provided that as of both the date of the LIBOR Request and the first day of the Interest Period, no Default or Event of

Default exists, Borrowers may, on any Business Day, convert any Base Rate Portion into a LIBOR Portion. If Borrowers desire to convert a Base Rate Portion, Borrower Representative shall give Agent a LIBOR Request no later than 1:00 p.m. (New York City time) on the third Business Day prior to the requested conversion date. After giving effect to any conversion of Base Rate Portions to LIBOR Portions, Borrowers shall not be permitted to have outstanding at any one time LIBOR Portions with more than five different Interest Periods.

- 3.1.8 Continuation of LIBOR Portions . Provided that, as of both the date of the LIBOR Request and the first day of the Interest Period, no Default or Event of Default has occurred and is continuing, Borrowers may, on any Business Day, continue any LIBOR Portions into a subsequent Interest Period of the same or a different permitted duration. If Borrowers desire to continue a LIBOR Portion, Borrower Representative shall give Agent a LIBOR Request no later than 1:00 p.m. (New York City time) on the third Business Day prior to the requested continuation date. After giving effect to any continuation of LIBOR Portions, Borrowers shall not be permitted to have outstanding at any one time LIBOR Portions with more than five different Interest Periods. If Borrower Representative shall fail to give timely notice of its election to continue any LIBOR Portion or portion thereof as provided above, or if such continuation shall not be permitted, such LIBOR Portion or portion thereof, unless such LIBOR Portion shall be repaid, shall automatically be converted into a Base Rate Portion at the end of the Interest Period then in effect with respect to such LIBOR Portion.
 - 3.2 Payments. The Obligations shall be payable as follows:
- 3.2.1 Principal. Principal on account of Revolving Credit Loans shall be payable by Borrowers to Agent for the ratable benefit of Lenders immediately upon the earliest of (i) the receipt by Agent or any Borrower of any proceeds of any of the Collateral that are required to be paid to Agent for the ratable benefit of Lenders as a mandatory prepayment pursuant to subsection 3.3.2, subject to Borrowers' rights to reborrow such amounts in accordance with subsection 1.1.1 hereof, (ii) the receipt by Agent or any Borrower of any proceeds of any of the Collateral, to the extent of said proceeds, following the delivery by Agent of a Notice of Exclusive Control in respect of a Dominion Account (for so long as Agent's right to exclusive control resulting from the delivery of the Notice of Exclusive Control remains in effect and subject to Borrowers' rights to reborrow such amounts in accordance with subsection 1.1.1 hereof), (iii) the occurrence of an Event of Default in consequence of which Agent or Majority Lenders elect to accelerate the maturity and payment of the Obligations in accordance with this Agreement (or in consequence of which the maturity and payment of the Obligations is automatically accelerated), and (iv) termination of this Agreement pursuant to Section 4 hereof; provided, however, that, if an Overadvance shall exist at any time, Borrowers shall repay the Overadvance on demand or as provided in subsection 3.3.1 hereof. Any proceeds required to be remitted to Agent pursuant to clauses (i) or (ii) of the previous sentence shall, until received by Agent, be held as Agent's property, for its benefit and the benefit of Lenders, by each applicable Borrower as trustee of an express trust for Agent's benefit. Notwithstanding clause (ii) above, during any period when (A) Agent is exercising exclusive control of Borrowers' Dominion Account(s) pursuant to a Notice of

Exclusive Control and (B) no Default or Event of Default has occurred and is continuing and (C) the proceeds of Collateral theretofore received by Agent have been applied to pay the Obligations set forth in items "<u>first</u>" through "<u>fifth</u>" of <u>subsection 3.4.2</u> hereof (subject to Borrowers' rights to maintain cash in a Dominion Account to avoid prepayment of LIBOR Portions prior to the last day of an Interest Period pursuant to <u>subsection 3.3.3</u> hereof), <u>then</u> Borrowers shall be entitled to withdraw funds from the Dominion Account(s) to pay normal operating costs of the Borrowers.

3.2.2 Interest.

- (a) <u>Base Rate Portion</u>. Interest accrued on the Base Rate Portion shall be due and payable upon each of the following dates: (1) the last day of each calendar quarter (for such calendar quarter), computed through the date of payment; (2) the occurrence of an Event of Default in consequence of which Agent or Majority Lenders elect to accelerate the maturity and payment of the Obligations in accordance with this Agreement (or in consequence of which the maturity and payment of the Obligations is automatically accelerated); and (3) the date of termination of this Agreement pursuant to <u>Section 4</u> hereof.
- (b) <u>LIBOR Portion</u>. Interest accrued on each LIBOR Portion shall be due and payable upon each of the following dates: (1) each LIBOR Interest Payment Date; (2) the prepayment thereof (unless Borrowers shall have exercised their option under <u>subsection 3.3.3</u> with respect to any applicable prepayment); (3) the occurrence of an Event of Default in consequence of which Agent or Majority Lenders elect to accelerate the maturity and payment of the Obligations in accordance with this Agreement (or in consequence of which the maturity and payment of the Obligations is automatically accelerated); and (4) the date of termination of this Agreement pursuant to Section 4 hereof.
- 3.2.3 <u>Costs, Fees and Charges</u>. Costs, fees and charges payable pursuant to this Agreement shall be payable by Borrowers to Agent, for distribution to Lenders, as appropriate, or to any other Person designated by Agent in writing, as and when provided in this Agreement or any other Loan Document.
- 3.2.4 Other Obligations. The balance of the Obligations requiring the payment of money, if any, shall be payable by Borrowers to Agent for distribution to Lenders, as appropriate, as and when provided in this Agreement or any other Loan Document or on demand, as applicable.
- 3.2.5 <u>Prepayment of/Failure to Borrow LIBOR Portions</u>. Borrowers may prepay a LIBOR Portion only upon three Business Days' prior written notice to Agent (which notice shall be irrevocable). In the event of (i) the payment of any principal of any LIBOR Portion other than on the last day of the Interest Period applicable thereto (including as a result of an Event of Default), (ii) the conversion of any LIBOR Portion other than on the last day of the Interest Period applicable thereto, or (iii) the failure, on the date specified in any notice delivered pursuant hereto, to borrow, convert, continue or prepay any LIBOR Portion, then, in any such event, Borrowers shall

compensate Lenders for the documented cost and expense attributable to such event, as determined by Agent in a manner consistent with its normal customs and practices.

- 3.3 Mandatory and Optional Prepayments .
- 3.3.1 Revolving Exposure in Excess of Borrowing Base. Unless an Overadvance is requested and granted with respect thereto pursuant to subsection 1.1.2, if at any time the outstanding Revolving Credit Loans plus the LC Exposure plus Reserves exceeds the Borrowing Base, Borrowers shall immediately repay the Revolving Credit Loans to the extent required to eliminate such excess; provided, however, that, if any such excess results from a discretionary reduction in the Borrowing Base by Agent pursuant to this Agreement after the Closing Date including the establishment of any reserve, then the applicable Borrowers shall not be required to repay the excess before the end of the tenth Business Day after receipt of such notice.
- 3.3.2 Proceeds of Sale, Loss, Destruction or Condemnation of Collateral. If any Borrower sells, leases or otherwise disposes of or transfers any of the Collateral (other than Inventory sold in the ordinary course of business) having a value in excess of \$10,000,000 or if any of the Collateral having a value in excess of \$10,000,000 is lost or destroyed or taken by condemnation, Borrowers shall promptly either (a) deliver to Agent a Borrowing Base Certificate showing that after giving effect to such disposition, loss, destruction, or condemnation, Availability is greater than zero, or (b) pay to Agent for the ratable benefit of Lenders as and when received by any Borrower and as a mandatory prepayment of the Loans, such amount as is needed to cause Availability to be greater than zero from the proceeds (including insurance payments but net of costs and taxes incurred in connection with such sale or event) received by such Borrower from such disposition, loss, destruction, or condemnation. Any such prepayment shall be applied to reduce the outstanding principal balance of the Revolving Credit Loans, but shall not permanently reduce the Revolving Loan Commitments.
- 3.3.3 <u>LIBOR Portions</u>. If the application of any payment made in accordance with the provisions of <u>Sections 3.2</u>, <u>3.3</u> and <u>3.4</u> at a time when no Event of Default has occurred and is continuing would result in termination of a LIBOR Portion prior to the last day of the Interest Period for such LIBOR Portion, then, at Borrowers' option, the amount of such prepayment shall not be applied to reduce the outstanding principal balance of the Revolving Credit Loans, but shall instead be deposited in a Dominion Account, and the LIBOR Portion and all of Borrowers' obligations in connection therewith, including with respect to payment of principal thereof and interest thereon, shall continue as if no prepayment of such LIBOR Portion had been required or made.
- 3.3.4 Optional Payments. Borrowers may, at their option from time to time upon not less than three Business Days' prior written notice from Borrower Representative to Agent in the case of LIBOR Portions and not later than 10:00 a.m., New York City time on the date of payment to Agent in the case of Base Rate Portions, repay Base Rate Portions or repay/prepay LIBOR Portions; provided that the amount of any such payment in the case of LIBOR Portions is at least \$500,000 (or such lesser

amount as constitutes all outstanding LIBOR Portions at such time) and in integral multiples of \$100,000 above \$500,000. Except for charges under <u>subsection 3.2.5</u> applicable to payments of LIBOR Portions, all such payments shall be without premium or penalty.

- 3.3.5 Optional Reductions of Revolving Loan Commitments. Borrowers may, at their option from time to time upon not less than three Business Days' prior written notice from Borrower Representative to Agent, terminate in whole or permanently reduce ratably in part, the unused portion of the Revolving Loan Commitments; provided, however, that each such partial reduction shall be in an amount of \$5,000,000 or integral multiples of \$1,000,000 in excess thereof.
 - 3.4 Application of Payments and Collections; Business Day Convention .
- 3.4.1 <u>Collections</u>. All items of payment received by Agent by 2:00 p.m., New York City time, on any Business Day shall be deemed received on that Business Day. All items of payment received after 2:00 p.m., New York City time, on any Business Day shall be deemed received on the following Business Day. If as the result of collections of Accounts as authorized by <u>subsection 6.2.4</u> hereof or otherwise, a credit balance exists in the Loan Account, such credit balance shall not accrue interest in favor of Borrowers, but shall be disbursed to Borrowers or otherwise at Borrower Representative's direction in the manner set forth in <u>subsection 3.1.2</u>, upon Borrower Representative's request at any time, so long as no Default or Event of Default has occurred and is continuing. Agent may, at its option, offset such credit balance against any of the Obligations upon and during the continuance of an Event of Default.
- 3.4.2 Apportionment, Application and Reversal of Payments. Principal and interest payments shall be apportioned ratably among Lenders (according to the unpaid principal balance of the Loans to which such payments relate held by each Lender), and fees, except as otherwise provided herein or in the Fee Letter, shall be apportioned ratably among Lenders. All payments shall be remitted to Agent and shall be applied ratably among Lenders, in accordance with the provisions of this Agreement as follows: first., to pay any fees or expense reimbursements (other than amounts related to Product Obligations) then due to Lenders from Borrowers; second, to pay interest due from Borrowers in respect of all Loans, including Swingline Loans, Overadvances, and Agent Loans; third, to pay or prepay principal of Swingline Loans and Agent Loans; fourth, to pay or prepay principal of Swingline Loans and Agent Loans; fourth, to pay or prepay principal of Swingline Loans and Agent Loans, but including Overadvances) and unpaid reimbursement obligations in respect of Letters of Credit; fifth, to pay principal of LIBOR Portions in the chronological order of expiration of the Interest Periods thereof; sixth, to pay as cash collateral or a Supporting Letter of Credit in an amount equal to 102% of the outstanding Letter of Credit Obligations (to the extent not supported by a Supporting Letter of Credit in such amount); seventh, to the payment of any other Obligation due to Agent or any Lender by Borrowers (other than Product Obligations); eighth, to pay amounts due to Bank or an Affiliate of Bank in respect of Product Obligations of the type described in clause (i) of the definition thereof; and ninth, to pay amounts due to Bank, any Affiliate of Bank, or any other Lender

the occurrence and during the continuance of an Event of Default, as between Agent and Borrowers, Agent shall have the continuing exclusive right to apply and reapply any and all such payments and collections received at any time or times hereafter by Agent against the Obligations, in such manner as Agent may deem advisable to comply with this <u>subsection 3.4.2</u>, notwithstanding any entry by Agent or any Lender upon any of its books and records.

- 3.4.3 <u>Business Day Convention</u>. Whenever any payment, report, document, or notice hereunder shall be stated to be due on a day other than a Business Day, the due date therefor shall be extended to the next Business Day, and in the case of a payment which accrues interest, interest thereon will be payable for the period of such extension; <u>provided</u>, <u>however</u>, that if such extension would cause payment of interest on or principal of any LIBOR Portion to be made in the next calendar month, such payment shall be made on the immediately preceding Business Day.
- 3.5 <u>All Loans to Constitute One Obligation</u>. The Loans and LC Obligations shall constitute one general Obligation of Borrowers, and shall be secured by Agent's Lien upon all of the Collateral.
- 3.6 <u>Loan Account</u>. Agent shall enter all Loans as debits to a loan account (the "<u>Loan Account</u>") and shall also record in the Loan Account all payments made by Borrowers on any Obligations and all other amounts credited to the Loan Account as provided herein, and may record therein, in accordance with customary accounting practice, other debits and credits, including interest and all other charges and expenses properly chargeable to Borrowers under the Loan Documents if Borrowers have not paid the same when due.
- 3.7 <u>Statements of Account</u>. Agent will account to Borrowers monthly with a statement of Loans, charges and payments made pursuant to this Agreement during the immediately preceding month, and such accounts rendered by Agent shall be *prima facie* evidence of the existence and amounts of the obligations recorded therein; provided that any failure by Agent to maintain and deliver such accounts or any error therein shall not affect Borrowers' obligation to repay the Obligations in accordance with the terms of this Agreement.
- 3.8 <u>Increased Costs</u>. If any law or any governmental or quasi-governmental rule, regulation, policy, guideline or directive (whether or not having the force of law) adopted or implemented after the date of this Agreement and having general applicability to all banks or finance companies within the jurisdiction in which any Lender operates (excluding, for the avoidance of doubt, the effect of and phasing in of capital requirements or other regulations or guidelines passed prior to the date of this Agreement), or any interpretation or application thereof after the date hereof by any governmental authority charged with the interpretation or application thereof, or the compliance of such Lender therewith, shall:
 - (a) (1) subject such Lender to any tax with respect to this Agreement (other than (a) any tax based on or measured by net income or otherwise in the

nature of a net income tax, including any franchise tax or any similar tax based on capital, net worth or comparable basis for measurement, (b) any tax collected by a withholding on payments and which is neither computed by reference to the net income of the payee nor in the nature of an advance collection of a tax based on or measured by the net income of the payee and (c) any tax covered by subsection 2.9(a) or (2) change the basis of taxation of payments to such Lender of principal, fees, interest or any other amount payable hereunder or under any Loan Documents (other than in respect of (a) any tax based on or measured by net income or otherwise in the nature of a net income tax, including any franchise tax or any similar tax based on capital, net worth or comparable basis for measurement, (b) any tax collected by a withholding on payments and which is neither computed by reference to the net income of the payee nor in the nature of an advance collection of a tax based on or measured by the net income of the payee and (c) any tax covered by subsection 2.9(a));

- (b) impose, modify or hold applicable any reserve (except any reserve taken into account in the determination of the applicable LIBOR), special deposit, assessment or similar requirement against assets held by, or deposits in or for the account of, or advances or loans by, or other credit extended by, any office of such Lender which is not otherwise included in the determination of LIBOR hereunder, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
 - (c) impose on such Lender any other condition affecting any Loan Document;

and the result of any of the foregoing is to increase the cost to such Lender of making, renewing or maintaining Loans hereunder or the result of any of the foregoing is to reduce the rate of return on such Lender's capital as a consequence of its obligations hereunder, or the result of any of the foregoing is to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of any of the Loans, then, in any such case, Borrowers shall pay such Lender, upon demand and certification not later than 60 days following receipt of notice by Borrower Representative of the imposition of such increased costs, such additional amount as will compensate such Lender for such additional cost or such reduction, as the case may be, to the extent such Lender has not otherwise been compensated, with respect to a particular Loan, for such increased cost as a result of an increase in the Base Rate or the LIBOR. An officer of the applicable Lender shall determine the amount of such additional cost or reduced amount using reasonable averaging and attribution methods and shall certify the amount of such additional cost or reduced amount to Borrowers, which certification shall include a written explanation and details of such additional cost or reduction to Borrowers and shall contain a representation and warranty on the part of the Lender to the effect that the Lender has complied with its obligations set forth in subsection 3.13.1 to eliminate or reduce such amount. Such certification shall be conclusive absent demonstrable error.

3.9 Suspension of LIBOR Portions . If Agent or the Majority Lenders shall have determined that:

- (i) reasonable means do not exist for ascertaining the LIBOR for any Interest Period; or
- (ii) Dollar deposits in the relevant amount and for the relevant maturity are not available in the London interbank market with respect to a proposed LIBOR Portion, or a proposed conversion of a Base Rate Portion into a LIBOR Portion; or
- (iii) any applicable law, treaty, regulation or directive, or any change therein or in the interpretation or application thereof after the date hereof, shall make it unlawful for any Lender (for purposes of this clause (iii), the term "Lender" shall include the office or branch where such Lender or any corporation or bank then controlling such Lender makes or maintains any LIBOR Portions) to make or maintain its LIBOR Portions, or adverse or unusual conditions in, or changes in applicable law relating to, the London interbank market make it, in the reasonable judgment of Agent, impracticable to fund therein any of the LIBOR Portions, or make the projected LIBOR unreflective of the actual costs of funds therefor to any Lender;
- then (a) Agent or such Lender shall give Borrower Representative prompt written or electronic notice of the determination of such effect, and thereupon the obligation of Agent and Lenders to make or continue affected types of LIBOR Portions or convert Base Rate Portions to affected types of LIBOR Portions hereunder shall be suspended during the pendency of such circumstances, (b) any request for an affected type of LIBOR Portion shall be made as a Base Rate Portion unless Borrower Representative shall notify Agent, no later than 1:00 p.m. (New York City time) three Business Days prior to the date of such proposed borrowing, that the request for such borrowing shall be canceled or made as an unaffected type of LIBOR Portion, (c) any Base Rate Portion or existing LIBOR Portion which was to have been converted to an affected type of LIBOR Portion shall be continued as a Base Rate Portion unless Borrower Representative shall notify Agent, no later than 1:00 p.m. (New York City time) three Business Days prior to the proposed conversion, that the request for such conversion shall be made as an unaffected type of LIBOR Portion, and (d) Borrowers shall, promptly upon reasonable request by Agent, convert any existing affected LIBOR Portions into Base Rate Portions or unaffected types of LIBOR Portions; provided, however, that before delivering any such notice, the affected Lender agrees to use all reasonable efforts to designate a different lending office if the making of such a designation would allow such lender to continue to perform its obligations to fund, continue or maintain LIBOR Portions and would not, in the reasonable judgment of such Lender, be significantly disadvantageous to such Lender.
- 3.10 <u>Sharing of Payments, Etc.</u> If any Lender shall obtain any payment (whether voluntary, involuntary, through the exercise of any right of setoff, or otherwise) on account of any Loan made by it in excess of its ratable share of payments on account of Loans made by all Lenders, such Lender shall forthwith purchase from each other Lender such participation in such Loan as shall be necessary to cause such purchasing Lender to share the excess payment ratably with each other Lender; provided, that, if all

or any portion of such excess payment is thereafter recovered from such purchasing Lender, such purchase from each Lender shall be rescinded and such Lender shall repay to the purchasing Lenders the purchase price to the extent of such recovery, together with an amount equal to such Lender's ratable share (according to the proportion of (i) the amount of such Lender's required repayment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered. Borrowers agree that any Lender so purchasing a participation from another Lender pursuant to this Section 3.10 may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of setoff) with respect to such participation as fully as if such Lender were the direct creditor of Borrowers in the amount of such participation. Notwithstanding anything to the contrary contained herein, all purchases and repayments to be made under this Section 3.10 shall be made through Agent. For the avoidance of doubt, this Section 3.10 shall not apply to payments received by a Lender from an assignee in connection with the assignment of its Loans or from a participant in connection with the sale or settlement of such participation.

3.11 Indemnity for Returned Payments. If after receipt of any payment which is applied to the payment of all or any part of the Obligations, Agent, any Lender, Bank or any Affiliate of Bank is for any reason compelled to surrender such payment or proceeds to any Person because such payment or application of proceeds is invalidated, declared fraudulent, set aside, determined to be void or voidable as a preference, impermissible setoff, or a diversion of trust funds, or for any other reason, then the Obligations or part thereof intended to be satisfied shall be revived and continued and this Agreement shall continue in full force as if such payment or proceeds had not been received by Agent or such Lender and Borrowers shall be liable to pay to Agent and Lenders, and each Borrower hereby does indemnify Agent and Lenders and hold Agent and Lenders harmless for, the amount of such payment or proceeds surrendered. The provisions of this Section 3.11 shall be and remain effective notwithstanding any contrary action which may have been taken by Agent or any Lender in reliance upon such payment or application of proceeds, and any such contrary action so taken shall be without prejudice to Agent's and Lenders' rights under this Agreement and shall be deemed to have been conditioned upon such payment or application of proceeds having become final and irrevocable. The provisions of this Section 3.11 shall survive the termination of this Agreement.

3.12 Nature and Extent of Each Borrower's Liability.

3.12.1 <u>Joint and Several Liability</u>. Each Borrower shall be liable for, on a joint and several basis, all of the Loans and other Obligations, regardless of which Borrower actually may have received the proceeds of any Loans or other extensions of credit hereunder or the amount of such Loans received or the manner in which Agent or any Lender accounts for such Loans or other extensions of credit on its books and records, it being acknowledged and agreed that Loans to any Borrower inure to the mutual benefit of all Borrowers and that Agent and Lenders are relying on the joint and several liability of Borrowers in extending the Loans and other financial accommodations hereunder. Each Borrower hereby unconditionally and irrevocably agrees that upon

default in the payment when due (whether at stated maturity, by acceleration or otherwise) of any principal of, or interest owed on, any of the Loans or other Obligations, such Borrower shall forthwith pay the same.

- 3.12.2 Unconditional Nature of Liability. Each Borrower's joint and several liability hereunder with respect to the Loans and other Obligations shall, to the fullest extent permitted by applicable law, be the unconditional liability of such Borrower irrespective of (i) the validity, enforceability, avoidance or subordination of any of the Obligations or of any other document evidencing all or any part of the Obligations, (ii) the absence of any attempt to collect any of the Obligations from any other Borrower or any Collateral or other security therefor, or the absence of any other action to enforce the same, (iii) the waiver, consent, extension, forbearance or granting of any indulgence by Agent or any Lender with respect to any provision of any instrument executed by any other Borrower evidencing or securing the payment of any of the Obligations, or any other agreement now or hereafter executed by any other Borrower and delivered to Agent or any Lender, (iv) the failure by Agent to take any steps to perfect or maintain the perfected status of its security interest in or Lien upon, or to preserve its rights to, any of the Collateral or other security for the payment or performance of any of the Obligations or Agent's release of any Collateral or of its Liens upon any Collateral, (v) Agent's or Lenders' election, in any proceeding instituted under the Bankruptcy Code, for the application of Section 1111(b)(2) of the Bankruptcy Code, (vi) any borrowing or grant of a security interest by any other Borrower, as debtor-in-possession under Section 364 of the Bankruptcy Code, (vii) the release or compromise, in whole or in part, of the liability of any other Borrower for the payment of any of the Obligations, (viii) any increase in the amount of the Obligations beyond any limits imposed herein or in the amount of any interest, fees or other charges payable in connection therewith, in each case, if consented to by Borrower Representative, or any decrease in the same, (ix) the disallowance of all or any portion of Agent's or any Lender's claims against any other Borrower for the repayment of any of the Obligations under Section 502 of the Bankruptcy Code, or (x) any other circumstance that might constitute a legal or equitable discharge or defense of any other Borrower. After the occurrence and during the continuance of any Event of Default, Agent may proceed directly and at once, without notice to any Borrower (except as provided herein), against any or all of Borrowers to collect and recover all or any part of the Obligations, without first proceeding against any other Borrower or against any Collateral or other security for the payment or performance of any of the Obligations, and each Borrower waives any provision that might otherwise require Agent under applicable law to pursue or exhaust its remedies against any Collateral or other Borrower before pursuing another Borrower. Each Borrower consents and agrees that Agent shall be under no obligation to marshal any assets in favor of any Borrower or against or in payment of any or all of the Obligations.
 - 3.13 Lender's Obligation to Mitigate; Replacement of Lenders.
- 3.13.1 <u>Lender's Obligation to Mitigate</u>. If any Lender requests compensation under <u>Section 3.8</u>, or if Borrowers are required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 2.9</u>, then such Lender shall use all commercially reasonable efforts

(subject to applicable legal and regulatory restrictions) to mitigate or eliminate the amount of such compensation or additional amount, including by designating a different lending office for funding or booking its Loans hereunder or by assigning its rights and obligations hereunder to another of its offices, branches or Affiliates; provided that no Lender shall be required to take any action pursuant to this Section 3.13 unless, in the reasonable judgment of such Lender, such designation or assignment or other action (i) would eliminate or reduce amounts payable pursuant to Section 3.8 or 2.9, as the case may be, in the future, (ii) would not subject such Lender to any material unreimbursed cost or expense and (iii) would not otherwise be materially disadvantageous to such Lender. Borrowers shall pay all reasonable and documented costs and expenses incurred by a Lender in connection with any such designation or assignment.

3.13.2 Replacement of Lenders. If (a) any Lender requests compensation under Section 3.8, or if Borrowers are required to pay any additional amount to an Lender or any Governmental Authority for the account of any Lender pursuant to Section 2.9, or (b) any Lender defaults in its obligation to fund Loans hereunder, then Borrowers may, at their sole expense and effort, upon notice to such Lender and Agent, require such Lender to assign, without recourse (in accordance with and subject to the restrictions contained in Section 12), all its interests, rights and obligations under this Agreement to an assignee that shall accept such assignment and assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided, that: (i) the replacement Lender shall be (a) an existing Lender or (b) another financial institution reasonably acceptable to Agent; (ii) the assigning Lender shall have received payment of an amount equal to the outstanding principal of its Loans and participations in unreimbursed Letter of Credit disbursements, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (to the extent of such outstanding principal and accrued interest and fees) or Borrowers (in the case of all other amounts); (iii) the assignee shall execute an Assignment and Acceptance Agreement pursuant to which it shall become a party hereto as provided in subsection 12.1.1; and (iv) in the case of any such assignments resulting from a claim for compensation under Section 3.8 or payments required to be made pursuant to Section 2.9, such assignment will result in a material reduction in such compensation or payments. Upon compliance with the provisions for assignment provided in subsection 12.1.1 and this subsection 3.13.2, such assignee shall constitute a "Lender" hereunder and the Lender being so replaced shall no longer constitute a "Lender" hereunder. A Lender shall not be required to make any such assignment if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling Borrowers to require such assignment cease to apply.

SECTION 4. TERM AND TERMINATION

4.1 <u>Term of Agreement</u>. This Agreement shall be in effect from and including the Closing Date, through and including the Stated Termination Date (the "Term"), unless earlier terminated as provided in Section 4.2 hereof.

4.2 Termination.

- 4.2.1 <u>Termination by Agent or Lenders</u>. Agent may, and at the direction of Majority Lenders shall, terminate this Agreement immediately without notice (other than any notice required by <u>Section 10</u>) upon the occurrence and during the continuance of an Event of Default.
- 4.2.2 Termination by Borrowers . Upon at least five days' prior written notice to Agent and Lenders, Borrower Representative may, at its option, terminate this Agreement; provided , however , that no such termination shall be effective until Borrowers have paid (or collateralized to Agent's reasonable satisfaction) all of the Obligations in immediately available funds, all Letters of Credit have expired, terminated or have been collateralized in accordance with subsection 1.2.10 and Borrowers have complied with subsection 3.2.5 . Any notice of termination given by Borrower Representative shall be irrevocable unless all Lenders otherwise agree in writing and no Lender shall have any obligation to make any Loans or issue or procure any Letters of Credit on or after the termination date stated in such notice; provided , however , that any such notice of termination may state that it is conditioned upon the availability of an alternate or replacement credit facility, and that if an alternate or replacement credit facility is not obtained, such notice of termination may be revoked by Borrower Representative. Borrower Representative may elect to terminate this Agreement in its entirety only. No section of this Agreement or type of Loan available hereunder may be terminated singly.
- 4.2.3 Effect of Termination . All of the Obligations shall be immediately due and payable upon the termination date stated in any notice of termination of this Agreement. All undertakings, agreements, covenants, warranties and representations of Borrowers contained in the Loan Documents shall survive any such termination and Agent shall retain its Liens in the Collateral and Agent and each Lender shall retain all of its rights and remedies under the Loan Documents notwithstanding such termination until all Obligations have been discharged or paid, in full, in immediately available funds, including all Obligations under subsection 3.2.5 resulting from such termination.

SECTION 5. SECURITY INTERESTS

- 5.1 <u>Security Interest in Collateral</u>. To secure the prompt payment and performance to the Secured Parties of the Obligations, each Borrower hereby grants to Agent, for the benefit of the Secured Parties, a continuing lien upon and security interest in all of the following assets of such Borrower, whether now owned or existing or hereafter created, acquired or arising and wheresoever located:
 - (i) Accounts;
 - (ii) the Deposit Accounts listed on Schedule 5.1 hereto; and
 - (iii) Inventory;

together with all books, records, writings, data bases, information, Documents, and Supporting Obligations directly relating to or evidencing, embodying, or incorporating any of the foregoing, and all Proceeds of and from any of the foregoing.

- 5.2 Excluded Collateral shall not include real property, Fixtures, Equipment, Securities of Subsidiaries, the Proceeds and products of any of the foregoing Property or any other Property not specifically designated as Collateral hereby.
- 5.3 <u>Lien Perfection; Further Assurances</u>. Subject to the provisions of <u>Section 6</u> hereof, Borrowers shall promptly execute such instruments, assignments or documents and take such other actions as are necessary or are reasonably requested by Agent to perfect or to continue the perfection of Agent's security interest in the Collateral and to ensure priority of such security interest (subject to Permitted Liens). Each Borrower hereby authorizes Agent to file financing statements that indicate the Collateral as being of an equal or lesser scope, or with greater or lesser detail, than as set forth in <u>Section 5.1</u>. Each Borrower also hereby ratifies its authorization for Agent to have filed in any jurisdiction any such financing statements or amendments thereto if filed prior to the date hereof. At Agent's reasonable request, each Borrower shall also promptly execute or cause to be executed and shall deliver to Agent any and all documents, instruments and agreements to give effect to or carry out the terms or intent of the Loan Documents.

SECTION 6. COLLATERAL ADMINISTRATION

6.1 General.

- 6.1.1 <u>Location of Collateral</u>. Except as otherwise permitted below in this <u>subsection 6.1.1</u>, all Collateral will at all times be kept by Borrowers at one or more of the business locations set forth in <u>Schedule 6.1.1</u> hereto, as updated by Borrowers by providing prompt written notice to Agent of any new location. With respect to any Inventory of Borrowers located in the United States but stored or consigned on real property not owned or leased by a Borrower, the applicable Borrower shall use commercially reasonable efforts to obtain a Bailee Certificate from the owner of such real property. Borrowers shall not be required to comply with the provisions of this <u>subsection 6.1.1</u> (a) in respect of any location at which no Collateral other than Inventory having a value of less than \$500,000 is kept, and (b) in respect of Inventory which is in transit.
- 6.1.2 <u>Insurance of Collateral</u>. Borrowers shall maintain insurance (subject to customary deductibles) with financially sound and reputable insurance carriers upon their properties and assets and with respect to the business of Borrowers against such risks and in such amounts as is customary for similar businesses. Each policy of property insurance covering the Borrowers' inventory shall name Agent as loss payee as its interest may appear and shall contain a clause requiring the insurer to give not less than 10 days' prior written notice to Agent in the event of cancellation of the policy for nonpayment of premium and not less than 30 days' prior written notice to Agent in the event of cancellation of the policy for any other reason and a clause reasonably satisfactory to Agent to the effect that the interest of Agent shall not be impaired or

invalidated by any act or neglect of any Borrower or any of its Subsidiaries. On the Closing Date and within 30 days after each renewal or replacement of the policies required to be carried hereby, Borrowers shall deliver to Agent an insurance certificate in form and substance reasonably satisfactory to Agent. Unless Borrowers provide Agent with the insurance certificate required by this <u>subsection 6.1.2</u>, in addition to the other rights and remedies Agent or Lenders may have, Agent may purchase insurance (subject to customary deductibles) at Borrowers' expense to protect Agent's interests in the Collateral for the benefit of the Secured Parties. Agent shall cancel such insurance promptly following receipt of satisfactory evidence that Borrowers have obtained insurance as required by this Agreement.

6.1.3 <u>Protection of Collateral</u>. Neither Agent nor any Lender shall be liable or responsible in any way for the safekeeping of any of the Collateral or for any loss or damage thereto (except for reasonable care in the custody thereof while any Collateral is in Agent's or any Lender's actual possession) or for any diminution in the value thereof, or for any act or default of any warehouseman, carrier, forwarding agency or other person whomsoever, but, as among Borrowers, Agent and Lenders, the same shall be at Borrowers' sole risk.

6.2 Administration of Accounts.

- 6.2.1 <u>Records, Schedules and Assignments of Accounts</u>. Borrowers shall keep accurate and complete records of their Accounts and all payments and collections thereon.
- 6.2.2 <u>Account Verification</u>. After the occurrence and during the continuance of a Default or an Event of Default, any of Agent's officers, employees or agents shall have the right, at any time or times hereafter, in the name of Agent, any designee of Agent or any Borrower, to verify the validity, amount or any other matter relating to any Accounts by mail, telephone, electronic communication or otherwise. Borrowers shall cooperate fully with Agent in an effort to facilitate and promptly conclude any such verification process.
- 6.2.3 Maintenance of Dominion Account. Borrowers shall maintain the Deposit Accounts listed on Schedule 5.1 as Dominion Accounts pursuant to Blocked Account Agreements. Each Blocked Account Agreement shall provide that the applicable bank shall comply with instructions originated by Agent directing disposition of the funds in the applicable Dominion Account or Accounts without further consent by the applicable Borrower, and that, following receipt by it of a notice of exclusive control (a "Notice of Exclusive Control") from Agent, (a) such bank shall not permit any funds or other assets to be transferred or withdrawn by any Borrower from such Dominion Account or Accounts, (b) such bank shall only comply with the instructions of Agent and no longer comply with instructions of any Borrower in respect of such Dominion Account or Accounts, and (c) such bank shall transfer all payments or other remittances received in the Dominion Account or Accounts to Agent's account for application on account of the Obligations as provided in subsection 3.2.1 and Section 3.4. Agent agrees that (x) it shall not deliver a Notice of Exclusive Control unless (i) a Default or Event of

Default has occurred and is continuing at the time or (ii) Availability falls below \$15,000,000, and (y) it shall provide Borrower Representative with prompt notice of its delivery of a Notice of Exclusive Control, which notice shall include a statement specifying with particularity the basis for delivery thereof (i.e., either that a Default or Event of Default has occurred and is continuing and the nature of the Default or Event of Default or that Availability has fallen below \$15,000,000); provided that the failure of Agent to deliver such notice to Borrowers shall not in any manner affect the validity and enforceability of any Notice of Exclusive Control; and provided, further that Agent's exclusive control shall be rescinded at such time as (A) Availability is equal to or greater than \$15,000,000 (determined using the most recent Borrowing Base Certificate delivered by Borrower Representative, either at the time required by subsection 8.1.4 hereof or voluntarily at any time for this purpose) and (B) no Default or Event of Default shall have occurred and be continuing, in each case for a period of 30 consecutive days. Agent assumes no responsibility for such blocked account arrangements, including any claim of accord and satisfaction or release with respect to deposits accepted by any bank thereunder.

- 6.2.4 Collection of Accounts, Proceeds of Collateral. Each Borrower agrees that all invoices rendered on and after the Closing Date and other requests made by any Borrower on or after the Closing Day for payment in respect of Accounts shall contain a written statement directing payment in respect of such Accounts to be paid to a lockbox or blocked account established pursuant to subsection 6.2.3. If a Borrower receives remittances on account of Accounts other than by direct payment to a Dominion Account, such Borrower shall cause such remittances to be deposited in a Dominion Account as soon as practicable. Agent retains the right at all times after the occurrence and during the continuance of an Event of Default to notify Account Debtors that Borrowers' Accounts have been assigned to Agent and to collect Borrowers' Accounts directly in its own name, or in the name of Agent's agent, and to charge the collection costs and expenses, including attorneys' fees, to Borrowers.
- 6.2.5 <u>Taxes</u>. If an Account includes a charge for any tax payable to any Governmental Authority, Agent is authorized, in its sole discretion, after the occurrence and during the continuance of an Event of Default, to pay the amount thereof to the proper Governmental Authority for the account of the applicable Borrower and to charge Borrowers therefor, except for taxes that (i) are being actively contested in good faith and by appropriate proceedings and with respect to which Borrowers maintain reasonable reserves on its books therefor and (ii) would not reasonably be expected to result in any Lien other than a Permitted Lien. In no event shall Agent or any Lender be liable for any taxes to any Governmental Authority that may be due by any Borrower.
- 6.3 <u>Administration of Inventory</u>. Borrowers shall keep records of their Inventory which records shall be complete and accurate in all material respects, and shall conduct a physical inventory no less frequently than annually.
- 6.4 <u>Payment of Charges</u>. All amounts properly chargeable to Borrowers under this <u>Section 6</u> shall be Obligations, shall be payable on demand and shall bear

interest from the date such advance was made until paid in full at the rate applicable to Base Rate Portions from time to time.

SECTION 7. REPRESENTATIONS AND WARRANTIES

- 7.1 <u>General Representations and Warranties</u>. To induce Agent and each Lender to enter into this Agreement and to make advances hereunder, Borrowers represent and warrant to Agent and each Lender, on a joint and several basis, that:
- 7.1.1 Organization, Existence and Qualification. Each Loan Party is a corporation, general partnership, limited partnership, or limited liability company, as applicable, duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation or organization. Each Loan Party is duly qualified and is authorized to do business and is in good standing as a foreign limited liability company, limited partnership or corporation, as applicable, in all states and jurisdictions in which the failure of such Loan Party to be so qualified would reasonably be expected to have a Material Adverse Effect.
- 7.1.2 Power and Authority; No Conflict. Each Loan Party has all requisite power and authority to conduct its business as currently conducted and own its Property and is duly authorized and empowered to enter into, execute, deliver and perform each of the Loan Documents to which it is a party. The execution, delivery and performance by each Loan Party of each of the Loan Documents to which it is a party have been duly authorized by all necessary corporate or other relevant action and do not and will not: (i) contravene, violate or result in a breach of or default under (a) any Loan Party's charter, articles or certificate of incorporation, certificate of formation, bylaws, limited liability company or partnership agreement, or other organizational documents (as the case may be), (b) any provision of any law, rule, regulation, order of any Governmental Authority, writ, judgment, injunction, decree, determination or award in effect having applicability to such Loan Party, the violation of which would reasonably be expected to have a Material Adverse Effect, or (c) any indenture or loan or credit agreement or any other agreement, lease or instrument binding on a Loan Party or its Properties, the breach of or default under which would reasonably be expected to have a Material Adverse Effect; or (ii) result in, or require, the creation or imposition of any Lien (other than Permitted Liens) upon or with respect to any of the Collateral now owned or hereafter acquired by such Loan Party.
- 7.1.3 <u>Legally Enforceable Agreement</u>. This Agreement is, and each of the other Loan Documents when executed and delivered will be, a legal, valid and binding obligation of each Loan Party party hereto or thereto, enforceable against it in accordance with its respective terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, or fraudulent transfer laws, or other similar laws affecting creditors' rights generally, and by general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

- 7.1.4 <u>Organizational Structure</u>. <u>Schedule 7.1.4</u> hereto contains, as of the date of this Agreement, a true and complete organizational structure chart of the Loan Parties and their respective Subsidiaries, which organizational structure chart includes the exact legal name of each Loan Party and each of its Subsidiaries and the percentage of Voting Stock owned by each owner of Voting Stock of each such entity (other than Century).
- 7.1.5 Names; Organization . Except as set forth on Schedule 7.1.5 , none of Borrowers has within the five years immediately preceding the Closing Date (a) used any legal, fictitious or trade names, or (b) been the surviving entity of a merger or consolidation or acquired all or substantially all of the assets of any Person. Each of each Borrower's state(s) of incorporation or organization, Type of Organization and Organizational I.D. Number is set forth on Schedule 7.1.5.
- 7.1.6 <u>Business Locations; Inventory</u>. Each of each Borrower's chief executive office, location of books and records and other places of business are as listed on <u>Schedule 6.1.1</u> hereto, as updated from time to time by Borrowers in accordance with the provisions of <u>subsection 6.1.1</u>. Except as shown on <u>Schedule 6.1.1</u> hereto, as updated from time to time by Borrowers in accordance with the provisions of <u>subsection 6.1.1</u>, no Inventory with a value in excess of \$500,000 is stored with a bailee, distributor, warehouseman or similar party, nor is any Inventory consigned to any Person.
- 7.1.7 <u>Title to Properties; Priority of Liens</u>. Each Borrower has good title to all of the Collateral owned by it from time to time free and clear of all Liens except Permitted Liens. The provisions of this Agreement and the other Loan Documents create valid Liens on the Collateral in favor of Agent, for the ratable benefit of the Secured Parties, and upon the filing of the financing statements and the consummation of the other actions listed on <u>Schedule 7.1.7</u> hereto, such Liens on the Collateral shall be perfected Liens having priority over all other Liens on the Collateral other than Permitted Liens having priority by operation of law.
- 7.1.8 Financial Statements; Absence of Material Adverse Change; Fiscal Year. Borrowers have delivered to Agent the (a) audited financial statements of Century and its Subsidiaries as of December 31, 2004, on a Consolidated basis prepared in accordance with GAAP, (b) unaudited interim financial statements of Century and its Subsidiaries as of June 30, 2005, on a Consolidated basis prepared in accordance with GAAP, and (c) unaudited interim financial statements of Borrowers and Guarantors as of June 30, 2005, on a consolidating basis. All such financial statements present fairly in all material respects the financial positions of such Persons at such dates and the results of such Persons' operations for such periods. As of the date of this Agreement, since June 30, 2005, there has been no material adverse change in the financial position of Century and its Subsidiaries, taken as a whole. The fiscal year of each Borrower and Guarantor ends on December 31 of each year.
- 7.1.9 <u>Full Disclosure</u>. Neither the Loan Documents nor the financial statements made or delivered by any Loan Party to Agent or any Lender on or prior to the date of this Agreement contain any untrue statement of a material fact or omit a material

fact necessary to make such statements or information not misleading in light of the circumstances under which such statements were made.

- 7.1.10 <u>Solvency</u>. Each Loan Party is, and, after giving effect to the initial Loans and any other Loans made hereunder and the initial Letters of Credit and any other Letters of Credit to be issued hereunder and all related transactions, will be, Solvent.
- 7.1.11 <u>Taxes</u>. Each Borrower (a) has filed all federal, state and local tax returns and other reports relating to taxes it is required by law to file and except for any such filing the failure of which to file would not reasonably be expected to have a Material Adverse Effect, and (b) has paid, or made provision for the payment of, all taxes, assessments, fees, levies and other governmental charges upon it, its income and Properties, except for any such payment or provision the failure of which to pay or make would not reasonably be expected to have a Material Adverse Effect.
- 7.1.12 <u>Intellectual Property Relating to Inventory</u>. Each Borrower owns, possesses or licenses or has the right to use all Intellectual Property necessary in such Borrower's reasonable judgment for the production, processing, use, and sale or other disposition of its Inventory (its "<u>Inventory IP</u>") without any known infringement upon the intellectual property rights of others, except for any such infringement that as would not reasonably be expected to have a Material Adverse Effect.
- 7.1.13 Governmental Consents. Each Loan Party has, and is in good standing with respect to, all governmental consents, approvals, licenses, authorizations, permits, certificates, inspections and franchises necessary to continue to conduct its business as heretofore or proposed to be conducted by it and to own or lease and operate its Properties as now owned or leased by it, except where any failure to so possess, have, or maintain any of the foregoing would not reasonably be expected to have a Material Adverse Effect.
- 7.1.14 Compliance with Laws. Each Loan Party has duly complied, and its Properties, business operations and leaseholds are in compliance with, the provisions of all federal, state and local laws, rules and regulations applicable to such Borrower, its Properties or the conduct of its business, except for any such non-compliance that would not reasonably be expected to have a Material Adverse Effect. There have been no citations, notices or orders of noncompliance with any applicable laws issued by any Governmental Authority to any Loan Party under any such law, rule or regulation, except for any such citations, notices or orders in respect of noncompliance the failure to comply with which would not reasonably be expected to have a Material Adverse Effect.
 - 7.1.15 Restrictive Agreements. No Loan Party is a party to or subject to any Restrictive Agreements.
- 7.1.16 <u>Litigation</u>. Except as set forth on <u>Schedule 7.1.16</u> hereto or in Century's form 10-K or any form 10-Q filed with the SEC on or after January 1, 2005

and before the Closing Date, there are no actions, suits, proceedings or investigations pending, or to the knowledge of any Borrower, threatened, against or directly affecting any Loan Party, or the business, operations, Properties, prospects, profits or condition of any Loan Party, which, individually or in the aggregate would reasonably be expected to be adversely determined, and, if adversely determined, would reasonably be expected to have a Material Adverse Effect. No Loan Party is in default with respect to any order, writ, injunction, judgment, decree or rule of any Governmental Authority, which default, individually or in the aggregate, if not cured, would reasonably be expected to have a Material Adverse Effect.

- 7.1.17 <u>ERISA</u>. No ERISA Event has occurred or is reasonably expected to occur that, when taken together with all other ERISA Events that have occurred, would reasonably be expected to have a Material Adverse Effect.
- 7.1.18 No Defaults . As of the Closing Date, no event has occurred and no condition exists which would, upon or after the execution and delivery of the Loan Documents or any Loan Party's performance hereunder, constitute a Default or an Event of Default.
- 7.1.19 <u>Related Businesses</u>. As of the Closing Date, Borrowers are primarily engaged in the business of producing value-added and standard-grade primary aluminum products, bauxite mining, and alumina refining and businesses and activities related thereto.
- 7.1.20 <u>Margin Regulations</u>. No Loan Party is engaged in the business of purchasing or selling Margin Stock or extending credit for the purpose of purchasing or carrying Margin Stock.
- 7.1.21 <u>Regulated Entities</u>. No Borrower, nor any Subsidiary of any Borrower, is an "investment company" within the meaning of the Investment Company Act of 1940. No Borrower or Subsidiary of any Borrower is subject to regulation under the Public Utility Holding Company Act of 1935, the Federal Power Act, or the Interstate Commerce Act.

7.1.22 Anti-Terrorism Laws.

- (a) <u>General</u>. None of the Loan Parties, any of their Subsidiaries or, to its knowledge, any of their Affiliates, is in violation of any Anti-Terrorism Law or to its knowledge engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any Anti-Terrorism Law.
- (b) <u>Executive Order No. 13224</u>. None of the Loan Parties, their Subsidiaries or, to its knowledge, any of their Affiliates, is any of the following (each a "<u>Blocked Person</u>"):

- (i) a Person controlled by, or acting for or on behalf of, any Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;
- (ii) a Person with which any Lender is expressly prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law;
 - (iii) a Person that committed "terrorism" as defined in Executive Order No. 13224; or
- (iv) a Person that is named as a "specially designed national" on the most current list published by the U.S. Treasury Department Office of Foreign Assets Control at its official website or any replacement website or other replacement official publication of such list.

None of the Loan Parties, their Subsidiaries or, to its knowledge, any of their Affiliates, (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or to its knowledge for the benefit of any Blocked Person or (ii) deals in, or otherwise engages in any transaction involving any property or interests in property expressly blocked pursuant to Executive Order No. 13224.

- 7.2 <u>Reaffirmation of Representations and Warranties</u>. Borrowers acknowledge that each Loan request made or deemed made hereunder shall constitute Borrowers' reaffirmation, as of the date of each such Loan request, of the truth and correctness in all material respects of each representation and warranty made or furnished to Agent or any Lender by or on behalf of any Loan Party in each of the Loan Documents (except to the extent any such representation or warranty expressly relates to an earlier date, in which case such reaffirmation shall be of the truth and correctness in all material respects of such representation or warranty as of such earlier date).
- 7.3 <u>Survival of Representations and Warranties</u>. All representations and warranties of Borrowers contained in this Agreement or any of the other Loan Documents shall survive the execution, delivery and acceptance thereof by Agent and each Lender and the parties thereto and the closing of the transactions described therein or related thereto.

SECTION 8. COVENANTS AND CONTINUING AGREEMENTS

- 8.1 <u>Affirmative Covenants</u>. During the Term, and thereafter for so long as there are any Obligations outstanding (other than contingent indemnity and expense reimbursement obligations for which no claim has been made), Borrowers jointly and severally covenant that they shall, unless otherwise consented to by Majority Lenders in writing in accordance with subsection 12.2.1:
- 8.1.1 <u>Visits and Inspections</u>. Subject to the limitations set forth in <u>Section 2.6</u> and <u>Section 6</u> permit representatives of Agent, and during the continuance of any Default or Event of Default any Lender, from time to time, as often as may be

reasonably requested, but only during normal business hours and without undue interruption of normal business activities, to visit and inspect the facilities of each Borrower, inspect and make extracts from its books and records, and discuss with its officers, each Borrower's business, assets, liabilities, financial condition and results of operations. Agent, if no Default or Event of Default then exists, shall give Borrowers reasonable prior notice of any such inspection.

- 8.1.2 <u>Notices</u>. Promptly, and in any event within three Business Days after a Responsible Officer of any Borrower obtains knowledge thereof, notify Agent in writing of the occurrence of (i) any Default or Event of Default, and (ii) any development that has had, or would reasonably be expected to have, a Material Adverse Effect.
- 8.1.3 <u>Records and Books</u>; <u>Financial Statements</u>. Keep adequate records and books of account with respect to its business activities in which proper entries are made in accordance with customary accounting practices reflecting its financial transactions; and cause to be prepared and furnished to Agent, the following:
 - (a) not later than 90 days after the close of each fiscal year of Borrowers, (A) unqualified (except for a qualification for a change in accounting principles with which the accountant concurs) audited financial statements of Century and its Subsidiaries as of the end of such year, on a Consolidated basis, certified by Deloitte & Touche or another firm of independent certified public accountants of recognized standing selected by Borrowers and reasonably acceptable to Agent (which financial statements (1) shall be prepared in accordance with GAAP, applied on a consistent basis, unless Borrowers' certified public accountants concur in any change therein and such change is disclosed to Agent and is consistent with GAAP, and (2) shall not contain any paragraph of emphasis or explanatory note calling in to question the ability of Century to continue as a going concern) and (B) unaudited financial statements of Borrowers and Guarantors for the fourth quarter of such fiscal year, on a consolidating basis (it being understood that the Hawesville Entities shall be presented as a single consolidating entity) certified by a Financial Officer of Century as fairly presenting in all material respects the financial position and results of operations of Century and its applicable Subsidiaries for such fiscal quarter, subject only to changes from audit and year-end adjustments and except that such statements need not contain notes;
 - (b) not later than 45 days after the end of Borrowers' first three fiscal quarters, unaudited interim financial statements (A) of Century and its Subsidiaries, on a Consolidated basis (which financial statements shall be prepared in accordance with GAAP, applied on a consistent basis, unless Borrowers' certified public accountants concur in any change therein and such change is disclosed to Agent and is consistent with GAAP), and (B) of Borrowers and Guarantors, on a consolidating basis (it being understood that the Hawesville Entities shall be presented as a single consolidated entity), in each case as of the end of such fiscal quarter and for the portion of the fiscal year then elapsed, and in

each case certified by a Financial Officer of Century as fairly presenting in all material respects the financial position and results of operations of Century and its applicable Subsidiaries for such fiscal quarter and period subject only to changes from audit and year-end adjustments and except that such statements need not contain notes;

- (c) from and after such time as average Availability is below \$50,000,000 for any period of 15 consecutive days, and until such time thereafter as average Availability is at least \$50,000,000 for any period of 30 consecutive days, not later than 30 days after the end of each month, including the last month of Borrowers' fiscal year, unaudited interim financial statements of Century and its Subsidiaries as of the end of such month on a Consolidated basis, certified by a Financial Officer of Century and fairly presenting in all material respects the financial position and results of operations of Century and its applicable Subsidiaries for such month and period subject only to changes from audit and year-end adjustments and except that such statements need not contain notes;
- (d) promptly after the sending or filing thereof, as the case may be, copies of any proxy statements, financial statements or reports of a financial nature which Borrower Representative has made available to its Securities holders generally and copies of any regular, periodic and material special reports or registration statements which Borrower Representative or any of its Subsidiaries files with the SEC or any Governmental Authority which may be substituted therefor or any national securities exchange; and
- (e) promptly after request, such other data and information (financial and otherwise) as Agent or a Lender may reasonably request from time to time, bearing upon or related to the Collateral or Borrowers' financial condition or results of operations.

Information required to be furnished pursuant to <u>subsections 8.1.3(a)</u>, (b) and (d) above shall be deemed to have been delivered on the date on which Century provides notice to Agent that such information has been posted on Century's website on the Internet at the website address listed on the signature pages hereof, at sec.gov/edaux/searches.htm or at another website identified in such notice and accessible by Lenders without charge; provided that (A) such notice may be included in a certificate delivered pursuant to <u>subsection 8.1.3(b)</u>, and (B) Borrowers shall deliver paper copies of the information referred to in <u>subsections 8.1.3(a)</u>, (b) and (d) to Agent if Agent requests delivery thereof.

Concurrently with furnishing the financial statements described in <u>subsections 8.1.3(a)</u> and <u>(b)</u>, or more frequently if reasonably requested by Agent, Borrowers shall cause to be prepared and furnished to Agent a Compliance Certificate.

8.1.4 <u>Borrowing Base Certificates</u>. (a) On or before the 15th day of each fiscal quarter of Borrowers, and (b) from and after such time as average Availability is below \$50,000,000 for any period of 15 consecutive days, and until such time

thereafter as average Availability is at least \$50,000,000 for any period of 30 consecutive days, but without duplication, on or before the 15th day of each month, Borrower Representative shall deliver to Agent a Borrowing Base Certificate as of the last day of the immediately preceding fiscal quarter (or month, in the case of Borrowing Base Certificates delivered under clause (b) above). If Borrowers deem it advisable, Borrowers may execute and deliver to Agent Borrowing Base Certificates more frequently than as required pursuant to this subsection 8.1.4.

- 8.1.5 <u>Projections</u>. No later than 60 days following the first day of each fiscal year of Century, deliver to Agent Projections for Century and its Subsidiaries, on a Consolidated basis, and for Borrowers and Guarantors, on a Consolidated basis, covering such fiscal year on a month-by-month basis.
- 8.1.6 <u>Taxes and Other Obligations</u>. (a) File when due all tax returns and other reports which any of them are required to file and the failure of which to file would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; (b) pay, or provide for the payment when due of, all taxes, fees, assessments and other governmental charges against it or upon its property, income and franchises, the failure of which to pay would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and (c) pay when due all Indebtedness owed by it and all claims of materialmen, mechanics, carriers, warehousemen, landlords, processors and other like Persons, and all other indebtedness owed by it and perform and discharge in a timely manner all other obligations undertaken by it, in each case the failure of which to pay, perform, or discharge would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.
- 8.1.7 <u>Legal Existence and Good Standing</u>. Maintain their respective legal existences and qualifications and good standing in all jurisdictions in which the failure to maintain such existence and qualification or good standing would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.
- 8.1.8 <u>Compliance with Law</u>. Comply with all Requirements of Law of any Governmental Authority having jurisdiction over them or their businesses except where the failure to comply with such Requirements of Law would not reasonably be expected to have a Material Adverse Effect.
- 8.1.9 <u>Maintenance of Property and Licenses</u>. (a) Maintain all of their Property material to the conduct of their businesses in good operating condition and repair, ordinary wear and tear excepted, and (b) obtain and maintain in effect at all times all franchises, governmental authorizations, Intellectual Property rights, licenses, and permits, in each case which are necessary for them to own their Property or conduct their business, except where the failure to obtain and/or maintain any of the foregoing would not be reasonably expected to have a Material Adverse Effect.
- 8.2 <u>Negative Covenants</u>. During the Term, and thereafter for so long as there are any Obligations outstanding (other than contingent indemnity and expense reimbursement obligations for which no claim has been made), Borrowers covenant that they shall not, unless otherwise consented to by Majority Lenders in writing in accordance with subsection 12.2.1:

- 8.2.1 <u>Mergers; Consolidations; Acquisitions; Structural Changes</u>. Merge or consolidate with any Person; nor change their states of incorporation or organization, Types of Organization or Organizational I.D. Numbers; nor change their legal names; nor acquire all or any substantial part of the Properties of any Person, except for:
 - (i) subject to compliance with <u>subsection 8.2.8</u>, mergers of any Person into a Borrower where the Borrower is the surviving entity;
 - (ii) acquisitions of assets consisting of fixed assets or real Property that constitute Capital Expenditures permitted under <u>subsection</u> 8.2.6;
 - (iii) acquisitions of all or any substantial part of the Securities or assets of any Person to the extent permitted under $\underline{\text{subsection 8.2.8}}$; and
 - (iv) changes in legal name, state of incorporation or organization, Type of Organization or Organizational I.D. made after 30 days' prior written notice to Agent.
 - 8.2.2 <u>Indebtedness</u>. Create, incur, assume, or suffer to exist any Indebtedness, except:
 - (i) Obligations owing to Agent or any Lender under this Agreement or any of the other Loan Documents;
 - (ii) Indebtedness existing on the date of this Agreement and listed on Schedule 8.2.2;
 - (iii) Permitted Purchase Money Indebtedness and Capital Lease Obligations (to the extent permitted by the definition of Permitted Purchase Money Indebtedness);
 - (iv) contingent liabilities arising out of endorsements of checks and other negotiable instruments for deposit or collection in the ordinary course of business;
 - (v) Guaranties of any Indebtedness permitted hereunder;
 - (vi) Indebtedness in respect of intercompany loans permitted under subsection 8.2.8;
 - (vii) Indebtedness consisting of the deferred purchase price for power or insurance premiums pursuant to any provision in a power contract or insurance policy (or related agreement) that permits payment of a portion thereof to be deferred;

- (viii) Indebtedness incurred to repurchase stock to the extent permitted by subsection 8.2.5;
- (ix) Derivative Obligations entered into for bona fide hedging purposes and not for speculative purposes;
- (x) Indebtedness to the extent not included in clauses (i) through (x) above, which is specified as "Permitted Debt" under Section 4.06 of the Indenture as in effect on the date of this Agreement; and
- (xi) renewals, extensions, and refinancings of Indebtedness permitted by this <u>subsection 8.2.2</u>; <u>provided</u> that any such renewal, extension, or refinancing is in an aggregate principal amount not greater than the principal amount of, and is on terms no less favorable taken as a whole to Borrower obligated thereunder, including as to weighted average maturity and final maturity, than the Indebtedness being renewed, extended, or refinanced.

Borrowers shall cause any agreements in respect of indebtedness secured by any Borrower's real property to contain provisions permitting Agent to access the premises in connection with the exercise of its remedies upon default (such provisions to be consistent with similar provisions contained in Section 5.02(a)(v) of the Deed of Trust).

- 8.2.3 <u>Affiliate Transactions</u>. Enter into, or be a party to, or permit any Guarantor to enter into or be a party to, any transaction with any Affiliate of Borrower, including any agreement or arrangement for payment of management, consulting or similar fees, except if such transaction would be permitted to be entered into by Century or its "Restricted Subsidiaries" pursuant to Section 4.14 of the Indenture as in effect on the date of this Agreement.
- 8.2.4 <u>Limitation on Liens</u>. Create or suffer to exist any Lien upon any of the Collateral, except (each of the following, a "Permitted Lien"):
 - (i) Liens at any time granted in favor of Agent for the benefit of the Secured Parties;
 - (ii) Liens for taxes, assessments or governmental charges (excluding any Lien imposed pursuant to any of the provisions of ERISA) which are not yet due, or which are being contested in good faith and by appropriate proceedings, and for which the applicable Borrower maintains reasonable reserves on its books;
 - (iii) Liens arising in the ordinary course of the business of any Borrower by operation of law or regulation (including liens of carriers, warehousemen, mechanics, materialmen and other like Liens), (A) securing obligations that are not overdue by more than 30 days or which are being contested in good faith by appropriate proceedings and for which the affected Borrower maintains reasonable reserves on its books and (B) which Liens do not, in the aggregate, materially detract from the value of the Collateral of such

Borrower or materially impair the use thereof in the operation of the business of such Borrower;

- (iv) Liens existing on the date of this Agreement and identified on Schedule 8.2.4;
- (v) Liens, claims, demands and arrangements under the Mt. Holly Owners Agreement, so long as such Liens, claims, demands and arrangements (A) do not relate to the Accounts, and (B) to the extent they relate to Inventory of Borrowers, such Liens, claims, demands and arrangements are subject to a Bailee Certificate between Agent and Alumax of South Carolina, Inc.;
- (vi) Liens incurred or deposits made in the ordinary course of business in connection with (A) workers' compensation, social security, unemployment insurance, pension and other like laws or (B) contracts, leases, statutory obligations, work in progress advances, bids, tenders, indemnity or performance bonds and other similar obligations incurred in the ordinary course of business and which are not incurred in connection with the borrowing of money or the payment of the deferred purchase price of Property;
 - (vii) judgment Liens that do not give rise to an Event of Default under subsection 10.1.13; and
 - (viii) such other Liens as Majority Lenders may hereafter approve in writing.
 - 8.2.5 Distributions. Declare or make, or permit any Guarantor to declare or make, any Distributions, except for:
 - (i) Distributions by any Person to a Borrower;
 - (ii) Distributions paid solely in Securities of the Borrower or Guarantor making the Distribution;
- (iii) Distributions by Century which would be permitted to be made by Century pursuant to Sections 4.07(b)(vi) and 4.07(b)(x) of the Indenture as in effect on the date of this Agreement;
- (iv) Distributions not included in paragraphs (i) through (iii) above; <u>provided</u> that, (i) no Default or Event of Default has occurred and is continuing and (ii) the total of (A) Availability, <u>plus</u> (B) immediately available funds in bank accounts of the Borrowers and readily marketable investments of the Borrowers of the type described in clauses (v) through (viii) of the definition of the term "Restricted Investments", <u>minus</u> (C) the amount of the proposed Distribution, shall (x) have averaged at least \$25,000,000 over the immediately preceding 30 days (determined using the Borrowing Base Certificate most recently delivered by Borrowers pursuant to <u>subsection 8.1.4</u>), and (y) be at least \$25,000,000 immediately following the proposed Distribution. For purposes of

calculating the average Availability over the preceding 30 days on a pro forma basis, (A) (1) net cash proceeds from equity financing, Indebtedness, or the sale of assets other than Collateral permitted to be raised hereunder and raised not more than 30 days prior to the date of the proposed Distribution and either immediately applied to the proposed Distribution or (without duplication) held in one or more bank accounts of the Borrowers or invested in readily marketable investments of the type described in clauses (v) through (viii) of the definition of the term "Restricted Investments" and (2) the amount of the proposed Distribution shall each be deemed to have been raised and distributed, as the case may be, on the first day of such 30-day period, and (B) all other immediately available funds and readily marketable investments of the type described in clauses (v) through (viii) of the definition of the term "Restricted Investments" shall be included in the foregoing calculation based on whether or not actually existing in the Borrowers' bank accounts or investment accounts on each date with respect to which the calculation is made; and

- (v) the payment of Distributions within 60 days after the date of declaration thereof, provided that the declaration thereof was permitted under this <u>subsection 8.2.5</u>.
- 8.2.6 <u>Capital Expenditures</u>. Make Capital Expenditures (including by way of capitalized leases) which, in the aggregate as to all Loan Parties, exceed \$30,000,000 during any fiscal year of Borrowers; <u>provided</u>, that the forgoing restriction shall only be effective from and after such time as Availability is below \$25,000,000, and until such time thereafter as Availability is at least \$25,000,000.
 - 8.2.7 <u>Disposition of Collateral</u>. Sell, lease or otherwise dispose of any of the Collateral to or in favor of any Person, except for:
 - (i) sales of Inventory in the ordinary course of business;
 - (ii) dispositions of Inventory that is damaged, obsolete, uneconomic, off-specification, or no longer used or useful so long as Borrowers comply with <u>subsection 3.3.2</u> in connection with such disposition;
 - (iii) transfers solely among Borrowers; and
 - (iv) other dispositions expressly authorized by this Agreement.
 - 8.2.8 Restricted Investment. Make or have, or permit any Guarantor to make or have, any Restricted Investment.
- 8.2.9 <u>Organizational Documents</u>. Agree to, or suffer to occur, any amendment, supplement or addition to its charter, articles or certificate of incorporation, certificate of formation, partnership agreement, bylaws, limited liability agreement, operating agreement or other organizational documents (as the case may be), that would reasonably be expected to have a Material Adverse Effect.

- 8.2.10 Fiscal Year End. Change its fiscal year end.
- 8.2.11 <u>Business Conducted</u>. Engage, directly or indirectly, in any line of business other than the businesses in which Borrowers are engaged on the Closing Date and Related Businesses.
 - 8.2.12 <u>Restrictive Agreements</u>. Enter into any Restrictive Agreements.
- 8.2.13 <u>Use of Proceeds</u>. Use any portion of the Loan proceeds, directly or indirectly, for any purpose other than (i) the satisfaction of existing Indebtedness of Borrowers, (ii) to issue standby or commercial letters of credit, (iii) to finance permitted Capital Expenditures, (iv) to finance ongoing working capital needs, and (v) for other general corporate purposes, including any purpose expressly permitted by this Agreement. Without limitation of the foregoing provision, Borrowers shall not use any portion of the Loan proceeds, directly or indirectly, (w) to purchase or carry Margin Stock, (x) to repay or otherwise refinance indebtedness of any Borrowers or others incurred to purchase or carry Margin Stock, (y) to extend credit for the purpose of purchasing or carrying any Margin Stock, or (z) to acquire any security in any transaction that is subject to Section 13 or 14 of the Exchange Act.

SECTION 9. CONDITIONS PRECEDENT

- 9.1 Conditions Precedent to Closing and Initial Loans and Letters of Credit. Notwithstanding any other provision of this Agreement or any of the other Loan Documents, and without affecting in any manner the rights of Agent or any Lender under the other sections of this Agreement, no Lender shall be required to make any Loan, nor shall Agent be required to issue or procure any Letter of Credit unless and until each of the following conditions has been and continues to be satisfied on the Closing Date:
 - 9.1.1 Documentation . Agent shall have received a duly executed copy of this Agreement and the other Loan Documents.
 - 9.1.2 No Default . No Default or Event of Default shall have occurred and be continuing.
- 9.1.3 Other Conditions. Each of the other conditions precedent required by the Loan Documents to have been satisfied on or prior to the Closing Date shall have been satisfied.
- 9.1.4 <u>Availability</u>. Agent shall have determined that as of the Closing Date and immediately after giving effect to any Loans to be made and Letters of Credit to be issued on the Closing Date, and the payment by Borrowers of all closing costs incurred in connection with the transactions contemplated hereby, Availability shall not be less than \$50,000,000.
- 9.1.5 No Litigation . No action, proceeding, investigation, regulation or legislation shall have been instituted, threatened or proposed before any court,

governmental agency or legislative body to enjoin, restrain or prohibit, or to obtain damages in respect of, or which is related to or arises out of this Agreement or the consummation of the transactions contemplated hereby.

- 9.1.6 <u>Material Adverse Effect</u>. As of the Closing Date, since June 30, 2005, no event or condition exists which has resulted in or would be reasonably likely to result in a Material Adverse Effect.
- 9.1.7 <u>Closing Fees</u>. On or prior to the Closing Date, (i) Agent shall have received, for its account or the account of Lenders, as applicable, payment in full by Borrowers of the fees required to be paid to Agent or Lenders under or in connection with this Agreement on the Closing Date (including the fees due on the Closing Date pursuant to the Fee Letter), and (ii) Agent shall have received evidence satisfactory to Agent that the reasonable and documented fees and expenses of Agent's counsel incurred or accrued through the Closing Date have been paid by Borrowers <u>provided</u>, that Agent shall have delivered the documentation of such fees and expenses to Borrower Representative not later than two Business Days prior to the Closing Date.
- 9.1.8 Other. On or prior to the Closing Date, Agent shall have received each of the following in form and substance reasonably satisfactory to it (and duly executed by each of the parties thereto, to the extent applicable):
 - (i) One or more Blocked Account Agreements with respect to the Deposit Accounts listed on Schedule 5.1;
 - (ii) Guaranty Agreement executed by each of the Guarantors in favor of Agent;
 - (iii) Certificate of a Secretary or other appropriate officer of Borrowers and Guarantors certifying as to (a) Certificate of Incorporation (or equivalent organizational document), (b) Bylaws (or equivalent organizational documents), and (c) resolutions approving the transactions contemplated hereby;
 - (iv) Certificate of a Secretary or other appropriate officer of Borrowers and Guarantors certifying as to the incumbency of each of the officers of Borrowers and Guarantors executing the Loan Documents;
 - (v) Certificate of a Vice President or other appropriate officer of Borrowers and Guarantors certifying as to representations and warranties, no Default, Solvency, and other matters;
 - (vi) Certificate of a Vice President or other appropriate officer of Borrowers and Guarantors certifying as to matters relating to the USA Patriot Act;
 - (vii) Good standing certificates for each of the Borrowers and Guarantors from the Secretary of State of its respective jurisdiction of organization;

- (viii) UCC terminations as shall be necessary to evidence the release of all Collateral except in connection with Permitted Liens;
- (ix) UCC financing statements covering the Collateral and naming each of the Borrowers as debtor, in due form for filing with the Secretary of State of the jurisdiction of organization of such Borrower;
 - (x) Evidence of all insurance coverage required by the Loan Documents;
- (xi) Completed initial Borrowing Base Certificate indicating that immediately after Lenders have made the initial Loans and after Agent has issued or procured the initial Letters of Credit contemplated hereby, and Borrowers have paid (or, if accrued, treated as paid), all closing costs incurred in connection with the transactions contemplated hereby, Availability shall not be less than \$50,000,000;
 - (xii) Fee Letter;
 - (xiii) Payoff letter terminating and releasing the Liens created under the Borrowers' existing credit facility;
 - (xiv) Opinion of Curtis, Mallet-Prevost, Colt & Mosle LLP, special New York counsel to the Borrowers and the Guarantors; and
 - (xv) Opinion of Bowles Rice McDavid Graff & Love LLP, special Kentucky counsel to NSA.
- 9.2 <u>Conditions Precedent to Each Loan and Letter of Credit</u>. The obligation of each Lender on any date (including the Closing Date) to make any Loan and of Letter of Credit Issuer on any date (including the Closing Date) to issue any Letter of Credit is subject to the satisfaction of each of the following conditions precedent:
- 9.2.1 <u>Representations and Warranties; No Default</u>. Both before and after giving effect thereto and, in the case of any Loan, to the application of the proceeds thereof:
 - (a) the representations and warranties set forth in each of the Loan Documents shall be true and correct in all material respects on and as of the Closing Date and on and as of such date with the same effect as though made on and as of such date (except to the extent such representations and warranties by their terms expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct, in all material respects, as of such earlier date); and
 - (b) No Default or Event of Default shall exist.

9.2.2 <u>Borrowing Base Certificate</u>. Borrowers shall have delivered the Borrowing Base Certificate most recently required to have been delivered by them pursuant to <u>subsection 8.1.4</u> and Borrowers shall have, in the Reasonable Credit Judgment of Agent, sufficient Availability pursuant thereto for the making of the proposed Loan or the issuance of the proposed Letter of Credit.

SECTION 10. EVENTS OF DEFAULT; RIGHTS AND REMEDIES ON DEFAULT

- 10.1 Events of Default. The occurrence of one or more of the following events shall constitute an "Event of Default":
- 10.1.1 Nonpayment of Obligations. Borrowers shall fail to pay (i) any principal of any Loan or any LC Obligations on the due date thereof (whether due at stated maturity, on demand, upon acceleration or otherwise), or (ii) any interest on the Obligations or any other Obligations (other than principal of Loans or LC Obligations) within three Business Days of when the same shall become due and payable (whether due at stated maturity, on demand, upon acceleration or otherwise).
- 10.1.2 <u>Misrepresentations</u>. Any representation or warranty made or furnished to Agent or any Lender by or on behalf of any Borrower or any Guarantor in any of the Loan Documents proves to have been false or misleading in any material respect when made, deemed made, furnished or reaffirmed pursuant to <u>Section 7.2</u> hereof, and, if the circumstances giving rise to such false or misleading representation or warranty are susceptible of being cured in all material respects, such false or misleading representation or warranty shall not be cured in all material respects for five days after the earlier to occur of (i) the date on which a Responsible Officer of Borrower Representative shall obtain knowledge thereof, or (ii) the date on which written notice thereof shall have been given to Borrowers by Agent.
- 10.1.3 <u>Breach of Specific Covenants</u>. Any Borrower shall fail or neglect to perform, keep or observe any covenant contained in <u>Section or subsection 5.3</u>, 6.1.1 (first sentence), 6.1.2 (first two sentences), 6.2.3, 6.2.4, 8.1.1, 8.1.2, 8.1.7, 8.1.9, or 8.2 hereof on the date that Borrowers are required to perform, keep or observe such covenant or shall fail or neglect to perform, keep or observe any covenant contained in <u>subsection 8.1.3</u>, 8.1.4, or 8.1.5 hereof within five days following the date on which Borrowers are required to perform, keep or observe such covenant.
- 10.1.4 <u>Breach of Other Covenants</u>. Borrowers shall fail or neglect to perform, keep or observe any covenant contained in this Agreement (other than a covenant which is addressed specifically elsewhere in this <u>Section 10.1</u>) and the breach of such other covenant is not cured within 30 days after the earlier to occur of any Borrower's receipt of notice of such breach from Agent or the date on which such failure or neglect first becomes known to any Responsible Officer of Borrower Representative.
- 10.1.5 <u>Default Under Other Loan Documents</u>. Any default or event of default shall occur under, or any Loan Party shall default in the performance or

observance of any term, covenant, condition or agreement (other than payment of any of the Obligations) contained in, any of the Loan Documents (other than this Agreement) and such default or event of default shall continue beyond any applicable grace period, or, if no grace period is specified, within 30 days after the sooner to occur of any Borrower's receipt of notice of such breach from Agent or the date on which such failure or neglect first becomes known to any Responsible Officer of Borrower Representative.

- 10.1.6 Other Defaults. There shall occur any default or event of default on the part of any Loan Party under any agreement, document or instrument to which such Loan Party is a party or by which such Loan Party or any of its Property is bound, evidencing or relating to any Indebtedness (other than the Obligations) with an outstanding principal balance in excess of \$10,000,000, if (i) the default or event of default results from the failure to pay such Indebtedness at maturity thereof or (ii) the payment or maturity of such Indebtedness is or could be accelerated as a result of such default or event of default.
- 10.1.7 <u>Insolvency and Related Proceedings</u>. Any Loan Party shall suffer the appointment of a receiver, trustee, custodian or similar fiduciary, or shall make an assignment for the benefit of creditors, or any petition for an order for relief shall be filed by or against any Loan Party under U.S. federal bankruptcy laws or under any other bankruptcy or insolvency act or law, state or federal, now or hereafter existing (and any such petition for relief filed against a Loan Party shall not be dismissed within 60 days after the filing or commencement thereof), or any Loan Party shall make (or shall call or attend a meeting for the purpose of making) any offer of settlement, extension or composition to their respective unsecured creditors generally or shall take any corporate action in furtherance thereof.
- 10.1.8 <u>Business Disruption</u>; <u>Condemnation</u>. (a) There shall occur a cessation of a substantial part of the business of any Loan Party, and such cessation would reasonably be expected to have a Material Adverse Effect, (b) any Loan Party shall be enjoined, restrained or in any way prevented by court, governmental or administrative order from conducting all or any substantial part of its business affairs, and such action would reasonably be expected to have a Material Adverse Effect, (c) any substantial portion of the Collateral shall be taken through condemnation, and such taking would reasonably be expected to have a Material Adverse Effect, or (d) the value of such Collateral shall be impaired through condemnation, and such impairment would reasonably be expected to have a Material Adverse Effect.
- 10.1.9 <u>Change of Ownership</u>. (a) any person (other than Glencore) or group of persons (within the meaning of the Exchange Act) shall own or control, beneficially or of record in excess of 50% of the issued and outstanding Securities and Voting Stock of Century or (b) Century shall cease to own and control, beneficially and of record (directly or indirectly), 100% of the issued and outstanding Securities and Voting Stock of each other Borrower and each Guarantor.

- 10.1.10 <u>ERISA</u>. An ERISA Event shall occur which when taken together with all other ERISA Events that have occurred and are continuing, would reasonably be expected to have a Material Adverse Effect.
- 10.1.11 <u>Challenge to Agreement</u>. Any Loan Party shall challenge or contest in any action, suit or proceeding the validity or enforceability of any of the Loan Documents, the legality or enforceability of any of the Obligations or the perfection or priority of any Lien granted to Agent pursuant to the Loan Documents; <u>provided</u> that no Default or Event of Default shall occur under this <u>subsection 10.1.11</u> solely as a result of any good faith dispute by a Loan Party as to the amount of any sum purported to be due under any Loan Document.
- 10.1.12 Repudiation of or Default Under Guaranty Agreement . Any Guarantor shall revoke or attempt to revoke the Guaranty Agreement executed by such Guarantor, or shall repudiate such Guarantor's liability thereunder or shall be in default under the terms thereof in any material respect.
- 10.1.13 <u>Judgments</u>. Any money judgment, writ of attachment or similar processes (collectively, "<u>Judgments</u>") are issued or rendered against any Loan Party or any of the Collateral (i) in the case of money judgments, in an amount of \$5,000,000 or more for all such judgments, attachments or processes in the aggregate, in each case in excess of (A) any applicable insurance with respect to which the insurer has admitted liability and (B) any reserves maintained for such purpose, and (ii) in the case of non-monetary Judgments, such Judgment or Judgments (in the aggregate) would reasonably be expected to have a Material Adverse Effect, in each of the cases described in clauses (i) and (ii) above, which Judgment is not paid, stayed, released, discharged or bonded pending appeal within 40 days.
- 10.2 <u>Acceleration of the Obligations</u>. Upon or at any time after the occurrence and during the continuance of an Event of Default, (i) the Revolving Loan Commitments shall, at the option of Agent or Majority Lenders, be terminated and/or (ii) Agent or Majority Lenders may declare all or any portion of the Obligations at once due and payable without presentment, demand protest or further notice by Agent or any Lender, and Borrowers shall forthwith pay to Agent the full amount of such Obligations; <u>provided</u>, that upon the occurrence of an Event of Default specified in <u>subsection 10.1.7</u> hereof, the Revolving Loan Commitments shall automatically be terminated and all of the Obligations shall become automatically due and payable, in each case without declaration, notice or demand by Agent or any Lender.
- 10.3 Other Remedies. Upon the occurrence and during the continuance of an Event of Default, Agent shall have and may exercise from time to time the following other rights and remedies:
- 10.3.1 All of the rights and remedies of a secured party under the UCC or under other applicable law, and all other legal and equitable rights to which Agent or Lenders may be entitled, all of which rights and remedies shall be cumulative

and shall be in addition to any other rights or remedies contained in this Agreement or any of the other Loan Documents, and none of which shall be exclusive.

- 10.3.2 The right to take immediate possession of the Collateral, and to (i) require each Borrower to assemble the Collateral, at Borrowers' expense, and make it available to Agent at a place designated by Agent which is reasonably convenient to both parties, and (ii) enter any premises where any of the Collateral shall be located and to keep and store the Collateral on said premises until sold (and if said premises be the Property of any Borrower or any Subsidiary of any Borrower, Borrowers agree not to charge, or permit any of its Subsidiaries to charge, Agent for storage thereof).
- 10.3.3 The right to sell or otherwise dispose of all or any Collateral in its then condition, or after any further manufacturing or processing thereof, at public or private sale or sales, with such notice as may be required by law, in lots or in bulk, for cash or on credit, all as Agent, in its sole discretion, may deem advisable. Agent may, at Agent's option, disclaim any and all warranties regarding the Collateral in connection with any such sale. Borrowers agree that 10 days' written notice to Borrowers or any of their Subsidiaries of any public or private sale or other disposition of Collateral shall be reasonable notice thereof, and such sale shall be at such locations as Agent may designate in said notice. Agent shall have the right to conduct such sales on any Borrower's or any of its Subsidiaries' premises, without charge therefor, and such sales may be adjourned from time to time in accordance with applicable law. Agent shall have the right to sell, lease or otherwise dispose of the Collateral, or any part thereof, for cash, credit or any combination thereof, and Agent, on behalf of Secured Parties, may purchase all or any part of the Collateral at public or, if permitted by law, private sale and, in lieu of actual payment of such purchase price, may set off the amount of such price against the Obligations. The proceeds realized from the sale of any Collateral may be applied, after allowing two Business Days for collection, first, to the costs, expenses and attorneys' fees incurred by Agent in collecting the Obligations, in enforcing the rights of Agent and Lenders under the Loan Documents and in collecting, retaking, completing, protecting, removing, storing, advertising for sale, selling and delivering any Collateral; second, to the interest due upon any of the Obligations; and third, to the principal of the Obligations. If any deficiency shall arise, each Borrower shall remain jointly and severally liable to Agent and Lenders therefor.
- 10.3.4 Agent is hereby granted a non-exclusive license to use, after the occurrence and during the continuance of an Event of Default, without charge, each Borrower's labels, patents, copyrights, licenses, rights of use of any name, trade secrets, trade names, trademarks and advertising matter, or any property of a similar nature, as it pertains to the Collateral, for and only to the extent necessary to complete (in respect of raw materials and work-in-process inventory), advertise for sale and sell any Collateral as permitted under the Loan Documents.
- 10.3.5 Agent may, at its option, require Borrowers to deposit cash collateral or a Supporting Letter of Credit in accordance with subsection 1.2.10 hereof and, if Borrowers fail to promptly make such deposit or post such Supporting Letter of Credit, Agent may advance such amount as a Revolving Credit Loan (whether or not an

Overadvance is created thereby). Each such Revolving Credit Loan shall be secured by all of the Collateral and shall constitute a Base Rate Portion. Any such deposit or advance shall be held by Agent as a reserve to fund future payments on future drawings against Letters of Credit. At such time as all Letters of Credit have been drawn upon or expired, any amounts remaining in such reserve shall be applied against any outstanding Obligations, or, if all Obligations have been indefeasibly paid in full, returned to Borrowers.

10.4 Setoff and Sharing of Payments. In addition to any rights now or hereafter granted under applicable law and not by way of limitation of any such rights, after the occurrence and during the continuance of any Event of Default, each Lender and each wholly-owned Subsidiary of any Lender is hereby authorized by Borrowers at any time or from time to time, with prior written consent of Agent and with reasonably prompt subsequent notice to Borrowers (any prior or contemporaneous notice to Borrowers being hereby expressly waived) to set off and to appropriate and to apply any and all (i) balances held by such Lender or wholly-owned Subsidiary at any of its offices for the account of any Borrower or any of its Subsidiaries (regardless of whether such balances are then due to a Borrower or its Subsidiaries), and (ii) other Property at any time held or owing by such Lender or wholly-owned Subsidiary to or for the credit or for the account of any Borrower or any of its Subsidiaries, against and on account of any of the Obligations. Any Lender exercising a right to set off (or whose wholly-owned Subsidiary has exercised a right of set off) shall, to the extent the amount of any such set off exceeds its Revolving Loan Percentage of the amount set off, purchase for cash (and the other Lenders shall sell) interests in each such other Lender's pro rata share of the Obligations as would be necessary to cause such Lender to share such excess with each other Lender may exercise its right to set off with respect to amounts in excess of its pro rata share of the Obligations and upon doing so shall deliver such excess to Agent for the benefit of all Lenders in accordance with the Revolving Loan Percentages.

10.5 Remedies Cumulative; No Waiver. All covenants, conditions, provisions, warranties, guaranties, indemnities, and other undertakings of Borrowers contained in this Agreement and the other Loan Documents, or in any document referred to herein or contained in any agreement supplementary hereto or in any schedule or in any Guaranty Agreement given to Agent or any Lender or contained in any other agreement between any Lender and Borrowers or between Agent and Borrowers heretofore, concurrently, or hereafter entered into, shall be deemed cumulative to and not in derogation or substitution of any of the terms, covenants, conditions, or agreements of Borrowers herein contained. The failure or delay of Agent or any Lender to require strict performance by Borrowers of any provision of this Agreement or to exercise or enforce any rights, Liens, powers, or remedies hereunder or under any of the aforesaid agreements or other documents or security or Collateral shall not operate as a waiver of such performance, Liens, rights, powers and remedies, but all such requirements, Liens, rights, powers, and remedies shall continue in full force and effect until all Loans and other Obligations owing or to become owing from Borrowers to Agent and each Lender have been fully satisfied. None of the undertakings, agreements, warranties, covenants and representations of Borrowers

contained in this Agreement or any of the other Loan Documents and no Default or Event of Default by Borrowers under this Agreement or any other Loan Documents shall be deemed to have been suspended or waived by Lenders, unless such suspension or waiver is by an instrument in writing specifying such suspension or waiver and signed by a duly authorized representative of Agent and directed to Borrowers.

SECTION 11. AGENT

11.1 Authorization and Action. Each Lender hereby appoints Bank as "Agent" under this Agreement and the other Loan Documents and each Lender hereby irrevocably authorizes Agent to take such action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers under this Agreement and the other Loan Documents as are expressly delegated to Agent by the terms hereof and thereof, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere in this Agreement or in any other Loan Document, Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall Agent have or be deemed to have any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" in this Agreement with reference to Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties. In performing its functions and duties under this Agreement, Agent shall act solely on behalf of the Secured Parties and shall not assume, or be deemed to have assumed, any obligation toward, or relationship of agency or trust with or for, any Borrower. The provisions of this Section 11 are solely for the benefit of Agent and Lenders, and Borrowers shall have no rights as third party beneficiaries of any provisions of this Section 11. As to any matters not expressly provided for by this Agreement and the other Loan Documents (including enforcement and collection of the Notes), Agent may, but shall not be required to, exercise any discretion or take any action, but shall be required to act or to refrain from acting (and shall be fully protected in so acting or refraining from acting) upon the instructions of the Majority Lenders, whenever such instruction shall be requested by Agent or required hereunder, or a greater or lesser number of Lenders if so required hereunder, and such instructions shall be binding upon all Lenders; provided that Agent shall be fully justified in failing or refusing to take any action which exposes Agent to any liability or which is contrary to this Agreement, the other Loan Documents or applicable law, unless Agent is indemnified to its satisfaction by the other Lenders against any and all liability and expense which it may incur by reason of taking or continuing to take any such action. If Agent seeks the consent or approval of the Majority Lenders (or a greater or lesser number of Lenders as required in this Agreement), with respect to any action hereunder, Agent shall send notice thereof to each Lender and shall notify each Lender at any time that the Majority Lenders (or such greater or lesser number of Lenders) have instructed Agent to act or refrain from acting pursuant hereto.

11.2 Agent's Reliance, Etc. Neither Agent, any Affiliate of Agent, nor any of their respective directors, officers, agents or employees shall be liable for any action taken or omitted to be taken by it or them under or in connection with this Agreement or the other Loan Documents, except for its or their own gross negligence or willful misconduct. Without limitation of the generality of the foregoing, Agent: (i) may treat each Lender party hereto as the holder of Obligations until Agent receives written notice of the assignment or transfer or such lender's portion of the Obligations signed by such Lender and in form reasonably satisfactory to Agent; (ii) may consult with legal counsel, independent public accountants and other experts selected by it and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such counsel, accountants or experts; (iii) makes no warranties or representations to any Lender and shall not be responsible to any Lender for any recitals, statements, warranties or representations made in or in connection with this Agreement or any other Loan Documents; (iv) shall not have any duty beyond Agent's customary practices in respect of loans in which Agent is the only lender, to ascertain or to inquire as to the performance or observance of any of the terms, covenants or conditions of this Agreement or the other Loan Documents on the part of Borrowers, to inspect the Property (including the books and records) of Borrowers, to monitor the financial condition of Borrowers or to ascertain the existence or possible existence or continuance of any Default or Event of Default; (v) shall not be responsible to any Lender for the due execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement or the other Loan Documents or any other instrument or document furnished pursuant hereto or thereto; (vi) shall not be liable to any Lender for any action taken, or inaction, by Agent upon the instructions of Majority Lenders pursuant to Section 11.1 hereof or refraining to take any action pending such instructions; (vii) shall not be liable for any apportionment or distributions of payments made by it in good faith pursuant to Section 3 hereof; (viii) shall incur no liability under or in respect of this Agreement or the other Loan Documents by acting upon any notice, consent, certificate, message or other instrument or writing (which may be by telephone, facsimile, telegram, cable or electronic mail) believed in good faith by it to be genuine and signed or sent by the proper party or parties; and (ix) may assume that no Event of Default has occurred and is continuing, unless Agent has actual knowledge of the Event of Default, has received notice from Borrowers' independent certified public accountants stating the nature of the Event of Default, or has received notice from a Lender stating the nature of the Event of Default and that such Lender considers the Event of Default to have occurred and to be continuing. If any apportionment or distribution described in clause (vii) above is determined to have been made in error, the sole recourse of any Person to whom payment was due but not made shall be to recover from the recipients of such payments any payment in excess of the amount to which they are determined to have been entitled.

11.3 <u>Bank and its Affiliates</u>. With respect to its commitment hereunder to make Loans, Bank shall have the same rights and powers under this Agreement and the other Loan Documents as any other Lender and may exercise the same as though it were not Agent; and the terms "<u>Lender</u>," "<u>Lenders</u>" or "<u>Majority Lenders</u>" shall, unless otherwise expressly indicated, include Bank in its individual capacity as a Lender. Bank and its Affiliates may lend money to, and generally engage in any kind of business with,

Borrowers, and any Person who may do business with or own Securities of any Borrower, all as if Bank were not Agent and without any duty to account therefor to any other Lender.

- 11.4 <u>Lender Credit Decision</u>. Each Lender acknowledges that it has, independently and without reliance upon Agent or any other Lender and based on the financial statements referred to herein and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement. Agent shall not have any duty or responsibility, either initially or on an ongoing basis, to provide any Lender with any credit or other similar information regarding Borrowers.
- 11.5 <u>Indemnification</u>. Lenders agree to indemnify Agent (to the extent not reimbursed by Borrowers), in accordance with their respective Aggregate Percentages, from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed on, incurred by, or asserted against Agent in any way relating to or arising out of this Agreement or any other Loan Document or any action taken or omitted by Agent under this Agreement; provided that no Lender shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from Agent's gross negligence or willful misconduct. Without limitation of the foregoing, each Lender agrees to reimburse Agent promptly upon demand for its ratable share, as set forth above, of any out-of-pocket expenses (including attorneys' fees) incurred by Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiation, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement and each other Loan Document, to the extent that Agent is not reimbursed for such expenses by Borrowers. The obligations of Lenders under this <u>Section 11.5</u> shall survive the payment in full of all Obligations and the termination of this Agreement. If after payment and distribution of any amount by Agent to Lenders, any Lender or any other Person, including Borrowers, any creditor of any Borrower, a liquidator, administrator or trustee in bankruptcy, recovers from Agent any amount found to have been wrongfully paid to Agent or disbursed by Agent to Lenders, in accordance with their respective Aggregate Percentages, shall reimburse Agent for all such amounts.
- 11.6 <u>Rights and Remedies to Be Exercised by Agent Only</u>. Each Lender agrees that, except as set forth in <u>Section 10.4</u>, no Lender shall have any right individually (i) to realize upon the security created by this Agreement or any other Loan Document, (ii) to enforce any provision of this Agreement or any other Loan Document, or (iii) to make demand under this Agreement or any other Loan Document.
- 11.7 <u>Agency Provisions Relating to Collateral</u>. Each Lender authorizes and ratifies Agent's entry into this Agreement and the Security Documents for the benefit of

Lenders. Each Lender agrees that any action taken by Agent with respect to the Collateral in accordance with the provisions of this Agreement or the Security Documents, and the exercise by Agent of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all Lenders. Agent is hereby authorized on behalf of all Secured Parties, without the necessity of any notice to or further consent from any Lender to take any action with respect to any Collateral or the Loan Documents which may be necessary to perfect and maintain perfected Agent's Liens upon the Collateral, for the benefit of the other Secured Parties. Lenders hereby irrevocably authorize Agent, at its option and in its discretion, to release any Lien granted to or held by Agent upon any Collateral (i) upon termination of the Agreement and payment and satisfaction of all Obligations; or (ii) constituting Property in which no Borrower owned any interest at the time the Lien was granted or at any time thereafter; or (iii) in connection with any foreclosure sale or other disposition of Collateral after the occurrence and during the continuance of an Event of Default; or (iv) if approved, authorized or ratified in writing by Agent at the direction of all Lenders. Upon request by Agent at any time, Lenders will confirm in writing Agent's authority to release particular types or items of Collateral pursuant hereto. Agent shall have no obligation whatsoever to any Lender or to any other Person to assure that the Collateral exists or is owned by any Borrower or is cared for, protected or insured or has been encumbered or that the Liens granted to Agent herein or pursuant to the Security Documents have been properly or sufficiently or lawfully created, perfected, protected or enforced or are entitled to any particular priority, or to exercise at all or in any particular manner or under any duty of care, disclosure or fidelity, or to continue exercising, any of its rights, authorities and powers granted or available to Agent in this Section 11.7 or in any of the Loan Documents, it being understood and agreed that in respect of the Collateral, or any act, omission or event related thereto, Agent may act in any manner it may deem appropriate, in its sole discretion, but consistent with the provisions of this Agreement, including given Agent's own interest in the Collateral as a Lender and that Agent shall have no duty or liability whatsoever to any Lender.

11.8 <u>Agent's Right to Purchase Commitments</u>. Agent shall have the right, but shall not be obligated, at any time upon written notice to any Lender and with the consent of such Lender, which may be granted or withheld in such Lender's sole discretion, to purchase for Agent's own account all of such Lender's interests in this Agreement, the other Loan Documents and the Obligations, for the face amount of the outstanding Obligations owed to such Lender, including all accrued and unpaid interest and fees.

11.9 Resignation of Agent; Appointment of Successor. Agent may resign as Agent by giving not less than 60 days' prior written notice to Lenders and Borrowers. If Agent shall resign under this Agreement, then, (i) subject to the consent of Borrowers (which consent shall not be unreasonably withheld and which consent shall not be required during any period in which a Default or an Event of Default exists), Majority Lenders shall appoint from among the Lenders a successor agent for Lenders or (ii) if a successor agent shall not be so appointed and approved within the 60 day period following Agent's notice to Lenders and Borrowers of its resignation, then Agent shall appoint a successor agent (subject to the consent of Borrowers as set forth in clause (i)

above) who shall serve as Agent until such time as Majority Lenders appoint a successor agent. Upon its appointment, such successor agent shall succeed to the rights, powers and duties of Agent and the term "Agent" shall mean such successor effective upon its appointment, and the former Agent's rights, powers and duties as Agent shall be terminated without any other or further act or deed on the part of such former Agent or any of the parties to this Agreement. After the resignation of any Agent hereunder, the provisions of this Section 11 shall inure to the benefit of such former Agent and such former Agent shall not by reason of such resignation be deemed to be released from liability for any actions taken or not taken by it while it was an Agent under this Agreement.

- 11.10 Audit and Examination Reports; Disclaimer by Lenders. By signing this Agreement, each Lender:
- (a) is deemed to have requested that Agent furnish such Lender, promptly after it becomes available, a copy of each audit or examination report (each a "Report" and collectively, "Reports") prepared by or on behalf of Agent;
- (b) expressly agrees and acknowledges that Agent (i) does not make any representation or warranty as to the accuracy of any Report, and (ii) shall not be liable for any information contained in any Report;
- (c) expressly agrees and acknowledges that the Reports are not comprehensive audits or examinations, that Agent or other party performing any audit or examination will inspect only specific information regarding Borrowers and will rely significantly upon Borrowers' books and records, as well as on representations of Borrowers' personnel;
- (d) agrees to keep all Reports confidential and strictly for its internal use, and not to distribute except to its participants, or use any Report in any other manner, in accordance with the provisions of <u>Section 12.16</u>; and
- (e) without limiting the generality of any other indemnification provision contained in this Agreement, agrees: (i) to hold Agent and any such other Lender preparing a Report harmless from any action the indemnifying Lender may take or conclusion the indemnifying Lender may reach or draw from any Report in connection with any loans or other credit accommodations that the indemnifying Lender has made or may make to Borrowers, or the indemnifying Lender's participation in, or the indemnifying Lender's purchase of, a loan or loans of Borrowers; and (ii) to pay and protect, and indemnify, defend and hold Agent and any such other Lender preparing a Report harmless from and against, the claims, actions, proceedings, damages, costs, expenses and other amounts (including attorneys' fees and expenses) incurred by Agent and any such other Lender preparing a Report as the direct or indirect result of any third parties who might obtain all or part of any Report through the indemnifying Lender.

11.11 <u>USA Patriot Act</u>. Each Lender or assignee or participant of a Lender that is not organized under the laws of the United States of America or a state thereof (and is not excepted from the certification requirement contained in Section 313 of the USA Patriot Act and the applicable regulations because it is both (i) an affiliate of a depository institution or foreign bank that maintains a physical presence in the United States or foreign country, and (ii) subject to supervision by a banking authority regulating such affiliated depository institution or foreign bank) shall deliver to Agent the certification, or, if applicable, recertification, certifying that such Lender is not a "shell" and certifying to other matters as required by Section 313 of the USA Patriot Act and the applicable regulations: (1) within 10 days after the Closing Date and (2) at such other times as are required under the USA Patriot Act.

SECTION 12. MISCELLANEOUS

- 12.1 <u>Right of Sale, Assignment, Participations</u>. Borrowers hereby consent to any Lender's participation, sale, assignment, transfer or other disposition, at any time or times hereafter, of this Agreement and any of the other Loan Documents, or of any portion hereof or thereof, including such Lender's rights, title, interests, remedies, powers and duties hereunder or thereunder subject to the terms and conditions set forth below:
- 12.1.1 Sales, Assignments. Each Lender hereby agrees that, with respect to any sale or assignment (i) no such sale or assignment shall be for an amount of less than \$5,000,000 (unless it is an assignment of all of a Lender's interest), (ii) each such sale or assignment shall be made on terms and conditions which are customary in the industry at the time of the transaction, (iii) Agent, Letter of Credit Issuer and, in the absence of a Default or Event of Default, Borrowers, must consent, such consent not to be unreasonably withheld, to each such assignment to a Person that is not a Lender or an Affiliate of a Lender having substantially similar credit quality as such Lender, (iv) the assigning Lender shall pay to Agent a processing and recordation fee of \$3,500 and any out-of-pocket attorneys' fees and expenses incurred by Agent in connection with any such sale or assignment and (v) Agent, the assigning Lender and the assignee Lender shall each have executed and delivered an Assignment and Acceptance Agreement. After such sale or assignment has been consummated (x) the assignee Lender thereupon shall become a "Lender" for all purposes of this Agreement and (y) the assigning Lender shall have no further liability for funding the portion of Revolving Loan Commitments assumed by such other Lender.
- 12.1.2 <u>Participations</u>. Any Lender may grant participations in its extensions of credit hereunder to any other Lender or other lending institution (a "<u>Participant</u>"); <u>provided</u> that (i) no such participation shall be for an amount of less than \$5,000,000, (ii) no Participant shall thereby acquire any direct rights under this Agreement, (iii) no Participant shall be granted any right to consent to any amendment, except to the extent any of the same pertain to (1) reducing the aggregate principal amount of, or interest rate on, or fees applicable to, any Loan in which such Participant participates or (2) extending the final stated maturity of any Loan or the stated maturity of any portion of any payment of principal of, or interest or fees applicable to, any of the Loans in which such Participant participates; <u>provided</u> that the rights described in this

subclause (2) shall not be deemed to include the right to consent to any amendment with respect to or which has the effect of requiring any mandatory prepayment of any portion of any Loan or any amendment or waiver of any Default or Event of Default, (iv) no sale of a participation in extensions of credit shall in any manner relieve the originating Lender of its obligations hereunder, (v) the originating Lender shall remain solely responsible for the performance of such obligations, (vi) Borrowers and Agent shall continue to deal solely and directly with the originating Lender in connection with the originating Lender's rights and obligations under this Agreement and the other Loan Documents, (vii) in no event shall any financial institution purchasing the participation grant a participation in its participation interest in the Loans without the prior written consent of Agent, and, in the absence of a Default or an Event of Default, Borrowers, which consents shall not unreasonably be withheld and (viii) all amounts payable by Borrowers hereunder shall be determined as if the originating Lender had not sold any such participation.

12.1.3 Foreign Lenders and Transferees.

- (a) Each Foreign Lender or other Lender that is a foreign person for the purposes of the Code shall (i) furnish to Borrower Representative (with a copy to Agent) either United States Internal Revenue Service Form W 8BEN or United States Internal Revenue Service Form W 8ECI (wherein such Lender claims entitlement to complete exemption from United States federal withholding tax on all interest payments hereunder), and (ii) provide to Borrower Representative (with a copy to Agent) a new Form W 8BEN or Form W 8ECI upon the obsolescence of any previously delivered form and comparable statements in accordance with applicable United States laws and regulations and amendments duly executed and completed by such Lender, and comply from time to time with all applicable United States laws and regulations with regard to such withholding tax exemption.
- (b) If, pursuant to this Section 12.1, any interest in this Agreement or any Loans is transferred to any transferee which is organized under the laws of any jurisdiction other than the United States or any state thereof or is otherwise a foreign person for the purposes of the Code, the transferor Lender shall cause such transferee (other than any Participant), and may cause any Participant, concurrently with and as a condition precedent to the effectiveness of such transfer, to (i) represent to the transferor Lender (for the benefit of the transferor Lender, Agent, and Borrowers) that under applicable law and treaties no taxes will be required to be withheld by Agent, any Borrowers or the transferor Lender with respect to any payments to be made to such transferee in respect of the interest so transferred, (ii) furnish to the transferor Lender, Agent and Borrower Representative either United States Internal Revenue Service Form W 8BEN or United States Internal Revenue Service Form W 8ECI (wherein such transferee claims entitlement to complete exemption from United States federal withholding tax on all interest payments hereunder), and (iii) agree (for the benefit of the transferor Lender, Agent and Borrowers) to provide the transferor Lender, Agent and Borrower Representative a new Form W 8BEN or Form W 8ECI upon the

obsolescence of any previously delivered form and comparable statements in accordance with applicable United States laws and regulations and amendments duly executed and completed by such transferee, and to comply from time to time with all applicable United States laws and regulations with regard to such withholding tax exemption.

12.1.4 <u>Assignment to Federal Reserve Bank</u>. In addition to the other assignment rights provided in this Section 12.1, each Lender may assign, as collateral or otherwise, and without notice to or consent of the Agent or any Borrower, any of its rights under this Agreement, whether now owned or hereafter acquired (including rights to payments of principal or interest on the Loans), to any Federal Reserve Bank pursuant to Regulation A of the Federal Reserve Board.

12.2 Amendments, etc.

12.2.1 Amendments, etc. No amendment or waiver of any provision of this Agreement or any other Loan Document (including any Note), nor consent to any departure by Borrowers therefrom, shall in any event be effective unless the same shall be in writing and signed by the Majority Lenders and Borrowers, and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided that no amendment, waiver or consent shall be effective, unless (i) in writing and signed by each Lender, to do any of the following: (1) increase the Revolving Credit Maximum Amount or any Lender's Revolving Loan Commitment, (2) reduce the principal of, or interest on, any amount payable hereunder or under any Note, other than those payable only to Agent or Letter of Credit Issuer in its capacity as such, which may be reduced by Agent unilaterally, (3) decrease any interest rate payable hereunder, (4) postpone any date fixed for any payment of principal of, or interest on, any amounts payable hereunder or under any Note, other than those payable only to Agent in its capacity as such, which may be postponed by Agent unilaterally, (5) increase any advance percentage contained in the definition of the term "Borrowing Base", (6) reduce the number of Lenders that shall be required for Lenders or any of them to take any action hereunder, (7) release or discharge any Person liable for the performance of any obligations of any Borrower hereunder or under any of the Loan Documents, (8) amend any provision of this Agreement that requires the consent of all Lenders or consent to or waive any breach thereof. (9) amend the definition of the term "Majority Lenders", (10) amend this Section 12.2 or (11) release any substantial portion of the Collateral, unless otherwise permitted pursuant to Section 11.7 hereof; or (ii) in writing and signed by Letter of Credit Issuer in addition to the Lenders required above to affect the rights or duties of Letter of Credit Issuer under this Agreement or any other Loan Document or (iii) in writing and signed by Agent in addition to the Lenders required above to affect the rights or duties of Agent under this Agreement, any Note or any other Loan Document. If a fee is to be paid by Borrowers in connection with any waiver or amendment hereunder, the agreement evidencing such amendment or waiver may, at the discretion of Agent (but shall not be required to), provide that only Lenders executing such agreement by a specified date may share in such fee (and in such case, such fee shall be divided among the applicable Lenders on a pro rata basis without including the interests of any Lenders who have not timely executed such agreement).

- 12.2.2 Replacement of Lenders. If any Lender does not consent to any amendment, modification, termination or waiver requested by Borrowers and supported by Agent, then Borrowers may, at their sole expense and effort, upon notice to such Lender and Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in this Agreement), all its interests, rights and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Lender that supports such amendment, modification or waiver, if a Lender accepts such assignment); provided, that: (i) the replacement Lender shall be (a) an existing Lender or (b) another financial institution reasonably acceptable to Agent; (ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (to the extent of such outstanding principal and accrued interest and fees) or Borrowers (in the case of all other amounts); (iii) the assignee shall execute an Assignment and Acceptance Agreement pursuant to which it shall become a party hereto as provided in subsection 12.1.1, and (iv) upon compliance with the provisions for assignment provided in subsection 12.1.1 and the payment of amounts referred to in clause (ii), such assignee shall constitute a "Lender" hereunder and the Lender being so replaced shall no longer constitute a "Lender" hereunder.
- 12.3 <u>Power of Attorney</u>. Each Borrower hereby irrevocably designates, makes, constitutes and appoints Agent (and all Persons designated by Agent) as such Borrower's true and lawful attorney (and agent-in-fact), solely with respect to the matters set forth in this <u>Section 12.3</u>, and Agent, or Agent's agent, may, without notice to any Borrower and in any Borrower's or Agent's name, but at the cost and expense of Borrowers:
- 12.3.1 At such time or times upon or after the occurrence and during the continuance of an Event of Default, endorse any Borrower's name on any checks, notes, acceptances, drafts, money orders or any other evidence of payment or proceeds of the Collateral which come into the possession of Agent or under Agent's control.
- 12.3.2 At such time or times upon or after the occurrence and during the continuance of an Event of Default, as Agent or its agent in its sole discretion may determine: (i) demand payment of the Accounts from the Account Debtors, enforce payment of the Accounts by legal proceedings or otherwise, and generally exercise all of any Borrower's rights and remedies with respect to the collection of the Accounts; (ii) settle, adjust, compromise, discharge or release any of the Accounts or other Collateral or any legal proceedings brought to collect any of the Accounts or other Collateral; (iii) sell or assign any of the Accounts and other Collateral upon such terms, for such amounts and at such time or times as Agent deems advisable, and at Agent's option, with all warranties regarding the Collateral disclaimed; (iv) take control, in any manner, of any item of payment or proceeds relating to any Collateral; (v) prepare, file and sign any Borrower's name to a proof of claim in bankruptcy or similar document against any Account Debtor or to any notice of lien, assignment or satisfaction of lien or similar document in connection with any of the Collateral; (vi) receive, open and dispose of all mail addressed to any Borrower and notify postal authorities to change the address for delivery thereof to such address as Agent may designate; (vii) endorse the name of any

Borrower upon any of the items of payment or proceeds relating to any Collateral and deposit the same to the account of Agent on account of the Obligations; (viii) endorse the name of any Borrower upon any chattel paper, document, instrument, invoice, freight bill, bill of lading or similar document or agreement relating to the Accounts, Inventory and any other Collateral; (ix) use any Borrower's stationery and sign the name of any Borrower to verifications of the Accounts and notices thereof to Account Debtors; (x) use the information recorded on or contained in any data processing equipment and Computer Hardware and Software relating to the Accounts, Inventory, Equipment and any other Collateral; (xi) make and adjust claims under policies of insurance; and (xii) do all other acts and things necessary, in Agent's determination, to fulfill any Borrower's obligations under this Agreement.

The power of attorney granted hereby shall constitute a power coupled with an interest and shall be irrevocable.

12.4 Indemnity. Each Borrower hereby agrees to indemnify (a) Agent, (b) Letter of Credit Issuer, (c) each Lender, (d) each of the Affiliates of each of the Persons listed in the foregoing clauses (a) through (c), and (e) each of the directors, members, managers, general partners, limited partners, officers, and employees of each of the Persons listed in the foregoing clauses (a) through (d) (collectively, the "Indemnified Persons ") and hold each of the Indemnified Persons harmless from and against any liability, loss, damage, suit, action or proceeding ever suffered or incurred by such Indemnified Person (including reasonable attorneys' fees and legal expenses) as the result of any Borrower's failure to observe, perform or discharge such Borrower's duties hereunder. In addition, each Borrower shall defend each Indemnified Person against and save it harmless from all claims of any Person with respect to the Collateral (except those resulting from the gross negligence or intentional misconduct of such Indemnified Person). Without limiting the generality of the foregoing, these indemnities shall extend to any claims asserted against any Indemnified Person by any Person under any Environmental Laws by reason of any Borrower's or any other Person's failure to comply with laws applicable to solid or hazardous waste materials or other toxic substances. Notwithstanding the foregoing, (i) the foregoing indemnity shall not be available to any Indemnified Person to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from such Indemnified Person's gross negligence or willful misconduct; (ii) such indemnity shall not be available to any Indemnified Person for losses, claims, damages, liabilities or related expenses arising out of a proceeding in which such Indemnified Person and a Borrower are adverse parties to the extent that such Borrower prevails on the merits, as determined by a court of competent jurisdiction by final and nonappealable judgment (it being understood that nothing in this Agreement shall preclude a claim or suit by a Borrower against any indemnitee for such Indemnified Person's failure to perform any of its obligations to Borrowers under the Loan Documents); (iii) Borrowers shall not, in connection with any such proceeding or related proceedings in the same jurisdiction and in the absence of conflicts of interest, be liable for the fees and expenses of more than one law firm at any one time for the Indemnified Person (which law firm shall be selected (x) by mutual agreement of Agent and Borrower Representative or (y) if no such agreement has been reached following Agent's good

faith consultation with Borrower Representative with respect thereto, by Agent in its sole discretion); (iv) each Indemnified Person shall give Borrower Representative (A) prompt notice of any such action brought against such Indemnified Person in connection with a claim for which it is entitled to indemnity under this Section 12.4 and (B) an opportunity to consult from time to time with such Indemnified Person regarding defensive measures and potential settlement; and (v) Borrowers shall not be obligated to pay the amount of any settlement entered in to without their written consent (which consent shall not be unreasonably withheld or delayed). Notwithstanding any contrary provision in this Agreement, the obligation of Borrowers under this Section 12.4 shall survive the payment in full of the Obligations and the termination of this Agreement.

- 12.5 <u>Sale of Interest</u>. No Borrower may sell, assign or transfer any interest in this Agreement, any of the other Loan Documents, or any of the Obligations, or any portion thereof, including such Borrower's rights, title, interests, remedies, powers and duties hereunder or thereunder.
- 12.6 <u>Severability</u>. Wherever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement shall be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Agreement.
- 12.7 <u>Successors and Assigns</u>. This Agreement and each of the other Loan Documents shall be binding upon and inure to the benefit of the successors and assigns of each Borrower, Agent and each Lender permitted under <u>Section 12.1</u> hereof.
- 12.8 <u>Cumulative Effect; Conflict of Terms</u>. The provisions of the other Loan Documents are hereby made cumulative with the provisions of this Agreement. Except as otherwise provided in any of the other Loan Documents by specific reference to the applicable provision of this Agreement, if any provision contained in this Agreement is in direct conflict with, or inconsistent with, any provision in any of the other Loan Documents, the provision contained in this Agreement shall govern and control.
- 12.9 Execution in Counterparts. This Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed to be an original and all of which counterparts taken together shall constitute but one and the same instrument.
- 12.10 <u>Notice</u>. Except as otherwise provided herein, all notices, requests and demands to or upon a party hereto, to be effective, shall be in writing, and shall be sent by certified or registered mail, return receipt requested, by personal delivery against receipt, by overnight courier or by facsimile and, unless otherwise expressly provided herein, shall be deemed to have been validly served, given, delivered or received immediately when delivered against receipt, three Business Days' after deposit in the mail, postage prepaid, one Business Day after deposit with an overnight courier or, in the

case of facsimile notice, when sent with respect to machine confirmed, addressed as follows:

(A) If to Agent: Bank of America, N.A.

1 South Wacker Drive

Suite 3400

Chicago, Illinois 60606

Attention: Loan Administration Manager

Facsimile No.: 312-332-6537

With a copy to: Skadden, Arps, Slate, Meagher & Flom LLP

333 West Wacker Drive, Suite 2100

Chicago, Illinois 60606

Attention: Seth E. Jacobson, Esq. Facsimile No.: 312-407-0411

(B) If to Borrowers: Century Aluminum Company

2511 Garden Road Building A, Suite 200 Monterey, CA 93940

Attention: Chief Financial Officer Facsimile No.:831-642-9082

With a copy to: Curtis, Mallet-Prevost, Colt & Mosle LLP

101 Park Avenue

New York, New York 10178-0061 Attention:Matias A. Vega and Roman A. Bninski

Facsimile No.: 212-697-1559

(C) If to any Lender, at its address indicated on the signature pages hereof or in an Assignment and Acceptance Agreement,

or to such other address as each party may designate for itself by notice given in accordance with this <u>Section 12.10</u>; <u>provided</u>, <u>however</u>, that any notice, request or demand to or upon Agent or a Lender pursuant to <u>subsection 3.1.1</u> or <u>4.2.2</u> hereof shall not be effective until received by Agent or such Lender.

12.11 <u>Consent</u>. Whenever Agent's, Majority Lenders' or all Lenders' consent is required to be obtained under this Agreement or any of the other Loan Documents as a condition to any action, inaction, condition or event, except as otherwise specifically provided herein, Agent, Majority Lenders or all Lenders, as applicable, shall be authorized to give or withhold such consent in its or their sole and absolute discretion and to condition its or their consent upon the giving of additional Collateral security for the Obligations, the payment of money or any other matter.

- 12.12 <u>Credit Inquiries</u>. Subject to the confidentiality provisions contained in <u>Section 12.16</u>, Borrowers hereby authorize and permit Agent and each Lender to respond to usual and customary credit inquiries from third parties concerning any Borrower or any of its Subsidiaries.
 - 12.13 Time of Essence . Time is of the essence of this Agreement and the other Loan Documents.
- 12.14 Entire Agreement. This Agreement and the other Loan Documents, together with all other instruments, agreements and certificates executed by the parties in connection therewith or with reference thereto, embody the entire understanding and agreement between the parties hereto and thereto with respect to the subject matter hereof and thereof and supersede all prior agreements, understandings and inducements, whether express or implied, oral or written.
- 12.15 <u>Interpretation</u>. No provision of this Agreement or any of the other Loan Documents shall be construed against or interpreted to the disadvantage of any party hereto by any court or other governmental or judicial authority by reason of such party having or being deemed to have structured or dictated such provision.
- 12.16 <u>Confidentiality</u>. Agent and each Lender shall hold all nonpublic information obtained pursuant to the requirements of this Agreement in accordance with Agent's and such Lender's customary procedures for handling confidential information of this nature and in accordance with safe and sound banking practices and in any event may make disclosure reasonably required by a prospective participant or assignee in connection with the contemplated participation or assignment or as required or requested by any governmental authority or representative thereof or pursuant to legal process and shall require any such participant or assignee to agree to comply with this <u>Section 12.16</u>.
- 12.17 GOVERNING LAW; CONSENT TO JURISDICTION . THIS AGREEMENT HAS BEEN NEGOTIATED, EXECUTED AND DELIVERED IN AND SHALL BE DEEMED TO HAVE BEEN MADE IN NEW YORK, NY. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK; PROVIDED, HOWEVER, THAT IF ANY OF THE COLLATERAL SHALL BE LOCATED IN ANY JURISDICTION OTHER THAN NEW YORK, THE LAWS OF SUCH JURISDICTION SHALL GOVERN THE METHOD, MANNER AND PROCEDURE FOR FORECLOSURE OF AGENT'S LIEN UPON SUCH COLLATERAL AND THE ENFORCEMENT OF AGENT'S OTHER REMEDIES IN RESPECT OF SUCH COLLATERAL TO THE EXTENT THAT THE LAWS OF SUCH JURISDICTION ARE DIFFERENT FROM OR INCONSISTENT WITH THE LAWS OF NEW YORK. AS PART OF THE CONSIDERATION FOR NEW VALUE RECEIVED, AND REGARDLESS OF ANY PRESENT OR FUTURE DOMICILE OR PRINCIPAL PLACE OF BUSINESS OF ANY BORROWER, AGENT OR ANY LENDER, EACH BORROWER HEREBY CONSENTS AND AGREES THAT THE COURTS OF THE STATE OF NEW YORK LOCATED IN THE CITY OF NEW YORK, OR, AT AGENT'S OPTION, THE UNITED STATES DISTRICT COURT FOR THE

SOUTHERN DISTRICT OF NEW YORK, SHALL HAVE EXCLUSIVE JURISDICTION TO HEAR AND DETERMINE ANY CLAIMS OR DISPUTES BETWEEN BORROWERS ON THE ONE HAND AND AGENT OR ANY LENDER ON THE OTHER HAND PERTAINING TO THIS AGREEMENT OR TO ANY MATTER ARISING OUT OF OR RELATED TO THIS AGREEMENT. EACH BORROWER EXPRESSLY SUBMITS AND CONSENTS IN ADVANCE TO SUCH JURISDICTION IN ANY ACTION OR SUIT COMMENCED IN ANY SUCH COURT, AND EACH BORROWER HEREBY WAIVES ANY OBJECTION WHICH ANY BORROWER MAY HAVE BASED UPON LACK OF PERSONAL JURISDICTION, IMPROPER VENUE OR FORUM NON CONVENIENS. EACH BORROWER HEREBY WAIVES PERSONAL SERVICE OF THE SUMMONS, COMPLAINT AND OTHER PROCESS ISSUED IN ANY SUCH ACTION OR SUIT AND AGREES THAT SERVICE OF SUCH SUMMONS, COMPLAINT AND OTHER PROCESS MAY BE MADE BY REGISTERED OR CERTIFIED MAIL ADDRESSED TO BORROWERS AT THE ADDRESS SET FORTH IN THIS AGREEMENT AND THAT SERVICE SO MADE SHALL BE DEEMED COMPLETED UPON THE EARLIER OF BORROWERS' ACTUAL RECEIPT THEREOF OR 3 DAYS AFTER DEPOSIT IN THE U.S. MAILS, PROPER POSTAGE PREPAID. NOTHING IN THIS AGREEMENT SHALL BE DEEMED OR OPERATE TO AFFECT THE RIGHT OF AGENT OR ANY LENDER TO SERVE LEGAL PROCESS IN ANY OTHER MANNER PERMITTED BY LAW, OR TO PRECLUDE THE ENFORCEMENT BY AGENT OR ANY LENDER TO SAVE LEGAL PROCESS IN ANY OTHER MANNER PERMITTED BY LAW, OR TO PRECLUDE THE ENFORCEMENT BY AGENT OR ANY LENDER TO SAVE LEGAL PROCESS IN ANY OTHER MANNER PERMITTED BY LAW, OR TO PRECLUDE THE ENFORCEMENT BY AGENT OR ANY LENDER TO SAVE LEGAL PROCESS IN ANY OTHER MANNER PERMITTED BY LAW, OR TO PRECLUDE THE ENFORCEMENT BY AGENT OR ANY LENDER TO SAVE LEGAL PROCESS IN ANY OTHER APPROPRIATE FORUM OR JURISDICTION.

12.18 <u>WAIVERS BY BORROWERS</u>. EACH BORROWER WAIVES (I) THE RIGHT TO TRIAL BY JURY (WHICH AGENT AND EACH LENDER HEREBY ALSO WAIVES) IN ANY ACTION, SUIT, PROCEEDING OR COUNTERCLAIM OF ANY KIND ARISING OUT OF OR RELATED TO THIS AGREEMENT OR ANY OF THE OTHER LOAN DOCUMENTS, THE OBLIGATIONS OR THE COLLATERAL, AND (II) EXCEPT AS PROHIBITED BY LAW, ANY RIGHT TO CLAIM OR RECOVER ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES OTHER THAN, OR IN ADDITION TO, ACTUAL DAMAGES. EACH BORROWER ACKNOWLEDGES THAT THE FOREGOING WAIVERS ARE A MATERIAL INDUCEMENT TO AGENT'S AND EACH LENDER'S ENTERING INTO THIS AGREEMENT AND THAT AGENT AND EACH LENDER IS RELYING UPON THE FOREGOING WAIVERS IN ITS FUTURE DEALINGS WITH BORROWERS. EACH BORROWER WARRANTS AND REPRESENTS THAT IT HAS REVIEWED THE FOREGOING WAIVERS WITH ITS LEGAL COUNSEL AND HAS KNOWINGLY AND VOLUNTARILY WAIVED ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

- 12.19 <u>Advertisement</u>. Borrowers hereby authorize Agent to publish the names of Borrowers and the amount of the credit facility provided hereunder in any "tombstone" or comparable advertisement which Agent elects to publish.
- 12.20 Reimbursement. The undertaking by Borrowers to repay the Obligations and each representation, warranty or covenant of each Borrower are and shall be joint and several. To the extent that any Borrower shall be required to pay a portion of the Obligations which shall exceed the amount of loans, advances or other extensions of credit received by such Borrower and all interest, costs, fees and expenses attributable to such loans, advances or other extensions of credit, then such Borrower shall be reimbursed by the other Borrowers for the amount of such excess. This Section 12.20 is intended only to define the relative rights of Borrowers, and nothing set forth in this Section 12.20 is intended or shall impair the obligations of each Borrower, jointly and severally, to pay to Agent and Lenders the Obligations as and when the same shall become due and payable in accordance with the terms hereof. Notwithstanding anything to the contrary set forth in this Section 12.20 or any other provisions of this Agreement, it is the intent of the parties hereto that the liability incurred by each Borrower in respect of the Obligations of the other Borrowers (and any Lien granted by each Borrower to secure such Obligations) not constitute a fraudulent conveyance or fraudulent transfer under the provisions of any applicable law of any state or other governmental unit ("Fraudulent Conveyance"). Consequently, each Borrower, Agent and each Lender hereby agree that if a court of competent jurisdiction determines that the incurrence of liability by any Borrower in respect of the Obligations of any other Borrower (or any Liens granted by such Borrower to secure such Obligations) would, but for the application of this sentence, constitute a Fraudulent Conveyance, such liability (and such Liens) shall be valid and enforceable only to the maximum extent that would not cause the same to constitute a Fraudulent Conveyance, and this Agreement and the other Loan Documents shall automatically be deemed to have been amended accordingly, nunc
- 12.21 <u>Section Headings</u>. Article and Section headings and the table of contents used herein are for convenience of reference only, are not part of this Agreement, and shall not affect the construction or interpretation hereof.

IN WITNESS WHEREOF, this Agreement has been duly executed on the day and year specified at the beginning of this Agreement.

BORROWERS:

CENTURY ALUMINUM COMPANY

By: /s/ Daniel J. Krofcheck

Name: Daniel J. Krofcheck

Title: Vice President and Treasurer

BERKELEY ALUMINUM, INC.

By: /s/ Daniel J. Krofcheck

Name: Daniel J. Krofcheck

Title: Vice President and Treasurer

CENTURY ALUMINUM OF WEST VIRGINIA, INC.

By: /s/ Daniel J. Krofcheck

Name: Daniel J. Krofcheck

Title: Vice President and Treasurer

CENTURY KENTUCKY, INC.

By: /s/ Daniel J. Krofcheck

Name: Daniel J. Krofcheck

Title: Vice President and Treasurer

NSA, LTD.

By: /s/ Daniel J. Krofcheck

Name: Daniel J. Krofcheck

Title: Vice President and Treasurer

BANK OF AMERICA, N.A.,

as Agent and as a Lender

By: /s/ Robert J. Lund

Name: Robert J. Lund Title: Senior Vice President

Revolving Loan Commitment: \$33,000,000.00

CREDIT SUISSE, CAYMAN ISLANDS BRANCH

By: /s/ Alain Daoust

Name: Alain Daoust Title: Director

By: /s/ Denise L. Alvarez

Name: Denise L. Alvarez

Title: Associate

Revolving Loan Commitment: \$17,500,000.00

68

CITIBANK, N.A.

By: /s/ Daniel J. Miller
Name: Daniel J. Miller Title: Vice President

Revolving Loan Commitment: \$17,500,000.00

JPMORGAN CHASE BANK, N.A.

By: /s/ Mark Cuccinello
Name: Mark Cuccinello Title: Vice President

Revolving Loan Commitment: \$17,500,00.00

WACHOVIA CAPITAL FINANCE CORPORATION (WESTERN)

By: /s/ Gary D. Cassianni
Name: Gary D. Cassianni
Title: Vice President

Revolving Loan Commitment:

\$14,500,000.00

71

APPENDIX A

GENERAL DEFINITIONS

When used in the Loan and Security Agreement dated as of September 19, 2005 (the "Agreement"), by and among Bank of America, N.A., the Lenders, and Century Aluminum Company, Berkeley Aluminum Inc., Century Aluminum of West Virginia, Inc., Century Kentucky, Inc., and NSA Ltd., as Borrowers, (a) the terms Account, Chattel Paper, Deposit Account, Document, Electronic Chattel Paper, Instruments, Inventory, Letter-of-Credit Rights, Payment Intangibles, Proceeds, Software, Supporting Obligations, and Tangible Chattel Paper have the respective meanings assigned thereto under the UCC; (b) all terms reflecting Collateral having the meanings assigned thereto under the UCC shall be deemed to mean such Property, whether now owned or hereafter created or acquired by any Borrower or in which such Borrower now has or hereafter acquires any interest; (c) capitalized terms which are not otherwise defined have the respective meanings assigned thereto in the Agreement; (d) accounting terms not otherwise specifically defined in the Agreement shall be construed in accordance with GAAP consistently applied; and (e) the following terms shall have the following meanings (terms defined in the singular to have the same meaning when used in the plural and vice versa):

Account Debtor — any Person who is or may become obligated under or on account of any Account.

<u>Affiliate</u>— with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under direct or indirect common control with, such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with") with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

Agent — as defined in the preamble to the Agreement.

<u>Agent Loans</u> — as defined in subsection 1.1.4 of the Agreement.

<u>Aggregate Percentage</u> — with respect to each Lender, the percentage equal to the quotient of (i) such Lender's Revolving Loan Commitment <u>divided</u> by (ii) the aggregate of all Revolving Loan Commitments.

<u>Agreement</u> — the Loan and Security Agreement referred to in the first sentence of this Appendix A, all Exhibits and Schedules thereto and this Appendix A, as each of the same may be amended from time to time.

Anti-Terrorism Law — those portions of the USA Patriot Act or any other statute, regulation, executive order, or other law pertaining to the prevention of future acts of terrorism, in each case as such law may be amended from time to time.

Applicable Margin — the percentages set forth in the table below with respect to the Base Rate Portion and each LIBOR Portion that corresponds to the total of the principal amount of Revolving Credit Loans then outstanding (including any amounts which Agent or any Lender may have paid for the account of any Borrower in accordance with any of the Loan Documents and which have not been reimbursed by Borrowers) <u>plus</u> the LC Exposure ("<u>Usage</u>") set forth in the table below. Applicable Margins shall be determined monthly based on the average daily Usage of the Facility during the immediately preceding month.

Usage	Base Rate Margin	LIBOR Margin
Greater than 50% of the Revolving Credit Maximum Amount	0.00%	1.50%
Greater than 25%, but less than or equal to 50%, of the Revolving Credit Maximum Amount	0.00%	1.25%
Less than or equal to 25%, of the Revolving Credit Maximum Amount	-0.25%	1.00%

<u>Assignment and Acceptance Agreement</u> — an assignment and acceptance agreement substantially in the form of Exhibit A to the Agreement.

Availability — at any time, the amount of additional money which Borrowers are entitled to borrow from time to time as Revolving Credit Loans, such amount being the difference derived when the sum of the principal amount of Revolving Credit Loans then outstanding (including any amounts which Agent or any Lender may have paid for the account of any Borrower in accordance with any of the Loan Documents and which have not been reimbursed by Borrowers), the LC Exposure, the Minimum Availability Reserve, and any Reserves is subtracted from the Borrowing Base. If the amount outstanding is equal to or greater than the Borrowing Base, Availability is 0. In no event shall Availability be greater than the Revolving Credit Maximum Amount.

Bailee Certificate — a letter agreement substantially in the form of Exhibit B to the Agreement.

Bank — as defined in the preamble to the Agreement.

Base Rate — for any day, the greater of (i) the rate of interest in effect for such day as publicly announced from time to time by the Bank in Charlotte, North Carolina as its "prime rate" (the "prime rate" being a rate set by the Bank based upon various factors including the Bank's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate) and (ii) the Federal Funds Rate plus .50%. Any change in the prime rate announced by Bank shall take effect at the opening of business on the day specified in the public announcement of such change. Each interest rate based upon the Base Rate shall be adjusted simultaneously with any change in the Base Rate.

<u>Base Rate Portion</u>—that portion of the Revolving Credit Loans that is not subject to a LIBOR Option.

<u>Blocked Account Agreement</u> — an agreement among a Borrower, Agent, and a depository bank substantially in the form of Exhibit C to the Agreement.

Borrower — as defined in the preamble to the Agreement.

Borrower Representative — Century Aluminum Company, acting on its own behalf as a Borrower and on behalf of all other Borrowers.

Borrowing — a borrowing of Loans or an issuance of Letters of Credit under the Agreement.

Borrowing Base — as at any date of determination thereof, an amount equal to the sum of:

- (i) (1) 85% of the net amount of Eligible Accounts (other than with respect to which Glencore is the Account Debtor) outstanding at such date; <u>plus</u> (2) the lesser of (x) 85% of the net amount of Eligible Accounts with respect to which Glencore is the Account Debtor outstanding at such date, and (y) \$20,000,000; and
 - (ii) 65% of the value of Eligible Inventory at such date.

The advance rates set forth above may not be adjusted downward by Agent. For purposes hereof, (1) the net amount of Eligible Accounts at any time shall be the face amount of such Eligible Accounts less any and all returns, rebates, discounts (which may, at Agent's option exercised using its Reasonable Credit Judgment, be calculated on shortest terms), credits, allowances or excise taxes of any nature at any time issued, owing, claimed by Account Debtors, granted, outstanding or payable in connection with such Accounts at such time and (2) the value of Eligible Inventory shall be determined on a first-in, first-out, lower of cost or market basis in accordance with GAAP.

Borrowing Base Certificate — a certificate of a Responsible Officer of Borrower Representative substantially in the form of Exhibit D to the Agreement (or another form acceptable to Agent) setting forth the calculation of the Borrowing Base. All calculations of the Borrowing Base in connection with the preparation of any Borrowing Base Certificate shall originally be made by Borrowers and certified to Agent; provided that Agent shall have the right to review and adjust, in the exercise of its Reasonable Credit Judgment, any such calculation after giving notice thereof to Borrowers, to the extent that Agent determines that such calculation is not in accordance with the Agreement.

<u>Business Day</u> — any day, excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York or is a day on which banking institutions located in such state are closed; provided that with respect to determinations in connection with, and payments of principal and interest on, LIBOR Portions, such day is also a LIBOR Business Day.

<u>Capital Expenditures</u> — expenditures made or liabilities incurred for the acquisition of any fixed or capital assets or improvements, replacements, substitutions or additions thereto which have a useful life of more than one year, including the total principal portion of Capitalized Lease Obligations; provided, however, that Capital Expenditures shall not include (a) expenditures to the extent they are paid with the proceeds of insurance settlements, condemnation awards, and other settlements in respect of lost, destroyed, damaged, or condemned fixed or capital assets, (b) expenditures to the extent they are financed with the proceeds of a sale or other disposition of fixed or capital assets that is expressly permitted under the Agreement, and (c) expenditures made in connection with the construction of any fixed or capital asset if the applicable Borrower intends to consummate a sale and leaseback transaction of such asset permitted under the Agreement within six months of the completion of such construction; <u>provided</u> that if such sale and leaseback transaction is not consummated within such time period, then all such expenditures will constitute Capital Expenditures in the period in which such six-month period ends.

<u>Capitalized Lease Obligation</u>— at the time of any determination thereof, any Indebtedness represented by obligations under a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

<u>Century</u> — as defined in the preamble to the Agreement.

<u>Closing Date</u> — the date on which all of the conditions precedent in Section 9.1 of the Agreement are satisfied or waived.

Code — the Internal Revenue Code of 1986.

<u>Collateral</u> — all of the Property and interests in Property of Borrowers described in Section 5.1 of the Agreement and not excluded pursuant to Section 5.2 of the Agreement, and all other Property and interests in Property that may hereafter be pledged to Agent for the benefit of the Secured Parties to secure the payment and performance of any of the Obligations.

<u>Compliance Certificate</u> — a certificate substantially in the form of Exhibit E to the Agreement executed by a Financial Officer of Borrower Representative.

Computer Hardware and Software — all of any Borrower's rights (including rights as licensee and lessee) with respect to (i) computer and other electronic data processing hardware, including all integrated computer systems, central processing units, memory units, display terminals, printers, computer elements, card readers, tape drives, hard and soft disk drives, cables, electrical supply hardware, generators, power equalizers, accessories, peripheral devices and other related computer hardware; (ii) all software and all software programs designed for use on the computers and electronic data processing hardware described in clause (i) above, including all operating system software, utilities and application programs in any form (source code and object code in magnetic tape, disk or hard copy format or any other listings whatsoever); (iii) any firmware associated with any of the foregoing; and (iv) any documentation for hardware, software and firmware

described in clauses (i), (ii) and (iii) above, including flow charts, logic diagrams, manuals, specifications, training materials, charts and pseudo codes.

Consolidated — the consolidation in accordance with GAAP of the accounts or other items as to which such term applies.

<u>Consolidated Net Income (Loss)</u> — with respect to any fiscal period, the net income (or loss) of the Loan Parties determined in accordance with GAAP on a Consolidated basis; <u>provided</u>, <u>however</u>, Consolidated Net Income shall not include:

- (i) the net income (or loss) of any Person (other than a Loan Party) in which a Loan Party has an ownership interest unless received in a cash distribution or requiring the payment of cash;
- (ii) the net income (or loss) of any Person accrued prior to the date it became a Subsidiary of a Loan Party or is merged into or consolidated with a Loan Party;
 - (iii) net after-tax extraordinary gains or losses as defined under GAAP;
 - (iv) net after-tax gains or losses from asset dispositions other than sales in the ordinary course of business;
 - (v) net after-tax gains or losses attributable to the early extinguishment of debt;
- (vi) any after-tax non-cash charges or gains, determined in accordance with GAAP, related to hedging contracts until such time as such contracts are settled;
 - (vii) the cumulative effect of a change in accounting principles;
- (viii) after-tax charges or gains relating to the valuation of inventory by application of the LIFO (last in/first out) method of inventory valuation; and
- (ix) all other after-tax non-cash charges or gains (excluding any such non-cash charge or gain to the extent that it represents an accrual of or reserve for cash expenditures in any future period).

<u>Current Assets</u> — at any date, the assets of a Person that would be properly classified as current assets on a balance sheet of such Person at such date in accordance with GAAP.

<u>Deed of Trust</u>— the Deed of Trust, Assignment of Leases and Rents, Security Agreement, Financing Statement and Fixture Filing, dated as of April 2, 2001, from Century Aluminum of West Virginia, Inc., to Charles B. Dollison, the Trustee for the

benefit of Wilmington Trust Company, as Collateral Agent, as originally executed by the parties thereto.

<u>Default</u> — an event or condition the occurrence of which would, with the lapse of time or the giving of notice, or both, become an Event of Default.

<u>Default Rate</u> — as defined in subsection 2.1.2 of the Agreement.

<u>Defaulting Lender</u> — as defined in subsection 3.1.3(b) of the Agreement.

<u>Derivative Obligations</u> — every obligation of a Person under any forward contract, futures contract, exchange contract, swap, option or other financing agreement or arrangement (including caps, floors, collars and similar arrangements), the value of which is dependent upon interest rates, currency exchange rates, commodities indices or other indices.

<u>Dilution</u>— with respect to any period for any Borrower, an amount, expressed as a percentage, equal to (a) bad debt write-downs, discounts, credits, or similar non-cash items with respect to Accounts of such Borrower during such period (but only to the extent Agent believes, in its Reasonable Credit Judgment, that such factors are likely to continue in effect) <u>divided by (b)</u> the aggregate of all collections with respect to Accounts of such Borrower during such period (excluding extraordinary items) <u>plus</u> the amount determined pursuant to clause (a).

<u>Distribution</u> — in respect of any Person, includes: (i) the payment of any dividends or other distributions on Securities (except distributions in such Securities) and (ii) the redemption, acquisition, or other retirement of Securities of such Person, as the case may be, unless made contemporaneously from the net proceeds of the sale of Securities.

<u>Dominion Account</u> — a Deposit Account established by a Borrower pursuant to and in accordance with subsection 6.2.3 of the Agreement.

<u>EBITDA</u>— with respect to any period, the sum of Consolidated Net Income (Loss) before Interest Expense, income taxes, depreciation and amortization for such period all as determined for Borrowers and Guarantors on a Consolidated basis and in accordance with GAAP.

<u>Eligible Account</u> — an Account arising in the ordinary course of the business of any Borrower from the sale of goods or rendition of services; provided that no Account shall be an Eligible Account if:

- (i) it arises out of a sale made or services rendered by a Borrower to a Subsidiary of a Borrower or an Affiliate of a Borrower or to a Person controlled by an Affiliate of a Borrower; or
- (ii) it remains unpaid more than 120 days after the original invoice date shown on the invoice or 60 days after the original due date shown on the invoice; or

- (iii) the Account Debtor has disputed liability or made a claim or exercised a right of setoff with respect to such Account; <u>provided</u>, that any such Account shall be eligible to the extent the amount thereof exceeds such dispute, claim, or right of setoff; or
- (iv) (A) the Account Debtor is also a creditor or supplier of a Borrower, or (B) the Account otherwise is subject to right of setoff by the Account Debtor; <u>provided</u>, that (1) any such Account shall be eligible to the extent such amount thereof exceeds such contract, setoff or similar right, and (2) any such Account shall be eligible to the extent such Account is subject to a No-Offset Letter executed by the applicable Account Debtor and delivered to Agent; or
- (v) the Account Debtor has commenced a voluntary case under the federal bankruptcy laws, as now constituted or hereafter amended, or made an assignment for the benefit of creditors, or a decree or order for relief has been entered by a court having jurisdiction in the premises in respect of the Account Debtor in an involuntary case under the federal bankruptcy laws, as now constituted or hereafter amended, or any other petition or other application for relief under the federal bankruptcy laws, as now constituted or hereafter amended, has been filed against the Account Debtor, or if the Account Debtor has failed, suspended business, ceased to be Solvent to the knowledge of the applicable Borrower, or consented to or suffered a receiver, trustee, liquidator or custodian to be appointed for it or for all or a significant portion of its assets or affairs; or
- (vi) it arises from a sale made or services rendered to an Account Debtor outside the United States, unless either (1) such sale is made on letter of credit, guaranty or acceptance terms, in each case acceptable to Agent in its Reasonable Credit Judgment or (2) such Account otherwise complies with the requirements of the definition of Eligible Account and Agent in its Reasonable Credit Judgment elects to treat such Account as an Eligible Account notwithstanding this clause (vi); or
- (vii) the Account Debtor is the United States of America, or any department, agency or instrumentality thereof, or any other Governmental Authority, unless either (1) the applicable Borrower assigns its right to payment of such Account to Agent in a manner satisfactory to Agent in its Reasonable Credit Judgment so as to comply with the Assignment of Claims Act of 1940 (31 U.S.C. §203 et seq., as amended) or similar state or foreign statutes to the extent applicable, or (2) such Account otherwise complies with the requirements of the definition of Eligible Account and Agent in its Reasonable Credit Judgment elects to treat such Account as an Eligible Account notwithstanding this clause (vii); or
- (viii) it is not at all times subject to Agent's duly perfected, first priority security interest or is subject to a Lien that is not a Permitted Lien; or

- (ix) the Account is evidenced by chattel paper or an instrument of any kind, or has been reduced to judgment; or
- (x) 50% or more of the Accounts owing from the Account Debtor are not Eligible Accounts hereunder by reason of the application of clause (ii) above; or
 - (xi) it represents service charges, late fees or similar charges; or
- (xii) it is not an existing account receivable which would be properly classified as such on the applicable Borrower's books in accordance with GAAP; or
 - (xiii) it is not otherwise acceptable to Agent in its Reasonable Credit Judgment.

Notwithstanding clause (i) of this definition relating to Affiliates, Accounts with respect to which Glencore is the Account Debtor will be considered to be Eligible Accounts provided such Accounts meet all criteria of this definition other than those set forth in clause (i), and provided that with respect to determining compliance with clause (xii), Glencore shall be deemed not an Affiliate of Borrower.

<u>Eligible Inventory</u> — Inventory of any Borrower (other than packaging and shipping materials and supplies, tooling, samples and literature). Without limiting the generality of the foregoing, no Inventory shall be Eligible Inventory if:

- (i) it is not raw materials (including saleable scrap), work-in progress or finished goods; or
- (ii) it is slow-moving, obsolete or unmerchantable; or
- (iii) except with respect to in transit Inventory addressed in clause (vi) below, it is not at all times subject to Agent's duly perfected, first priority security interest or is subject to a Lien that is not a Permitted Lien; or
- (iv) it is not located at one or more of the business locations set forth in <u>Schedule 6.1.1</u> to the Agreement, as updated by Borrowers in accordance with the Agreement; <u>provided</u> that in-transit Inventory shall constitute Eligible Inventory notwithstanding this clause (iv) so long as such in-transit Inventory otherwise complies with the applicable requirements of the definition of Eligible Inventory; or
- (v) it is located on real property not owned or leased by a Borrower unless Borrowers shall have obtained a Bailee Certificate from the owner of the real property on which such Inventory is located; <u>provided</u> that in-transit Inventory shall constitute Eligible Inventory notwithstanding this clause (v) so long as such in-transit Inventory otherwise complies with the applicable requirements of the definition of Eligible Inventory; or

- (vi) it is in transit unless such inventory is either (A) in transit in the United States of America and is either (1) in transit between premises in the United States of America which are either owned or leased by (and in each case controlled by) a Borrower or (2) the subject of an appropriate financing statement filed under the UCC, or (B) in transit outside of the United States of America (including its inland waterways) and the title documents in respect thereof have been consigned to a Title Document Agent pursuant to a Title Document Agency Agreement; provided that the maximum amount of in-transit inventory not located in the United States of America (including its inland waterways) at any one time included as Eligible Inventory shall not exceed \$30,000,000; or
 - (vii) it is located outside of the United States of America and is not in transit; or
- (viii) is subject to a third party's trademark or other proprietary right, unless Agent is reasonably satisfied that it could sell such inventory on satisfactory terms in connection with the exercise of its remedies following an Event of Default; or
 - (ix) it is not otherwise acceptable to Agent in its Reasonable Credit Judgment.

<u>Environmental Laws</u> — all federal, state and local laws, rules, regulations, ordinances, orders and consent decrees relating to health, safety and environmental matters.

<u>ERISA</u> — the Employee Retirement Income Security Act of 1974, as amended, and any successor statute, and all rules and regulations from time to time promulgated thereunder.

<u>ERISA Affiliate</u> — any trade or business (whether or not incorporated) under common control with any Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

ERISA Event — (a) any Reportable Event (except an event for which the 30-day notice period is waived); (b) the existence with respect to any Pension Plan of an "accumulated funding deficiency" (as defined in Section 412 of the Code or Section 302 of ERISA), whether or not waived; (c) the filing pursuant to Section 412(d) of the Internal Revenue Code or Section 303(d) of ERISA of an application for a waiver of the minimum funding standard with respect to any Pension Plan; (d) the incurrence by Borrowers or any ERISA Affiliate of any liability under Title IV of ERISA with respect to the termination of any Pension Plan; (e) the receipt by Borrowers or any ERISA Affiliate from the PBGC or a plan administrator of any notice relating to an intention to terminate any Pension Plan or to appoint a trustee to administer any Pension Plan; (f) the incurrence by Borrowers or any ERISA Affiliate of any liability with respect to withdrawal or partial withdrawal from any Pension Plan or Multiemployee Plan; or (g)

the receipt by Borrowers or any ERISA Affiliate of any notice, or the receipt by any Multiemployer Plan from Borrowers or any ERISA Affiliate of any notice, concerning the imposition of Withdrawal Liability or a determination that a Multiemployer Plan is, or is expected to be, insolvent or in reorganization, within the meaning of Title IV of ERISA.

Event of Default — as defined in Section 10.1 of the Agreement.

Exchange Act — the Securities Exchange Act of 1934.

<u>Facility</u> — as defined in the preamble to the Agreement.

Facility Increase — as defined in Section 1.3 of the Agreement.

<u>Facility Increase Amount</u> — as defined in Section 1.3 of the Agreement.

<u>Facility Increase Effective Date</u> — as defined in Section 1.3 of the Agreement.

<u>Fee Letter</u> — as defined in Section 2.3 of the Agreement.

<u>Federal Funds Rate</u> — for any day, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; <u>provided</u> that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upwards, if necessary, to the next 1/100 of 1%) of the quotations for such day for such transactions received by the Agent from three Federal funds brokers of recognized standing selected by it.

Financial Officer — with respect to any Borrower, the chief financial officer, controller or treasurer of such Borrower.

<u>Fixed Charge Coverage Ratio</u> — with respect to any period, the ratio of (i) EBITDA for such period minus the sum of (a) any provision for (plus any benefit from) income taxes paid in cash included in the determination of net earnings (or loss) for such period <u>plus</u> (b) non-financed Capital Expenditures during such period, to (ii) Fixed Charges for such period, all as determined for Borrowers and Guarantors on a Consolidated basis and in accordance with GAAP.

<u>Fixed Charges</u> — with respect to any period, the sum of: (i) scheduled principal payments required to be made during such period in respect of indebtedness for Money Borrowed (including the principal portion of Capitalized Lease Obligations), <u>plus</u> (ii) Interest Expense for such period (excluding the amortization of financing costs and

original issue discounts as determined in accordance with GAAP), all as determined for Borrowers and Guarantors on a Consolidated basis and in accordance with GAAP.

<u>Foreign Lender</u> — any Lender that is organized under the laws of a jurisdiction outside the United States.

<u>GAAP</u> — generally accepted accounting principles in the United States of America in effect from time to time.

<u>Glencore</u> — Glencore International AG, a Swiss corporation, and its Subsidiaries.

<u>Governmental Authority</u> — any nation or government, any state or other political subdivision thereof, any central bank (or similar monetary or regulatory authority) thereof, any entity exercising executive, legislative, judicial, arbitral, regulatory or administrative functions of or pertaining to government, and any corporation or other entity owned or controlled, through stock or capital ownership or otherwise, by any of the foregoing.

<u>Guarantors</u> — Metalsco, Ltd., Virgin Islands Alumina Corporation LLC, Hancock Aluminum LLC, Skyliner, Inc., Century Aluminum of Kentucky, LLC, and each other Person who now or hereafter guarantees payment or performance of the whole or any part of the Obligations.

<u>Guaranty Agreements</u> — each guaranty executed by any Guarantor guaranteeing payment or performance of the whole or any part of the Obligations.

<u>Hawesville Entities</u>— Century Kentucky, Inc., Hancock Aluminum LLC, Century Aluminum of Kentucky, LLC, Metalsco, Ltd., Skyliner, Inc. and NSA, Ltd.

<u>Indebtedness</u> — as applied to a Person, without duplication:

- (i) all indebtedness of such Person for borrowed money;
- (ii) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments;
- (iii) all obligations of such Person to pay the deferred and unpaid price of property or services to the extent recorded as liabilities under GAAP, excluding trade payables, accruals and accounts payable arising in the ordinary cause of business (in each case, to the extent not overdue):
 - (iv) all Capitalized Lease Obligations of such Person;
- (v) all obligations of such Person (whether contingent or otherwise) in respect of bankers' acceptances, letters of credit, surety or other bonds, and similar instruments;

- (vi) all financial obligations of other Persons secured by a Lien upon Property of such Person;
- (vii) all obligations of other Persons which such Person has guaranteed;
- (viii) all reimbursement obligations in connection with letters of credit or letter of credit guaranties issued for the account of such Person; and
 - (ix) all Derivative Obligations of such Person.

<u>Indenture</u>— the Indenture, dated as of August 26, 2004, among Century, the Guarantors party thereto and Wilmington Trust Company governing Century's 7 ¹/₂% Senior Notes due 2014, as in effect on the date of the Agreement, a copy of which is attached to the Agreement as Exhibit F.

Intellectual Property — all past, present and future: trade secrets, know-how and other proprietary information; trademarks, internet domain names, service marks, trade dress, trade names, business names, designs, logos, slogans (and all translations, adaptations, derivations and combinations of the foregoing) indicia and other source and/or business identifiers, and the goodwill of the business relating thereto and all registrations or applications for registrations which have heretofore been or may hereafter be issued thereon throughout the world; copyrights (including copyrights for computer programs) and copyright registrations or applications for registrations which have heretofore been or may hereafter be issued throughout the world and all tangible property embodying the copyrights, unpatented inventions (whether or not patentable); patent applications and patents; industrial design applications and registered industrial designs; license agreements related to any of the foregoing and income therefrom; books, records, writings, computer tapes or disks, flow diagrams, specification sheets, computer software, source codes, object codes, executable code, data, databases and other physical manifestations, embodiments or incorporations of any of the foregoing; the right to sue for all past, present and future infringements of any of the foregoing; all other intellectual property; and all common law and other rights throughout the world in and to all of the foregoing.

<u>Interest Expense</u> — with respect to any period, interest expense paid or accrued for such period, including the interest portion of Capitalized Lease Obligations, all as determined for Borrowers and Guarantors on a Consolidated basis and in accordance with GAAP.

Interest Period — as applicable to any LIBOR Portion, a period commencing on the date such LIBOR Portion is advanced, continued or converted, and ending on the date which is one month, two months, three months, or six months later, as may then be requested by Borrower; provided that (i) any Interest Period which would otherwise end on a day which is not a Business Day shall end in the next preceding or succeeding Business Day as is Agent's custom in the market to which such LIBOR Portion relates; (ii) there remains a minimum of one month, two months, three months or six months

(depending upon which Interest Period Borrower selects) in the Term, unless Borrowers and Lenders have agreed to an extension of the Term beyond the expiration of the Interest Period in question; and (iii) all Interest Periods of the same duration which commence on the same date shall end on the same date.

<u>LC Amount</u> — at any time, the aggregate undrawn face amount of all Letters of Credit then outstanding.

<u>LC Exposure</u>— at any time, the sum of (i) the LC Amount as of such time, plus (ii) the aggregate amount of unreimbursed LC Obligations as of such time.

<u>LC Obligations</u> — any Obligations that arise from any draw against any Letter of Credit.

<u>Lender</u>— as defined in the preamble to the Agreement.

Letter of Credit — as defined in subsection 1.2.1 of the Agreement.

<u>Letter of Credit Issuer</u> — as defined in the preamble to the Agreement.

LIBOR — as applicable to any LIBOR Portion, for the applicable Interest Period, the rate per annum (rounded upward, if necessary, to the nearest 1/16 of one percent) as determined on the basis of the offered rates for deposits in U.S. dollars, for a period of time comparable to such Interest Period which appears on the Telerate page 3750 as of 11 a.m. (London, England time) on the day that is two LIBOR Business Days preceding the first day of such Interest Period; provided, however, that if the rate described above does not appear on the Telerate System on any applicable interest determination date, the LIBOR shall be the rate (rounded upwards as described above, if necessary) for deposits in U.S. dollars for a period substantially equal to the Interest Period on the Reuters Page "LIBO" (or such other page as may replace the LIBO Page on that service for the purpose of displaying such rates), as of 11:00 a.m. (London, England time), on the day that is two LIBOR Business Days prior to the first day of such Interest Period. If both the Telerate and Reuters systems are unavailable, then the rate for that date will be determined on the basis of the offered rates for deposits in U.S. dollars for a period of time comparable to such Interest Period which are offered by four major banks in the London interbank market at approximately 11:00 a.m. (London, England time), on the day that is two LIBOR Business Days preceding the first day of such Interest Period as selected by Agent. The principal London office of each of the major London banks so selected will be requested to provide a quotation of its U.S. dollar deposit offered rate. If at least two such quotations are provided, the rate for that date will be the arithmetic mean of the quotations. If fewer than two quotations are provided as requested, the rate for that date will be determined on the basis of the rates quoted for loans in U.S. dollars to leading European banks for a period of time comparable to such Interest Period offered by major banks in New York City at approximately 11:00 a.m. (New York City time), on the day that is two LIBOR Business Days preceding the first day of such Interest Period. If that Agent is unable to obtain any such quotation as provided above, it will be determined that LIBOR pursuant to an Interest Period cannot be determined. If the Board of Governors

of the Federal Reserve System shall impose a Reserve Percentage with respect to LIBOR deposits of Bank, then for any period during which such Reserve Percentage shall apply, LIBOR shall be equal to the amount determined above divided by an amount equal to one minus the Reserve Percentage.

<u>LIBOR Business Day</u>. — any day on which commercial banks are open for international business (including dealings in U.S. Dollar deposits) in London or if commercial banks in London are not open, such other LIBOR interbank market as may be selected by Agent in its reasonable judgment exercised in good faith.

<u>LIBOR Interest Payment Date</u> — as to any LIBOR Portion in respect of which the Interest Period is (A) three months or less, the last day of such Interest Period and (B) more than three months, the date that is three months from the first day of such Interest Period and, in addition, the last day of such Interest Period.

<u>LIBOR Option</u>— the option granted pursuant to Section 3.1 of the Agreement to have the interest on all or any portion of the principal amount of the Revolving Credit Loans be based on the LIBOR.

<u>LIBOR Portion</u> — that portion of the Revolving Credit Loans specified in a LIBOR Request (including any portion of Revolving Credit Loans which is being borrowed by Borrower concurrently with such LIBOR Request) which, as of the date of the LIBOR Request specifying such LIBOR Portion, has met the conditions for basing interest on the LIBOR in Section 3.1 of the Agreement and the Interest Period of which has not terminated.

<u>LIBOR Request</u> — a notice in writing (or by telephone confirmed electronically or by telecopy or other facsimile transmission on the same day as the telephone request) from Borrower Representative to Agent requesting that interest on a Revolving Credit Loan be based on the LIBOR, specifying: (i) the first day of the Interest Period (which shall be a Business Day); (ii) the length of the Interest Period; (iii) whether the LIBOR Portion is a new Loan, a conversion of a Base Rate Portion, or a continuation of a LIBOR Portion; and (iv) the dollar amount of the LIBOR Portion, which shall be in an amount not less than \$500,000 or an integral multiple of \$100,000 in excess thereof.

<u>Lien</u> — any mortgage, security interest, pledge, hypothecation, assignment, attachment, deposit arrangement, encumbrance, lien (statutory, judgment or otherwise), charge (whether fixed or floating), preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including any similar such interest arising under the laws of any applicable domestic or foreign jurisdiction and including any conditional sale or other title retention agreement, any financing lease involving substantially the same economic effect as any of the foregoing and the filing of any financing statement under the UCC or comparable law of any domestic or foreign jurisdiction).

<u>Loan Account</u> — as defined in Section 3.6 of the Agreement.

<u>Loan Documents</u> — the Agreement, the Guaranty Agreements, and any and all other agreements, instruments and documents heretofore, now or hereafter executed and/or delivered to Agent or any Lender by any Loan Party in respect of the transactions contemplated by the Agreement.

Loan Party — each Borrower and each Guarantor.

<u>Loans</u> — all loans and advances of any kind made by Agent, any Lender, or any Affiliate of Agent or any Lender, pursuant to the Agreement.

Majority Lenders — as of any date, Lenders holding greater than 50% of the Revolving Loan Commitments determined on a combined basis and following the termination of the Revolving Loan Commitments, Lenders holding greater than 50% or more of the then outstanding Loans and LC Exposure; provided that if prior to termination of the Revolving Loan Commitments, any Lender breaches its obligation to fund any requested Revolving Credit Loan, for so long as such breach exists, (i) its voting rights hereunder shall be calculated with reference to its then outstanding Loans and LC Exposure, rather than its Revolving Loan Commitment and (ii) in determining the total amount of Revolving Loan Commitments of all Lenders, the breaching Lender's Revolving Loan Commitment will be deemed to be equal to its then outstanding Loans and LC Exposure.

Margin Stock — "margin stock" as such term is defined in Regulation T, U or X of the Federal Reserve Board.

Material Adverse Effect — (i) a material adverse effect on the business, condition (financial or otherwise), operation, performance or properties of Borrowers and their Subsidiaries, taken as a whole, (ii) a material adverse effect on the rights and remedies of Agent or Lenders under the Loan Documents, (iii) the impairment of the ability of Borrowers or Guarantors to perform their material obligations hereunder or under the Loan Documents taken as a whole, or (iv) a material adverse effect on Agent's security interest in the Collateral.

Maximum Rate — as defined in subsection 2.1.3 of the Agreement.

Minimum Availability Reserve — \$15,000,000.

Money Borrowed — (i) Indebtedness arising from the lending of money by any Person to any Borrower or any Guarantor; (ii) Indebtedness, whether or not in any such case arising from the lending by any Person of money to any Borrower or any Guarantor, (1) which is represented by notes payable or drafts accepted that evidence extensions of credit, (2) which constitutes obligations evidenced by bonds, debentures, notes or similar instruments, (3) upon which interest charges are customarily paid (other than accounts payable), or (4) that was issued or assumed as full or partial payment for Property; (iii) Indebtedness that constitutes a Capitalized Lease Obligation; (iv) reimbursement obligations with respect to letters of credit or guaranties of letters of credit and (v) Indebtedness of any Borrower or any Guarantor under any guaranty of obligations that would constitute Indebtedness for Money Borrowed under clauses (i) through (iii) hereof,

if owed directly by such Borrower or Guarantor. Money Borrowed shall not include trade payables or accrued expenses.

Mt. Holly Aluminum Facility — the aluminum reduction facility located in Mt. Holly, South Carolina, which is owned jointly by Berkley Aluminum and Alumax of South Carolina, Inc., a subsidiary of Alcoa, pursuant to the Mt. Holly Owners Agreement.

Mt. Holly Owners Agreement — the Amended and Restated Owners Agreement, dated as of January 26, 1996 between Alumax of South Carolina, Inc. and Berkeley Aluminum, Inc. (as amended by a First Amendment, dated as of January 26, 1996 and an Amendment Agreement effective June 15, 1998.

Multiemployer Plan — has the meaning set forth in Section 4001(a)(3) of ERISA.

No-Offset Letter — a letter agreement substantially in the form of Exhibit G to the Agreement.

<u>Notes</u> — the Secured Promissory Notes to be executed by Borrowers on or about the Closing Date in favor of each Lender to evidence the Revolving Credit Loans, which shall be in the form of Exhibit H to the Agreement, together with any replacement or successor notes therefor.

Notice of Exclusive Control — as defined in subsection 6.2.3 of the Agreement.

Notice of Facility Increase — as defined in Section 1.3 of the Agreement.

Obligations — all Loans, all LC Obligations, and all other advances, debts, liabilities, and obligations, together with all interest (including all interest that accrues (or, but for the commencement of any bankruptcy, insolvency or similar proceeding, would accrue) after the commencement of any insolvency, bankruptcy or other similar proceeding of any Borrower, whether or not a claim for post-filing interest is allowed in such proceeding), fees and other charges thereon, of any kind or nature, present or future, whether or not evidenced by any note, guaranty or other instrument, whether direct or indirect (including those acquired by assignment), absolute or contingent, primary or secondary, due or to become due, now existing or hereafter arising and however acquired, owing, arising, due or payable (a) from any Borrower to Agent, for its own benefit, or to Bank or any other Affiliate of Agent, in each case arising under any of the Loan Documents, (b) from any Borrower to Bank, any Affiliate of Bank, or any Lender in respect of Product Obligations, or (c) from any Borrower to Agent for the benefit of any Lender or to any Lender directly, in each case under any of the Loan Documents.

Offerees — as defined in Section 1.3 of the Agreement.

<u>Organizational I.D. Number</u> — with respect to any Person, the organizational identification number assigned to such Person by the applicable governmental unit or agency of the jurisdiction of organization of such Person.

Overadvance — as defined in subsection 1.1.2 of the Agreement.

<u>Pension Plan</u> — a pension plan (as defined in Section 3(2) of ERISA) subject to Title IV of ERISA which any Borrower or any ERISA Affiliate sponsors, maintains, or to which it makes, is making, or is obligated to make contributions.

<u>Permitted Liens</u> — as defined in subsection 8.2.4 of the Agreement.

<u>Permitted Purchase Money Indebtedness</u> — Purchase Money Indebtedness of any Borrower incurred after the Closing Date which is secured by a Purchase Money Lien and the principal amount of which, when aggregated with the principal amount of all other such Purchase Money Indebtedness and Capitalized Lease Obligations of Borrowers at the time outstanding, does not exceed \$10,000,000. For the purposes of this definition, the principal amount of any Purchase Money Indebtedness consisting of capitalized leases (as opposed to operating leases) shall be computed as a Capitalized Lease Obligation.

<u>Person</u> — an individual, partnership, corporation, limited liability company, joint stock company, land trust, business trust, or unincorporated organization, or a government or agency or political subdivision thereof.

<u>Plan</u>— an employee benefit plan (as defined in Section 3(3) of ERISA) which any Borrower sponsors or maintains or to which any Borrower makes, is making, or is obligated to make contributions and includes any Pension Plan.

<u>Product Obligations</u> — every obligation of any Borrower under and in respect of one or more of the following types of services extended to a Borrower by Bank, any Affiliate of Bank or any Lender as agreed from time to time in writing by Borrower Representative: (i) cash management (including controlled disbursement services) and (ii) Derivative Obligations.

<u>Projections</u> — with respect to any Person or Persons, forecasted (i) balance sheets, (ii) profit and loss statements, and (iii) cash flow statements of such Person or Persons.

<u>Property</u>— any interest in any kind of property or asset, whether real, personal or mixed, or tangible or intangible.

<u>Purchase Money Indebtedness</u> — includes (i) Indebtedness (other than the Obligations) for the payment of all or any part of the purchase, lease or improvement of any fixed assets, (ii) any Indebtedness (other than the Obligations) incurred at the time of or within 10 days prior to or after the acquisition of any fixed assets for the purpose of financing all or any part of the purchase, lease or improvement price thereof, and (iii) any renewals, extensions or refinancings thereof, but not any increases in the principal amounts thereof outstanding at the time.

<u>Purchase Money Lien</u> — a Lien upon fixed assets which secures Purchase Money Indebtedness, but only if such Lien shall at all times be confined solely to the fixed assets

the purchase price of which was financed through the incurrence of the Purchase Money Indebtedness secured by such Lien.

<u>Reasonable Credit Judgment</u> — reasonable credit judgment, exercised in good faith, in accordance with Bank's customary business practices in its capacity as agent for asset-based loan facilities comparable to the Facility.

<u>Related Business</u> — the business of mining, reducing, refining, processing and selling bauxite, alumina, primary aluminum and aluminum products, and any business reasonably related, incidental or ancillary thereto.

Reportable Event — any of the events set forth in Section 4043(c) of ERISA.

Requirements of Law — as to any Person, any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or of a Governmental Authority, in each case applicable to or binding upon the Person or any of its property or to which the Person or any of its property is subject.

<u>Reserve Percentage</u> — the maximum aggregate reserve requirement (including all basic, supplemental, marginal and other reserves) which is imposed on member banks of the Federal Reserve System against "Euro-currency Liabilities" as defined in Regulation D.

Reserves — as defined in subsection 1.1.1 of the Agreement.

<u>Responsible Officer</u>— shall mean the Chairman, Chief Executive Officer, President, Chief Financial Officer, Chief Operating Officer, Vice President, Treasurer or Secretary of any Person.

Restricted Investment — any investment by a Person in another Person made by delivery of Property to any Person, whether by (a) acquisition of Securities, Indebtedness or other obligations of another Person, (b) loan, advance, extension of credit or capital contribution to another Person or commitment to do any of the foregoing (including the purchase of Property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such Property to such Person or its Affiliate), (c) acquisition of all or a significant part of the assets of a business conducted by any other Person or all or substantially all of the assets constituting the business of a division, branch, or other unit operation of any other Person, whether through purchase of assets, merger, or otherwise, (d) becoming a partner in any partnership or joint venture, (e) investments in time deposits, certificates of deposit, bankers acceptances and money market, mutual or similar funds, (f) investments arising out of forward contracts, futures contracts, exchange contracts, swaps, options or other financing agreements or arrangements (including caps, floors, collars and similar arrangements), the value of which is dependent upon interest rates, currency exchange rates, commodities indices or other indices, or (g) guaranties of obligations or liabilities of another Person, except the following:

- (i) investments by a Loan Party in one or more of its Subsidiaries that are not also Loan Parties to the extent existing on the Closing Date, and any renewals, extensions, and refinancings of such investments consisting of loans; <u>provided</u> that any such renewal, extension, or refinancing is in an aggregate principal amount not greater than the principal amount of the original investment, and is on terms no less favorable taken as a whole to Borrower or Guarantor making the investment);
- (ii) subject to compliance by Borrowers with subsection 8.2.7 of the Agreement, investments by a Loan Party at any time in one or more other Loan Parties;
 - (iii) investments consisting of Capital Expenditures permitted by subsection 8.2.6 of the Agreement;
 - (iv) Current Assets arising in the ordinary course of business;
- (v) investments in direct obligations of the United States of America, or any agency thereof or obligations guaranteed by the United States of America; provided that such obligations mature within one year from the date of acquisition thereof;
- (vi) investments in time deposit accounts, certificates of deposit, bankers acceptances and money market deposits maturing within one year of the date of acquisition thereof issued by any Lender or a bank or trust company which is organized under the laws of the United States of America, any State thereof or any foreign country recognized by the United States of America, and which bank or trust company has capital, surplus and undivided profits aggregating in excess of \$500,000,000 (or the foreign currency equivalent thereof) and whose long-term debt is rated not less than A-1 if rated by Standard and Poor's Rating Group or P-1 if rated by Moody's Investor Service, Inc. or any money market fund sponsored by a registered broker dealer or mutual fund distributor;
- (vii) investments in commercial paper given a rating of not less than A-1 if rated by Standard and Poor's Rating Group or P-1 if rated by Moody's Investor Service, Inc. and maturing not more than 270 days from the date of creation thereof;
- (viii) investments in money market, mutual or similar funds having assets in excess of \$100,000,000 and the investments of which are limited to investment grade securities;
- (ix) investments made from time to time (A) in Gramercy Alumina LLC and St. Ann Bauxite Limited (1) in an amount not to exceed the sum of \$11.5 million <u>plus</u> any closing or post-closing purchase price adjustments, which investments are used to finance the acquisition of such facility, partnership interests and related assets, (2) in amounts necessary to fund obligations of such

entities with respect to environmental costs, workers' compensation, pensions and benefit plans or self insurance liabilities and other related expenses in an amount not to exceed \$15 million, and (3) made or deemed to be made as a result of Borrowers' funding or obligation to fund one-half of such entities' Capital Expenditures; (B) as required by the Mt. Holly Owners Agreement to fund the general operating activities of the Mt. Holly Aluminum Facility and (C) in connection with the purchases of raw materials (bauxite and alumina) from Gramercy Alumina LLC and St. Anne Bauxite Limited;

- (x) investments and commitments to make investments existing on the date of the Agreement and listed on Schedule 8.2.8 to the Agreement;
- (xi) investments arising out of forward contracts, futures contracts, exchange contracts, swaps, options or other financing agreements or arrangements (including caps, floors, collars and similar arrangements), the value of which is dependent upon interest rates, currency exchange rates, commodities indices or other indices, solely to the extent entered into for bona fide hedging purposes and not for speculative purposes;
 - (xii) guaranties of obligations of other Persons to the extent permitted by subsection 8.2.2 of the Agreement; and
 - (xiii) investments not included in paragraphs (i) through (x) above; <u>provided</u> that:
 - (1) no Default or Event of Default is continuing at the time of such investment or would occur as a result thereof;
 - (2) the total of (A) Availability, <u>plus</u> (B) immediately available funds in bank accounts of the Borrowers and readily marketable investments of the Borrowers of the type described in clauses (v) through (viii) of this definition of the term "Restricted Investments", <u>minus</u> (C) the amount of the proposed investment, shall (x) have averaged at least \$25,000,000 over the immediately preceding 30 days (determined using the Borrowing Base Certificate most recently delivered by Borrowers pursuant to subsection 8.1.4 of the Agreement), and (y) be at least \$25,000,000 immediately following the proposed investment. For purposes of calculating the average Availability over the preceding 30 days on a pro forma basis, (A) (1) net cash proceeds from equity financing, Indebtedness, or the sale of assets other than Collateral permitted to be raised hereunder and raised not more than 30 days prior to the date of the proposed investment and either immediately applied to the proposed investment or (without duplication) held in one or more bank accounts of the Borrowers or invested in readily marketable investments of the type described in clauses (v)

through (viii) of this definition of the term "Restricted Investments" and (2) the amount of the proposed investment shall each be deemed to have been raised and made, as the case may be, on the first day of such 30-day period, and (B) all other immediately available funds and readily marketable investments of the type described in clauses (v) through (viii) of this definition of the term "Restricted Investments" shall be included in the foregoing calculation based on whether or not actually existing in the Borrowers' bank accounts or investment accounts on each date with respect to which the calculation is made; and

(3) solely with respect to the acquisition of all or substantially all of the capital stock or other equity interest or assets of a Person that becomes a Borrower or a Guarantor, Borrowers and Guarantors shall have a <u>pro forma</u> Fixed Charge Coverage Ratio of not less than 1.25 to 1.0 as of the last day of the immediately preceding four fiscal quarters for which financial statements have been filed with the SEC, taken as a single period or, if not so filed, then for which such financial statements were required to have been delivered under the Agreement.

<u>Restrictive Agreement</u> — an agreement (other than any of the Loan Documents) that, if and for so long as a Borrower is a party thereto, would prohibit, condition or restrict such Borrower's right to (a) repay any of the Obligations or perform any of its other obligations under the Loan Documents, (b) grant Liens in favor of Agent pursuant to the Loan Documents upon any of such Borrower's Collateral, or (c) amend, modify or extend any of the Loan Documents.

Revolving Credit Loan — a Loan made by any Lender pursuant to Section 1.1 of the Agreement.

<u>Revolving Credit Maximum Amount</u> — \$100,000,000, as such amount may be increased or reduced from time to time pursuant to the terms of the Agreement.

<u>Revolving Loan Commitment</u> — with respect to any Lender, the amount of such Lender's Revolving Loan Commitment pursuant to subsection 1.1.1 of the Agreement, as set forth below such Lender's name on the signature page hereof or any Assignment and Acceptance Agreement executed by such Lender.

<u>Revolving Loan Percentage</u> — with respect to each Lender, the percentage equal to the quotient of such Lender's Revolving Loan Commitment divided by the aggregate of all Revolving Loan Commitments.

<u>SEC</u> — the United States Securities and Exchange Commission.

Secured Parties — Lenders, Letter of Credit Issuer, Agent, and any other holder of any Obligation.

<u>Security</u> — all shares of stock, partnership interests, membership units or other ownership interests in any Person and all warrants, options or other rights to acquire the same.

Solvent — as to any Person, that such Person (i) owns Property whose fair saleable value is greater than the amount required to pay all of such Person's Indebtedness (including contingent debts), (ii) is able to pay all of its Indebtedness as such Indebtedness matures and (iii) has capital sufficient to carry on its business and transactions and all business and transactions in which it is about to engage.

Stated Termination Date — September 19, 2010.

<u>Subsidiary</u> — any Person of which another Person owns, directly or indirectly through one or more intermediaries, more than 50% of the Voting Stock at the time of determination.

<u>Supporting Letter of Credit</u> — as defined in subsection 1.2.9 of the Agreement.

Swingline Loans — as defined in subsection 1.1.3 of the Agreement.

<u>Tax Liabilities</u> — as defined in subsection 2.9.1 of the Agreement.

<u>Term</u> — as defined in Section 4.1 of the Agreement.

<u>Title Document Agent</u> — UPS Supply Chain Solutions, Inc. or any other Person selected by Borrower Representative after written notice by Borrower Representative to Agent who is reasonably acceptable to Agent to receive and retain possession of negotiable documents (as defined in Section 7-104 of the UCC) issued for any Inventory or other property of Borrowers in accordance with a Title Document Agency Agreement, such receipt and retention of possession being for the purpose of more fully perfecting and preserving Agent's security interests in such negotiable documents and the property represented thereby.

<u>Title Document Agency Agreement</u> — an agreement among a Borrower, a Title Document Agent, and Agent, substantially in the form of Exhibit I to the Agreement.

<u>Type of Organization</u>— with respect to any Person, the kind or type of entity by which such Person is organized, such as a corporation or limited liability company.

<u>UCC</u>— the Uniform Commercial Code as in effect in the State of New York on the date of this Agreement, as it may be amended or otherwise modified.

<u>Unused Letter of Credit Subfacility</u> — at any time, an amount equal to \$25,000,000 minus the LC Exposure at such time.

Unused Line Fee — as defined in Section 2.5 of the Agreement.

<u>Voting Stock</u> — Securities of any class or classes of a corporation, limited partnership or limited liability company or any other entity the holders of which are ordinarily, in the absence of contingencies, entitled to vote with respect to the election of corporate directors (or Persons performing similar functions).

<u>Withdrawal Liability</u> — with respect to a Multiemployer Plan, any "complete withdrawal" or "partial withdrawal", as each of such terms are defined under Sections 4203 and 4205 of ERISA.

Certain Matters of Construction. The terms "herein", "hereof" and "hereunder" and other words of similar import refer to the Agreement as a whole and not to any particular section, paragraph or subdivision. The term "including" means "including without limitation." Any pronoun used shall be deemed to cover all genders. The section titles, table of contents and list of exhibits appear as a matter of convenience only and shall not affect the interpretation of the Agreement. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. All references to any of the Loan Documents shall include any and all modifications thereto and any and all extensions or renewals thereof.

LIST OF EXHIBITS AND SCHEDULES*

Exhibit A	Form of Assignment and Acceptance
Exhibit B	Form of Bailee Certificate
Exhibit C	Form of Blocked Account Agreement — Deposit Accounts
Exhibit D	Form of Borrowing Base Certificate
Exhibit E	Form of Compliance Certificate
Exhibit F	Indenture
Exhibit G	Form of No-Offset Letter
Exhibit H	Form of Note
Exhibit I	Form of Title Document Agency Agreement
Schedule 5.1	Pledged Deposit Accounts
Schedule 6.1.1	Business Locations
Schedule 7.1.4	Organizational Structure Chart
Schedule 7.1.5	Names; Organization
Schedule 7.1.7	Actions to Perfect
Schedule 7.1.16	Litigation
Schedule 8.2.2	Existing Indebtedness
Schedule 8.2.4	Existing Liens
Schedule 8.2.8	Existing Investments

*SCHEDULES AND EXHIBITS ARE OMITTED AND WILL BE FURNISHED TO THE SECURITIES AND EXCHANGE COMMISSION UPON REQUEST

[SCHEDULES AND EXHIBITS ARE OMITTED AND WILL BE FURNISHED TO THE SECURITIES AND EXCHANGE COMMISSION UPON REQUEST]

CERTIFICATION OF DISCLOSURE IN CENTURY ALUMINUM COMPANY'S QUARTERLY REPORT FILED ON FORM 10-O

- I, Craig A. Davis, certify that:
- 1) I have reviewed this quarterly report on Form 10-Q of Century Aluminum Company;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report the Company's conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer and I have disclosed, based on the Company's most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2005

/s/ CRAIG A. DAVIS

Title: Chairman and Chief Executive Officer

CERTIFICATION OF DISCLOSURE IN CENTURY ALUMINUM COMPANY'S QUARTERLY REPORT FILED ON FORM 10-Q

- I, David W. Beckley, certify that:
- 1) I have reviewed this quarterly report on Form 10-Q of Century Aluminum Company;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report the Company's conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- The registrant's other certifying officer and I have disclosed, based on the Company's most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information;
 and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2005

/s/ DAVID W. BECKLEY

Name: David W. Beckley

Title: Executive Vice President and

Chief Financial Officer

Certification of the Chief Executive Officer and Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. 1350)

In connection with the quarterly report on Form 10-Q of Century Aluminum Company (the "Company") for the quarter ended September 30, 2005, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Craig A. Davis, as Chief Executive Officer of the Company, and David W. Beckley, as Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- 1. This Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in this Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Craig A. Davis

By: Craig A. Davis

Title: Chief Executive Officer Date: November 9, 2005

/s/ David W. Beckley

By: David W. Beckley Title: Chief Financial Officer Date: November 9, 2005

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.